Fabrication and Characterisation of Carbon Nanofibres from Cellulose Nanofibres

By

Ehsan Jazaeri

Submitted in fulfilment of requirements for the degree of

Master of Engineering

Deakin University July 2012

DEAKIN UNIVERSITY CANDIDATE DECLARATION



I certify that the thesis entitled

Fabrication and Characterisation of Carbon Nanofibres from Cellulose Nanofibres

submitted for the degree of

Master of Engineering

is the result of my own work and that where reference is made to the work of others, due acknowledgment is given.

I also certify that any material in the thesis which has been accepted for a degree or diploma by any university or institution is identified in the text.

'I certify that I am the student named below and that the information provided in the form is correct'

Full Name	Ehsan Jazaeri (Please Print)	
Signed	Signature Redacted by Library	
Date	22/07/2012	

DEAKIN UNIVERSITY

ACCESS TO THESIS - A

I am the author of the thesis entitled

Fabrication and Characterisation of Carbon Nanofibres from Cellulose Nanofibres

submitted for the degree of

Master of Engineering

This thesis may be made available for consultation, loan and limited copying in accordance with the Copyright Act 1968.

'I certify that I am the student named below and that the information provided in the form is correct'

Full Name	Ehsan Jazaeri (Please Print)
Signed	Signature Redacted by Library
Date	22/07/2012

Dedication

I dedicate my work to my parents Mr. Seyed Mohammadreza Jazaeri and Mrs. Nosrat Imani, my brothers Amir, Shahram, Peiman and Mohammadjavad for their endless love and support in my life.

Acknowledgement

First and foremost, I would like to thank my supervisor Associate Professor Takuya Tsuzuki for his indefatigable guidance and encouragement throughout my study. Without his support and patience, I would not have accomplished this work.

I also thank my co-supervisor Professor Xungai Wang for giving me the opportunity to study at Deakin University as well as great suggestions and help throughout the study.

I would also like to thank Professor Toyoda at Oita University, Japan, for his kind assistance in conducting high temperature graphitization experiments using his furnace. Special thanks also go to Dr. Jinfeng Wang for TEM analysis, Dr. Rajath Hedge for XRD analysis, Dr. Jeff Church for Raman analysis and Dr. Warren Batchelor for providing Pulp.

I wish to thank my friends and colleagues at Deakin University with whom I enjoyed throughout my life in Deakin University. I wish them all the best in endeavour for a better and prosperous life.

I also thank my brother Dr. Seyed Amir Jazaeri for his kind guidance through a different perspective in human sciences that helped me define myself before the world as someone with enormous abilities.

Finally, I would like to extend my sincere gratitude to my wonderful parents who taught me greatly and helped me through hardship and guided me through happy moments in life.

Publication List

Journal articles

1. Jazaeri E, Zhang L, Wang X, Tsuzuki T, Fabrication of carbon nanofibre by pyrolysis of freeze-dried cellulose nanofibre, *Cellulose*, 2011, 18(6), 1481-85.

 Jazaeri E, Tsuzuki T. Effect of Pyrolysis Conditions on the Properties of Carbon Nanofibers Obtained from Freeze-dried Cellulose Nanofibers, *Cellulose* (reviews completed).

Conference papers

- Zhang L, Jazaeri E, Wang X, Tsuzuki T, Production and Characterisation of Cellulose Nanofibre from Softwood Pulp by Soft Mechanical Decomposition, ARNAM/ARCNN Workshop (2010), Adelaide;
- Jazaeri E, Zhang L, Wang X, Tsuzuki T, Eco-friendly fabrication of carbon nanofibre from plant-based nanofibre, The 8th Asia Pacific Conference on Sustainable Energy &Environmental Technologies (APSCEET 2011) (2011), Adelaide;
- Jazaeri E, Tsuzuki T, Wang X. Fabrication and Characterization of Carbon Nanofibre by Pyrolysis of Freeze-Dried Celllulose Nanofibre. Early Career Symposium (Australian Nanotechnology Network), November 2011, Sydney.
- Jazaeri E, Tsuzuki T, Wang X, A Study of Pyrolysis Conditions on Carbon Nanofibre Properties Produced from Freeze-dried Cellulose Nanofibres, The Fibre Society Spring Conference (May 2012), St. Gallen, Switzerland.

Awards

- Recipient of the Commercialization Training Scholarship for the degree of Graduate Certificate in Research Commercialization, 2011;
- 2. Recipient of the prestigious Nano Venture Award for the project: Fabrication of Carbon Nanofibre from Freeze-dried Cellulose Nanofibre, 2011.

Abstract

Although the production of carbon fibres (CFs) from cellulosic precursors has been studied since the 1960s, the fabrication of nano-scale CFs from cellulose nanofibres (CellNFs) is fairly new and many aspects of this research area are still unknown. In addition, the properties of CellNF precursors are expected to greatly affect those of the carbon nanofibres (CNFs) and yet only few types of CellNF precursors have been investigated. Up to date, there has been little research conducted on the carbonisation of CellNFs originated from softwood pulp, due to unavailability of appropriate fabrication methods. Our recent successful fabrication of CellNFs from renewable natural plant products using a green and scalable technique, has led us to investigate the possibility of fabricating CNFs from such materials.

Although carbonisation of micron-sized cellulose fibres have been widely investigated in the past, in this thesis, the novel CellNFs obtained from ball-milled softwood pulp was used to investigate the carbonisation process. Various pyrolysis conditions including the effect of heating stages, heating rates and maximum temperatures in each stage, and holding time at the maximum temperatures were investigated for their effects on the morphological, structural and chemical properties of resulting carbon residues. It was found that the methods to dry CellNF precursors significantly affected the morphology of resulting CNFs. Heat treatment conditions around the thermal decomposition temperature of cellulose also greatly influenced the morphological properties of CNFs. Longer holding time at 240°C and slower heating rate from 240°C to 400°C were critical to preserve the original fibrous structures of the precursor CellNFs after pyrolysis. In addition, the method to dry CellNF precursor influenced the yield of resulting CNFs. Although mass loss occurred during carbonisation, due to the removal of oxygen, hydrogen and carbon, appropriate pyrolysis conditions could reduce the total mass loss after carbonisation. Furthermore, the possibility of graphitizing CNFs at temperatures above 1600°C was demonstrated. Since graphitization starts from the fibre surface and develops towards the core of the fibre, smaller diameters would help the graphitization process to be initiated at lower temperatures. It was shown that a certain crystal structure was developed in the carbonised fibre at lower temperatures (1600°C) than the graphitization temperature of conventional CFs (~ 2800°C).

Table of Contents

Π
Π
V
V
Ί
1
1
1
1
2
4
4
4
6
6
7
9
2
5
5
5
6
6
6

2.1.4.1. Structural composites	16
2.1.4.2. Biomedical composites	16
2.1.4.3. Polymer composites for sports and leisure	17
2.2. Carbon nanofibre	17
2.2.1. Overview	17
2.2.2. Properties	18
2.2.2.1. Mechanical property	18
2.2.2.2. Thermal property	18
2.2.2.3. Electrical property	18
2.2.3. Applications	19
2.2.3.1. Biological applications	19
2.2.3.2. Electrical applications	19
2.2.3.4. Aerospace applications	20
2.2.5. Production	21
2.2. 5. 1. Chemical vapour deposition (CVD)	21
2.2.5.2. Pyrolysis	22
2.2.5.2.1. Pyrolysis of Electrospun nanofibres	23
2.2.5.2.2. Pyrolysis of CellNFs	25
2.2.6. Thermal decomposition mechanism of cellulose	26
2.3. Summary	29
Chapter 3: Experimental	31
3.1. Cellulose nanofibre preparation	31
3.1.1. Overview	31
3.1.2 Cellulose pulp suspension	31
3.1.4. Drying	32
3.2. Carbonisation	32
3.2.1. Below 1000°C	32

3.2.2. Above 1600°C	33
3.3. Characterisation of nanofibres	34
3.3.1. Scanning Electron Microscopy (SEM)	34
3.3.2. Transmission Electron Microscopy (TEM)	35
3.3.3. Fourier Transform Infrared Spectroscopy (FT-IR)	35
3.3.4. X-ray Diffraction (XRD)	36
3.3.5. Thermal Gravimetry/ Differential Scanning Calorimetry (TG/DSC)	37
3.3.6. Raman spectroscopy	37
Chapter 4: Carbonisation of oven-dried cellulose nanofibre	
4.1. Introduction	
4.2. Experimental procedure	
4.2.1. Carbonisation equipment	
4.2.2. Carbonisation conditions	
4.3. Results and discussion	41
4.3.1. Characteristics of oven-dried CellNFs before heat treatment	41
4.3.1.1. Morphology	41
4.3.1.2. Crystallinity	42
4.3.1.3. Chemical bonding	43
4.3.1.4. Thermal properties	43
4.3.2. Effects of heating conditions below 170°C	44
4.3.2.1. Morphology	44
4. 3. 2. 2. Crystallinity	45
4.3.2.3. Chemical structure	46
4.3.3. Effect of holding time at 170°C	47
4.3.3.1. Morphology	47
4.3.3.2. Crystallinity	49
4.3.3.3. Chemical structure	50

4.3.4. Effect of heating rate between 170°C to 240°C	50
4.3.4.1. Morphology	50
4.3.4.2. Crystallinity	51
4.3.4.3. Chemical structure	
4.3.5. Effect of holding time at 240°C	53
4.3.5.1. Morphology	53
4.3.5.2. Crystallinity	54
4.3.5.3. Chemical structure	55
4.3.6. Effect of heating rate between 240°C to 400°C	56
4.3.6.1. Morphology	56
4.3.6.2. Crystallinity	56
4.3.6.3. Chemical structure	57
4.4. Conclusion	58
Chapter 5: Carbonisation of freeze-dried cellulose nanofibre	60
5. 1. Introduction	60
5.2. Experimental procedure	60
5.2.1. Freeze-drying	60
5.2.2. Pyrolysis	60
5.2.2. Carbonisation of freeze-dried CellNFs	61
5.3. Results and Discussion	62
5.3.1. Effect of holding time at 170°C	62
5.3.1.1. Morphology	62
5.3.1.2. Crystallinity	63
5.3.1.3. Chemical bonding	65
5.3.1.4. Discussion on the effect of holding time at 170°C	65
5.3.2. Effect of heating rate between 170°C to 240°C	65
5.3.2.1. Morphology	66

5.3.2.2. Crystallinity	67
5.3.2.3. Chemical structure	68
5.3.2.4. Discussion on the effect of heating rate between 170°C and 240°C	68
5.3.3. Effect of holding time at 240°C	69
5.3.3.1. Morphology	69
5.3.3.2. Crystallinity	70
5.3.3.3. Chemical structure	71
5.3.3.4. Discussion on the effect of holding time at 240°C	72
5.3.4. Effect of heating rate to 400°C	72
5.3.4.1. Morphology	72
5.3.4.2. Crystallinity	74
5.3.4.3. Chemical structure	75
5.3.4.4. Discussion on the effect of heating rates and holding time	76
5.4. Discussion on the effect of heat treatment conditions and drying methods	76
5.4.1. Sensitivity of heat treatment conditions to the formation of CNFs	76
5.4.2. Influence of drying methods on the morphology of carbon residue	79
5.5. Conclusion	80
Chapter 6: Graphitization of carbon nanofibres from freeze-dried cellulos	ie
nanofibre	82
6.1. Introduction	82
6.2. Experimental	82
6.3. Results and discussion	82
6.3.1. Morphology	82
6.3.2. XRD analysis	85
6.3.3. FT-IR analysis	86
6.3.4. Raman analysis	86
6.3.4. Source of contaminant particles	89

6.4. Conclusion	89
Chapter 7: Conclusion and future work	91
7.1. Main conclusions	91
7.2. Suggested future work	93
REFERENCES	96

List of figures

Figure 2.1. Schematic processes of CFs production [9-11].	6
Figure 2.2. Fusion of fibres during carbonisation [34].	
Figure 2.3. Schematic structure of PAN molecules [37-39]	10
Figure 2.4. SEM image of PAN-based CF [110]	12
Figure 2.5. Molecular configuration of cellulose [128]	13
Figure 2.6. Crystal structure of cellulose-I [142].	14
Figure 2.7. Use of CFs in artificial feet [184]	17
Figure 2.8. Schematic illustration of the structure of a fuel cell [169]	20
Figure 2.9. Cabin of Proteus aircraft fabricated by Viper 7-axis fibre placement	
system [184]	20
Figure2. 10. CNF growth using CVD methods.	21
Figure 2.11. CNF lengthening [237]	22
Figure 2.12. TEM image of a CVD grown CNF [238].	22
Figure 2.13. Schematic illustration of the pyrolysis of organic materials	23
Figure 2.14. TEM images of the PAN-based electrospun CNFs [213].	24
Figure 2.15. Internal dehydration of cellulose	27
Figure 2.16. Formation of levoglucosan	27
Figure 2.17. Chain scission of cellulose and formation of gaseous species	28
Figure 2.18. Aromatization of chains and formation of graphitic carbon.	29
Figure 3.1. Spex 8000M shaker mill.	32
Figure 3.2. Tube furnace, TF55035C-1, Lindberg/Blue M.	33
Figure 3.3. Tanmann furnace, school of applied chemistry, faculty of Engineering,	
Oita University	33
Figure 4.1. (a) SEM image of the oven-dried CellNFs and (b) fibre diameter	
distribution analysed from SEM images	42
Figure 4.2. XRD graphs of softwood pulp and oven-dried CellNFs.	42
Figure 4.3. FT-IR spectra of softwood pulp and oven-dried CellNFs	43
Figure 4.4. TGA/DSC graphs of CellNFs.	44
Figure 4.5. SEM images of heat treated samples to 170°C with heating rate of	
1°C/min (NF1), 2°C/min (NF2) and 10°C/min (NF3).	44

Figure 4.6. XRD graphs of heat treated samples with heating rates of 1°C/min (NF1),	
2°C/min (NF2) and 10°C/min (NF3).	45
Figure 4.7. FT-IR spectra of heat treated CellNFs with heating rates of 1°C/min	
(NF1), 2°C/min (NF2) and 10°C/min (NF3); (a) 4000-800cm ⁻¹ , (b) 2000-	
800cm ⁻¹	47
Figure 4.8. SEM images of heat treated samples at 170°C, with holding times of 1 h	
(NF4), 2 h (NF5) and 3 h (NF6).	48
Figure 4.9. Fibre diameter distribution (average of three samples) of NF4, NF5 and	
NF6 estimated from SEM images	48
Figure 4.10. XRD graphs of heat treated samples at 170°C with holding times of 1 h	
(NF4), 2 h (NF5) and 3 h (NF6).	49
Figure 4.11. FT-IR spectra of heat treated CellNFs with holding times of 1 h (NF4),	
2 h (NF5) and 3 h (NF6).	50
Figure 4.12. SEM images of samples heat treated to 240°C, with heating rates of	
1°C/min (NF7), 2°C/min (NF8) and 10°C/min (NF9).	51
Figure 4.13. XRD patterns of heat treated CellNFs, with heating rate of 1°C/min	
(NF7), 2°C/min (NF8) and 10°C/min (NF9)	51
Figure 4.14. FT-IR spectra of heat treated samples, with heating rate of 1°C/min	
(NF7), 2°C/min (NF8) and 10°C/min (NF9)	52
Figure 4.15. SEM images of samples held at 240°C, with holding times of 1 h	
(NF10), 2 h (NF11) and 3 h (NF12).	53
Figure 4.16. Fibre diameter distribution (average of NF-10, NF11 and NF12)	
estimated from SEM images	54
Figure 4.17. XRD graphs of samples heat treated at 240°C, with holding times of 1 h	
(NF10), 2 h (NF11) and 3 h (NF12).	54
Figure 4.18. FT-IR spectra of heat treated samples, with holding times of 1 h (NF10),	
2 h (NF11) and 3 h (NF12).	55
Figure 4.19. SEM images of heat treated samples to 400°C, with heating rate of	
1°C/min (NF13), 2°C/min (NF14) and 10°C/min (NF15).	56
Figure 4.20. XRD graphs of samples heat treated at 400°C, with heating rate of	
1°C/min (NF13), 2°C/min (NF14) and 10°C/min (NF15)	57
Figure 4.21. FT-IR spectra of samples, with heating rate of 1°C/min (NF13), 2°C/min	
(NF14) and 10°C/min (NF15); a) 4000-800cm ⁻¹ , (b) 2000-800cm ⁻¹	58

Figure 5.1. (a) TG/DSC graphs of CellNFs and (b) mass loss at 240°C as a function
of holding time
Figure 5.2. SEM images and fibre diameter distributions of heat treated samples at
170°C, with holding time of (a) 0 h (NF16), (b) 2 h (NF17) and (c) 3h (NF18)63
Figure 5.3. XRD patterns of freeze-dried sample before heat treatment and heat
treated samples at 170°C, with holding time of 0 h (NF16), 2 h (NF17) and 3h
(NF18)64
Figure 5.4. FT-IR spectra of heat treated samples at 170° C with holding time of (a) 0
h (NF16), (b) 2 h (NF17) and (c) 3h (NF18); (a) 4000-800 cm ⁻¹ , (b) 2000-800
cm ⁻¹
Figure 5.5. SEM images of freeze-dried samples heat treated samples to 240°C with
heating rates of 1°C/min (NF19), 2°C/min (NF20) and 10°C/min (NF21)66
Figure 5.6. Fibre diameter distribution (average of NF19, NF20 and NF21) estimated
from SEM images
Figure 5.7. XRD graphs of freeze-dried samples heat treated to 240°C with heating
rates of 1°C/min (NF19), 2°C/min (NF20) and 10°C/min (NF21)67
Figure 5.8. FT-IR spectra of freeze-dried samples heat treated to 240°C with heating
rates of 1°C/min (NF19), 2°C/min (NF20) and 10°C/min (NF21); (a) 4000-800
cm^{-1} , (b) 2000-800 cm ⁻¹
Figure 5.9. SEM images of freeze-dried samples heat treated to 240°C with holding
times of 0 h (NF22), 3 h (NF23) and 17 h (NF24)69
Figure 5.10. Fibre diameter distribution (average of NF22, NF23 and NF24)
estimated from SEM images
Figure 5.11. XRD graphs of freeze-dried samples heat treated to 240°C with holding
times of 0 h (NF22), 3 h (NF23) and 17 h (NF24)70
Figure 5.12. FT-IR spectra of freeze-dried samples heat treated to 240°C with
holding times of 0 h (NF22), 3 h (NF23) and 17 h (NF24); (a) 4000-800 cm ⁻¹ ,
(b) 2000-800 cm ⁻¹
Figure 5.13. SEM images of samples heat treated to 400°C under different
conditions; holding time at 240°C and heating rate to 400°C are 17 h and
1°C/min (NF25), 17 h and 10°C/min (NF26), and 1 h and 1°C/min (NF27)73
Figure 5.14. fibre diameter distribution of sample NF25 estimated from SEM
images74

Figure 5.15. Appearance of (a) freeze-dried CellNFs and (b) carbon residue after	
heat treatment at 400°C	74
Figure 5.16. XRD graphs of samples heated to 400°C under different conditions;	
holding time at 240°C and heating rate to 400°C are 17 h and 1°C/min (NF25),	
17 h and 10°C/min (NF26), and 1 h and 1°C/min (NF27)	75
Figure 5.17. FT-IR spectra of samples heat treated to 400°C under different	
conditions; holding time at 240°C and heating rate to 400°C are 17 h and	
1°C/min (NF25), 17 h and 10°C/min (NF26), and 1 h and 1°C/min (NF27); (a)	
$4000-800 \text{ cm}^{-1}$, (b) 2000-800 cm ⁻¹	75
Figure 5.18. Intramolecular dehydration in cellulose chains	77
Figure 5.19. Formation of levoglocosan associated with depolymerisation.	78
Figure 5.20. Model of cellulose pyrolysis [324]	78
Figure 6.1. TEM images of heat treated samples; a) 1600°C, b) 1800°C, c) 2000°C,	
d) 2200°C, e) 2500°C and f) 2800°C.	83
Figure 6.2. SEM images of heat treated samples; a) 1600°C, b) 1800°C, c) 2000°C, d)	
2200°C, e) 2500°C and f) 2800°C	84
Figure 6.3. SEM images of fine fibrillar structure in 2000°C sample; a) lower	
magnification and b) higher magnification	85
Figure 6.4. XRD patterns of heat treated samples.	85
Figure 6.5. FT-IR spectra of heat treated samples; a) 4000-800 cm ⁻¹ , (b) 2000-800	
cm ⁻¹	86
Figure 6.6. Raman spectra of heat treated samples; a) 1600°C, B) 1800°C, C)	
2000°C, D) 2200°C, E) 2500°C and F) 2800°C	87
Figure 6.7. Raman spectra of CVD grown CNF and graphite	87
Figure 6.8. D/G peak intensity ratio as a function of heat treatment temperature	88
Figure 6.9. Raman shift of the peak G as a function of heat treatment temperature	89

List of Tables

Glossary

CF: Carbon Fibre CellNF: Cellulose NanoFibre CNF: Carbon NanoFibre FT-IR: Fourier Transform Infra Red PAN: PolyAcryloNitrile SEM: Scanning Electron Microscopy TEM: Transmission Electron Microscopy XRD: X Ray Diffraction

Chapter 1 Chapter 1: Introduction

1.1 Significance of research

1.1.1 Research problem

The production of micron-scale carbon fibres (CFs) from cellulosic precursors has been studied since the 1960s. However, the fabrication of nano-scale CFs from cellulose nanofibres (CellNF) is fairly new and many aspects of fabrication processes are still unknown. In addition, the properties of CellNF precursors are expected to greatly affect those of the carbon nanofibres (CNF) and yet only few types of CellNF precursors have been investigated. To date, there has been little research conducted on the carbonisation of CellNFs originated from softwood pulp, due to unavailability of appropriate fabrication methods. Recently, Deakin University demonstrated successful fabrication of CellNFs from renewable natural plant products using a green and scalable technique. This has led us to investigate the possibility of fabrication of CNFs from such materials. In the current project, fabrication of CNFs from such CellNFs by pyrolysis is investigated.

1.1.2 Research aim

In order to study the potential of producing CNFs from plant-based CellNFs, this research project focuses on the following three specific aims:

1- Study of the effects of carbonisation conditions, on the structural properties of carbon residues

In this project, the production of CNFs is studied using pyrolysis. These carbonisation conditions are expected to affect the structural properties of the carbon residues such as crystallinity and morphology, because heat treatment conditions influence the depolymerisation of the cellulose molecular chains and the formation of new atomic bonding between carbon atoms during pyrolysis. Hence, the effects of pyrolysis conditions on the structural properties of carbon residues are studied. The process parameters studied in each heating stage include (i) a heating rate by which the sample is heated to a certain temperature, (ii) a maximum temperature to which the sample reaches

in each stage and (iii) holding time at the maximum temperature. All the heat treatments are conducted in an inert gas atmosphere under nitrogen gas flow.

2- Study of the effects of the methods to dry CellNF precursors on the structural properties of carbon residues

The spatial distance between the fibres may play an important role in maintaining the fibrous morphology in carbon residues. Oven-drying techniques result in a cellulosic film that contains highly compact fibres. On the other hand, freeze-drying techniques yield a 3-dimensional loose nanofibre network. Therefore, carbonisation of oven-dried and freeze-dried CellNFs will lead to carbon residues with different structural and morphological properties. In this project, the effect of the techniques to dry CellNF precursors, on the morphological properties of carbon residues is studied, using oven-drying and freeze-drying techniques.

3- Feasibility of the graphitization of amorphous CNFs

Graphitization of carbonised nanofibres is expected to occur when the nanofibres are exposed to very high temperatures above 2000°C. Normally, graphitization of conventional micron-sized fibres requires temperatures above 2800°C. Since graphitization starts from the fibre surface, it is expected that the graphitization takes place at lower temperatures in nanofibres than in conventional micron-sized fibres. This means that graphitized CNFs may be produced with less energy and cost, than conventional CFs. In this study, after the carbonisation conditions are optimized and CNFs with improved structural and morphological properties are produced, the possibility of graphitizing carbonised nanofibres is investigated at the temperatures between 1600°C and 2800°C.

1.2. Thesis outline

This thesis consists of seven chapters including this chapter. The outlines of each chapter are described as follow:

Chapter 2 is a literature review about CNFs as well as conventional micron-sized CFs. The chapter describes the advances in CF research, the properties of CFs, commercial production methods and industrial applications of CFs. The descriptions of CNFs

including their properties, production methods and various applications are also described.

Chapter 3 gives the experimental procedures. The methods to prepare CellNF precursors, pyrolysis conditions, drying techniques, characterisation instruments and their set-ups are explained.

Chapter 4 presents the study on the pyrolysis of oven-dried CellNFs. The effects of heating rates and holding time in each pyrolysis stage were studied. Oven-dried CellNFs were pyrolysed in three heating stages: 25-170°C, 170-240°C and 240-350°C. The heating stages were designed on the basis of the thermal behaviour of CellNFs. The samples were characterised after each pyrolysis stage to understand the effect of heating conditions on the morphology, crystal structure and chemical bonds of CellNFs.

Chapter 5 presents the study on the pyrolysis of freeze-dried CellNFs. The pyrolysis conditions such as heating rate and holding time were investigated. Based on the thermal behaviour of CellNFs, three heating stages were selected: 25-170°C, 170-240°C and 240-400°C. CNFs were successfully fabricated from freeze-dried CellNFs. It was found that the heating conditions largely affect the fibrous morphology during carbonisation. Comparing the results in Chapter 4 and Chapter 5, the effects of drying techniques on the morphological properties of carbon residues are discussed.

Chapter 6 investigates graphitization of CNFs. Amorphous CNFs obtained from the pyrolysis at 400°C were heated to higher temperatures (1600°C, 1800°C, 2000°C, 2200°C, 2500°C and 2800°C) and the effect of the temperature on the morphological, crystalline and chemical properties were investigated.

Chapter 7 summarizes the main findings of the project about the effects of drying methods and pyrolysis conditions of CellNFs, on the properties of carbon residues. Proposed future works are also presented.

Chapter 2: Literature review

In this chapter, the advances in CF research, the properties of CFs, commercial production methods and industrial applications of CFs are presented. The descriptions of CNFs including their properties, production methods and various applications are also described.

2.1. Carbon fibre

2.1. 1. Overview

CFs have > 92% carbon atoms in their structure. Such structures can be amorphous, semi crystalline or crystalline. Crystalline CFs have the hexagonally oriented rings of graphite. Carbon atoms have connected to each other via covalent bonding and form hexagonal rings. Such rings then connect in the cylindrical plane to form graphitic layers. These layers have connected in directions perpendicular to the layer, via van der Waals bonding. Such a molecular configuration gives the CFs two properties: a) they are good electrical and thermal conductors in the planes and good thermal insulators in the direction perpendicular to the planes and b) they have high strength (high modulus of elasticity) in the planes and low strength (low modulus of elasticity) in the direction perpendicular to the planes [1]. In CFs graphitic layers are highly aligned along the fibre axis. This gives the fibres high elastic modulus, large crystallite sizes (D_{cr}), high density, high tensile modulus, high electrical and thermal conductivities along the fibre axis. The highly aligned graphitic layers along the fibre axis also give electrical and thermal insulation properties to the fibre axis [2]. Graphite is made up of graphene layers that are stacked up in a way that only half of one graphene layer (layer A) is on top of another layer (layer B). This orientation is called ABAB sequence. Graphene layers can also be stacked up in ABCABC sequence which is found when half of layer B is on top of layer A while half of layer C is on top of layer B. While most of the graphitic materials have ABAB or ABCABC sequences [3-5], graphene layers in CFs can be stacked up parallel to one another but with no particular sequence. They can also be amorphous and are stacked up unparallel to each other although the layers were highly developed in the direction parallel to the fibre axis.

			Modulu				
	Densit		s of			Specific	
	У	Tensile	elasticit	Ductilit	Meltin	modulu	Specific
	(g/cm3	Strengt	У	У	g temp	S	strength
Material)	h (GPA)	(GPa)	(%)	(°C)	$(10^{\circ} m)$	(10^4 m)
E-glass 2.							
55	2.55	3.4	72.4	4.7	<1725	2.9	14
S-glass 2.	0.5	4.5	06.0	5.0	1705	2.56	10
50	2.5	4.5	86.9	5.2	<1725	3.56	18
S1Ot 2. 19	2.19	5.9	72.4	8.1	1728	3.38	27.4
SiOt 3. 95	3.95	2.1	380	0.55	2015	9.86	5.3
ZrOz 4. 84	4.84	2.1	340	0. 62	2677	7.26	4.3
CF (high-			• • • •			10.0	10
strength)	1.5	5.7	280	2	3700	18.8	19
CF (high-	1.5	1.0	520	0.26	2700	26.2	10
modulus)	1.5	1.9	530	0.36	3700	36.3	13
Cellulose	-	0.9	41	2.5	-	-	-
BN	1.9	1.4	90	1.6	2730	4.78	7.4
Boron	2.36	3.4	380	0.89	2030	16.4	9.9
B4C	2.36	2.3	480	0.48	2450	20.9	5.1
SiC	4.09	2.1	480	0.44	2700	12	0.3
TiB ₂	4.48	0.1	510	0.02	2980	11.6	7.1
Be	1.83	1.28	300	0.4	1277	19.7	2
W	19 4	4	410	0.98	3410	2 2	27 4
Polvethyle	17.1		110	0.90	5110	2.2	27
ne	0.97	2.59	120	2.2	147	12.4	25.7
Kevlar	1.44	4.5	120	3.8	500	8.81	53.3
Al ₂ O ₃							
whiskers	3.96	21	430	4.9	1982	11	47
BeO							
whiskers	2.85	13	340	3.8	2550	12.3	56.1
B ₄ C							
whiskers	2.52	14	480	2.9	2 4 5 0	19.5	66.5
SiC							
whiskers	3.18	21	480	4.4	2 700	15.4	44.5
Si ₃ N ₄							
whiskers	3.18	14	380	3.7	-	12.1	12
Graphite	1		702	_	2 500	42	100
whiskers	1.66	21	703	3	3 700	43	128
Cr		0.0	0.40	2.7	1.000		10
whiskers	7.2	8.9	240	3.7	1 890	3.4	12

Table 2.1. some properties of synthetic fibres [6, 7].

Table 2.1 compares the properties of CFs with those of some other fibres. It is noted that 'high-strength CFs' have the highest tensile strength among all fibres (5.7 GPa) while 'high-modulus CFs' have the highest modulus of elasticity (530 GPa). Moreover, CFs

exhibit the lowest density among the fibres (1.5 g/cm³) which makes the specific modulus (modulus/density) of CFs significantly higher than that of even Kevlar fibres.

Commercial CFs are produced for three main purposes:

- 1- General purpose CFs: They are mainly amorphous in structure with low modulus, strength and production cost.
- 2- High performance CFs: They are graphitic in structure with high modulus, strength and higher price than general purpose CFs.
- 3- Activated CFs: They are porous in their micro structures which make them highly adsorptive to other chemicals [8].



Graphiuzation Graphiuzation Graphiuzation Activation

Figure 2.1. Schematic processes of CFs production [9-11].

2.1.2. Production

2.1.2.1. Overview

Commercially available CFs are mainly produced from pyrolysis of synthetic precursors such as polyacrylonitrile (PAN), oil- or coal-based precursors such as pitch, and natural precursors such as cellulose (Figure 2.1). Pitch-based CFs have higher strength than PAN-based CFs, since pitch can be better graphitized in higher temperatures. The cost of producing CFs from PAN is higher than pitch or cellulose (Table 2.2) [10].

	Cost of precursor (\$/kg)	Cost of CFs (\$/kg)
PAN-based	0.4	60
Cellulose	0.31	4
Isotropic pitch-based	0.6	22

Table 2.2. Cost of CF production [10].

Polymeric precursors such as polyvinylidene chloride, polyvinyl alcohol and PAN are commonly used for CF production. Table 2.3 shows the mass loss during carbonisation of PAN, pitch and cellulose fibres.

Precursor
fibresWeight loss
(%)Pitch30PAN60, 67cellulose88

Table 2.3. precursor mass loss during carbonisation [12, 13].

2.1.2.2. Pitch

Pitch is a black/brown sticky material that is usually solid at room temperature and melts at higher temperatures. It can be produced from partial pyrolysis of some polymers such as polyamides and polyesters, or coal and petroleum [14]. Pitch is a combination of some monomers that form a three- to eight-ring molecular structure with a molecular weight of 300 to 400 [15].

In order for pitch to be used as the precursor in the production of CFs, pitch should fulfil certain requirements. It should be free from metallic ions and insoluble particles that would reduce the mechanical properties of CFs [16]. In addition, it should have high carbon content and be mesophase in its structure. Mesophase pitch is a mixture of aromatic hydrocarbons containing anisotropic liquid-crystalline particles that can be obtained by heat treatment of pitch at 250-400°C. Mesophase helps produce fibres with high molecular orientation that are suitable for CF production [17, 18], because liquid-crystalline materials orient easily during fibre formation [19].

CFs with high degrees of molecular orientation have high elastic moduli and thermal conductivities whereas CFs with more discontinuous and less ordered fibrillar structures have higher tensile strengths. The latter can be used as reinforcement components for high-strength composites [20, 21]. By increasing the molecular orientation during fibre formation, the strength of pitch-based CFs can be improved [22, 23]. Linearizing the molecular orientation during fibre formation can yield pitch-based fibres with improved thermal conductivities. Controlling such structures during fibre formation can optimize CF as well as improve processing conditions [21, 24].

The crystal structure of pitch is isotropic in nature and hence pitch requires some heat treatment to become anisotropic suitable for fibre spinning. In order for the pitch to become anisotropic, it is held at 400°C for some time (14-32 h [25]). Spinning of anisotropic pitch into pitch fibres is not an easy process, and several alternative methods, such as melt spinning, jet spinning and centrifugal spinning, were proposed to obtain pitch fibres. Melt spinning is however the most favourable method among all. Conventional melt spinning methods can be used to prepare pitch fibres. Molten anisotropic pitch with a certain viscosity passes through spinnerets that forms continuous pitch filaments. Air is flown to the spun filaments to cool them, and then the filaments are collected [7, 26-28]. In order to obtain CFs with high tensile strength and high modulus, pitch fibres are drawn to increase molecular orientation during the spinning process. The stretching can be done within 5mm of the spinneret while the pitch fibres are still hot enough to be stretched. Pitch fibres are then carbonised to obtain CFs. The spinneret shape and the pressure applied to the molten pitch during spinning govern the mechanical and structural properties of the resultant CNFs. For instance, stirring the molten pitch in the spinneret nozzles helps formation of more oriented and even fibres. Applying the correct pressure during spinning also avoids the off-gassing phenomenon.

Pitch-based CFs are produced by pyrolysis at various temperatures. It is critical to maintain their fibrous structure and avoid infusion during pyrolysis. Stabilization is a process to partially introduce oxygen atoms in pitch polymer molecules in order to prevent the pitch fibres from fusing together during the carbonisation (Figure 2.2) [29-32]. Surface oxidation helps pitch fibres to behave as if they are thermoset fibres instead of thermoplastic fibres, so that the fibrous structures are retained during carbonisation [33].



Figure 2.2. Fusion of fibres during carbonisation [34].

The oxidative stabilization step is normally performed at 350°C. Pitch fibres are then suspended in graphite or carbon black to enhance the fibre stability during carbonisation. Stabilized pitch fibres are then carbonised in a series of ascending heating temperatures. CFs are then graphitized at above 2000°C.

2.1.2.3. PAN

PAN is a polymer with the molecular weight of 120,000 g/mol and acrylonitrile content of >85%. PAN is widely used for the production of CNFs (Figure 2.3). Its decomposition temperature is below melting point (350°C). Hence PAN fibres could not be produced by melt-spinning. Commercial production of PAN fibres uses wet spinning techniques with dimethyl formamide (DMF) as a solvent for PAN.

As a precursor of commercial CFs, high purity PAN is not commonly used. PAN precursors usually contain other monomers such as itaconic acid, acrylic acid, and methacrylic acid [35]. Although mass loss is higher with PAN than pitch, PAN is a cheaper and more favourable precursor than pitch in CF production [36].



Figure 2.3. Schematic structure of PAN molecules [37-39].

The process to produce PAN fibre precursors is critical to assuring the high quality in resulting CFs. All stages of the fibre production process should be done accurately in order to obtain PAN fibres with least defects in their molecular structures. The molecular defects such as amorphous regions, folded and entangled parts can reduce fibre strength [40]. To minimize such defects, partial oxidation of PAN fibres during stabilization [41-43] and stretching [44-47] is applied. However, this causes reduction of the tensile strength and elongation at break of PAN fibres [48-50].

Once PAN fibres are spun via wet spinning, they undergo stabilization at 300°C and carbonisation above 1000°C to convert into CFs. The stabilization stage comprises three main reactions; cyclization, dehydrogenation and oxidation. The stabilization stage is necessary to increase carbon yield in the carbonisation stage [9]. During cyclization

reaction, C=N groups are converted into C=N and benzene rings are formed along the fibre axis. Dehydrogenation reaction comprises removal of H atoms from the benzene rings and formation of C=C groups in the rings [51]. During oxidation process, OH groups are formed on the rings to further stabilize the fibres. Up to 180° C, the heating rates can be as high as 5° C/min [52-55]. However, it should be maintained from 180° C up to 350° C [56-59]. The maximum temperature should be kept below the decomposition temperature of PAN to let the volatile gases released from the fibres before chain scission and to obtain higher carbon yield [35, 46, 54, 60-64]. During the carbonisation, nitrogen, oxygen and hydrogen are released from the molecules in the form of gaseous by-products [54, 58, 65-71]. This stage is then completed at 1000° C and CFs with carbon content of >92% in the molecules are left.

During carbonisation, volatile gases such as HCN, NH₃, CH₄ and H₂ are formed and released from the structure [12]. Heating rates below 600°C should be less than 5°C/min to allow the volatile gases to be removed from the fibres with least damage to the fibre structure. Above 600°C, higher heating rates can be applied since only carbon and nitrogen are left in the fibre molecules [72]. Carbonisation of PAN fibres are completed at 1500°C where the residual nitrogen is removed from the fibre structure and CFs with < 95% carbon in the molecule is produced [73-75]. Although graphitization or carbonisation takes only minutes to complete, the stabilization stage takes some hours [76]. This makes carbonisation of PAN fibres costly. In addition, the mass loss during carbonisation is over 50%. These factors add to the final price [77] and make PAN-based CFs expensive. HCN which is released during carbonisation is an extremely toxic gas with a linear molecule comprising the molecular structure of H- C=N. The treatment of PAN-based CFs.

The structural properties of CFs greatly influence their electrical, mechanical and thermal properties. Important structural aspects are: (i) crystallinity such as crystallite size (D_{cr}), degree of crystallinity (I_{cr}) and interlayer spacing d, (ii) graphene layers orientation along and perpendicular to the fibre axis (texture), and (iii) shape and orientation of defects and voids [78-90]. Generally larger crystallite sizes, higher degrees of crystallinity, lower interlayer spacing and higher graphene alignments along the fibre axis result in CFs with greater tensile and modulus values and electrical and thermal conductivities [91-96]. However, fibre production methods, especially heat treatment and processing, affect the

structures of CFs and influence such properties. PAN-based CFs exhibit different structures in the fibre core from that of the skin [97-103]. Heat treatment to over 2000°C results in graphitic layers along the fibre surface. Graphene layers along the skin prohibit the heat to get absorbed into the fibre core and hence the core does not develop the graphitic structure. Therefore, PAN-based CFs cannot be graphitized as much as pitch-based CFs [104-108]. Shorter distances between graphene layers, d, results in higher mechanical strength and improved electrical and thermal conductivities along the fibre axis [109]. Figure 2.4. shows the scanning electron microscopy image of typical PAN-based CFs [110].



Figure 2.4. SEM image of PAN-based CF [110].

2.1.2.4. Cellulose

Compared to pitch or PAN precursors, cellulose is more widely available and comparably cheaper. Cellulose is a renewable raw material and widely available on earth with the estimated yearly-production of 10^{10} tons. Cellulose is found in plants, sea shelves and some animals such as tunicates and fungi [111-114]. It consists of a long linear homopolymer of $1\rightarrow 4$ - β -D-glucosidic rings with the formula of (C₆H₁₀O₅)_n (n=degree of polymerization of glucose) that are combined to form a highly ordered cellulosic chains (Figure 2.5) [115, 116]. Natural cellulose has the cellulose-I structure (Figure 2.6) with a high degree of polymerization (DP) of over 10000 but processing of cellulose-I often reduces the DP to 2500 [10]. These chains form parallel nanoscale fibrillar structures which again link together via strong hydrogen bonds to form cellulose fibres [117]. These intermolecular hydrogen bonds break in the aqueous state because hydroxyl groups (OH) in water compete with those in the cellulose. The intra-molecular hydrogen bonds are

comparatively stiff and make the anhydro-glucose rings stable. Intermolecular hydrogen bonds are responsible for the fibrillar nature of cellulose [118-120]. The distance between chains is ~ 0.45 nm and the distance between two adjacent rings is ~ 0.5 nm.

Cellulose has a high carbon content of 44.4 wt% in addition to 49.4 wt% of oxygen and 6.2 wt% of hydrogen [121, 122]. When cellulose is heated to its decomposition temperature (\sim 320°C), the polymeric chains break down and volatile gases such as CO, CO₂ and H₂O are formed, and hence cellulose loses around 40% of its weight [123-127].



Figure 2.5. Molecular configuration of cellulose [128].

The degree of crystallinity varies in different celluloses from 30% to over 70%. Chemical and mechanical treatment can reduce the crystallinity to less than 50% [129-137]. Highly crystalline cellulose has long cellulosic chains parallel to one another and oriented along the direction of the chains (Figure 2.5) [138, 139]. Mechanical properties such as tensile strength also increase when crystallinity improves [140, 141].

Cellulose crystals have several polymorphs [142]. Native plant cellulose has crystal structures of I_{α} or I_{β} . It is known that treatment of native cellulose under different temperatures and chemical environments can transform the crystal structure to many other forms [142]. Both cellulose I_{α} and I_{β} consist of sheets placed parallel to each other [144]. In the cellulose I_{β} structure, the sheets are placed parallel to the unit cell axis b. On the other hand, in the cellulose I_{α} structure, the sheets are placed parallel to the diagonal direction of the cell axis a and b. Due to the close similarity in the structure, it is difficult to distinguish cellulose I_{α} and I_{β} structures by X-ray powder diffraction [144]. Normally nuclear magnetic resonance spectroscopy or 2-dimensional X-ray diffraction study is used to identify these crystal phases.



It has been reported that cellulose I_{β} form dominates in softwoods [143].

Figure 2.6. Crystal structure of cellulose-I [145].

Cellulose was used as a precursor to produce CFs in Thomas Edison's laboratory in 1880 in search of finding a filament for his light bulbs. However, it took another 80 years, in 1959, before the National Carbon Company used cellulose-based precursor to produce CFs. Since then, many researchers have studied the production of CFs by pyrolysis of cellulose using various heat treatments.

Currently most of the commercial CFs with micron size diameters are produced by the pyrolysis of precursors made from fossil fuels, such as PAN [146] or oil-/coal-based pitch [147]. Viscose rayon precursors have also been used while prepared by some hazardous chemicals to enhance the properties. Hence those methods pose environmental and cost-related concerns. Despite this drawback, the investigation of CF production by the pyrolysis of renewable materials such as native cellulose has rarely appeared in the literature [148, 149]. Although the pyrolysis of bulk cellulosic materials has been extensively investigated in the past, most of the studies focused on the production of energy or bulk char rather than fibres [111, 149-160]. The formation of CFs by pyrolysis of cellulose is more challenging than the formation of activated carbon, as the fibrous structure needs to be preserved during carbonisation. In particular, the formation of volatile and often sticky tar (levoglucosan) should be avoided.

2.1.3 Properties

2.1.3.1. Mechanical properties

Mechanical properties of CFs are usually explained by tensile strength and Young's modulus. The actual tensile strength of CFs is lower than the theoretical one, because the alignment of graphitic layers along the fibre axis is not perfect and crystallite size is lower than the ideal CFs [161]. Table 2.4 shows the theoretical and actual values for tensile strength and Young's modulus of CFs [162].

CF					
precursor	Tensil	e strength	Young's modulus		
	theoretical value (GPa)	actual value (GPa)	theoretical value (GPa)	actual value (GPa)	
Pitch	25	3-4	2000	500	
PAN	15	2-3	1000	250	
cellulose	8	0. 8-1	1000	80	

Table 2.4. Mechanical properties of CFs [146].

Tensile strength depends on the orientation of the graphene layers along the fibre axis as well as the interlayer spacing; the higher the alignment and lower interlayer spacing (d), the higher the tensile strength. The tensile strength increases as the fibre diameter decreases [163]. In addition, pitch-based CFs show higher tensile strength than PAN- or cellulose-based CFs, because the graphene layers are highly oriented along the fibre axis in pitch-based CFs. However, increase in tensile strength results in decrease in shear modulus, since the higher orientation of graphene layers along the fibre axis makes fibres weaker to shear forces.

2.1.3.2. Electrical properties

Although CFs are electrically conductive to a certain extent, their conductivity is much lower than that of metals except for highly graphitic CFs. One method to increase the electrical conductivity of CFs is intercalation [162]. In this method, metal particles such as copper [164], nickel [165, 166] or bromine [167, 168] are inserted in between graphene layers so as to increase the electrical conductivity up to 70% [169]. This

method is only possible in graphitic CFs which have well-developed graphitic structures [170, 171].

2.1.3.3. Thermal properties

Thermal conductivity of highly graphitized CFs is generally higher than that of metals such as copper [172, 173]. In addition, increase in tensile strength increases the thermal conductivity due to the more developed graphitic structure. [174-176].

2.1.3.4. Chemical properties

Graphitized CFs exhibit high chemical resistance in harsh chemical environments. Due to highly oriented graphene layers as well as highly stable benzene rings, graphitized CFs have low tendency to react with acidic or basic agents [177-180].

CFs can be oxidised to modify some physico-chemical properties such as polarity. Polar CFs have enhanced electromechanical properties suitable for electrical and composite applications [181-186]. Partial oxidation may increase the weight while complete oxidation results in 100% mass loss depending on the oxidation duration.

2.1.4. Application

CFs are widely used in the production of composite materials for various purposes. Excellent mechanical properties and lightweight of CFs are suitable as reinforcement components in polymer and many other matrices.

2.1.4.1. Structural composites

Inclusion of CFs in polymer matrices can increase electrical and thermal conductivity, chemical resistivity and also mechanical strength [187]. Pitch- or PAN-based CF/cement composites show lower deflection than cement matrix when a given load is applied [188]. The composites can be used in a variety of applications from aerospace to sport and biomedical applications. Some examples are given in the following sections.

2.1.4.2. Biomedical composites

CF composites are used to produce artificial feet (Figure 2.7) [189]. The strength and flexibility of CFs enable the artificial feet to adjust to the person's pace and direction.



Figure 2.7. Use of CFs in artificial feet [187].

CFs are also widely used in medical applications such as in dental [190, 191] and surgical instruments [192-194].

2.1.4.3. Polymer composites for sports and leisure

CFs are widely used in the production of sport equipment such as bicycle, yacht, bodybuilding equipment, hockey, golf shaft, tennis and badminton rackets, fishing and ski equipment in a form of polymer composites [195, 196]. CF bicycles are much lighter than conventional aluminium bicycles with the same or better strength and flexibility. A 100% CF bicycle can weigh only 1.05 kg. CF composites reduce production costs significantly while enhancing the performance [197-199].

2.2. Carbon nanofibre

2.2.1. Overview

Due to their excellent mechanical, thermal and electrical properties, CNFs have attracted much attention in many applications. Graphitic CNFs have successfully used in nanocomposites in automotive [200-202] and sensors and electrode in electronics [203], gas sensor [204, 205] and solar cells [206, 207]. Graphitized CNFs consist of long graphitic layers that are oriented along the fibre axis and stacked up perpendicular to the fibre axis [5-7]. Such structural orientation has provided CNFs with very attractive properties such as ultra high aspect ratios (the ratio of fibre length to its diameter) and high specific surface area. They exhibit very high tensile strength and modulus which are
comparative with metals. Superior properties of CNFs have promised bright future in many applications where strength is needed combined with flexibility.

2.2.2. Properties

2.2.2.1. Mechanical property

Theoretically, CNFs show tensile strength of 100 GPa and modulus of 1000 GPa. However, in reality, CNFs from various precursors would have tensile strength and modulus of 2-5 GPa and 300-700 GPa, respectively. Some factors, such as defects on the surface or within the fibres, handling procedures and impurities in the nanofibre structure, affect the mechanical properties of CNFs [105].

The aspect ratio of nanofibres is another important property that influences the mechanical properties of composites. Since CNFs have diameters in nano-scale and lengths of up to several micro meters, their high aspect ratio enables them to distribute the load evenly throughout the matrix with no fibres pulled out of the matrix, and hence increase the reinforcement effects in composites [208, 209].

2.2.2.2. Thermal property

Thermal stability of CNFs depends on the atmosphere in which they are used. CNFs can be oxidized in air and great mass loss occurs above 400°C [210]. Graphitized CNFs show higher stability than un-graphitized CNFs. However, mass loss still occurs above 600°C. Therefore, if CNFs are to be used in atmospheric environments, their working temperature should be below 400°C. Thermal stability of CNFs also depend on the amount of impurities such as Na, Si, Mg and Ca [211]. Although these impurities may increase thermal conductivity, they lower the stability of CNFs at high temperatures [212].

2.2.2.3. Electrical property

Oxidation reduces the electrical conductivity of CNFs. Oxidation occurs when CNFs are exposed to air at high temperatures (>600°C) [213]. Electrical resistivity is increased by oxidation and higher initial resistivity leads to larger increase in resistivity [214]. Electrical conductivity depends greatly on the physical deformation or damage in CNF, because any change in the cross-section and length of the graphitic layers would reduce the conductivity [215].

Electrical properties are influenced by the production conditions. For instance, PAN based CNFs become semi-conductive when they are partially oxidized in air and heat treated at 700-900°C [216]. Electrical conductivity can be increased by optimizing the carbonisation stages of the precursor fibres [217].

As shown in Table 2.5, thermal and electrical properties of CNFs along the fibre axis and perpendicular to the fibre axis vary due to the anisotropic graphitic structure of the CNFs.

Property	Along the fibre axis	Perpendicular to the fibre axis	
Coefficient of thermal expansion 0–1000 °C (K ⁻¹)	3. 0 ×10 ⁻⁶	2. 8×10 ⁻⁶	
Electrical resistivity (Ω m)	1.1×10^{-3}	4. 1×10^{-3}	

Table 2.5. Properties of CNFs along and perpendicular to the fibre axis.

2.2.3. Applications

2.2.3.1. Biological applications

Any material that is going to be used inside human body has to be safe, non-toxic and non-reactive in acidic or basic environments [218]. Use of CNFs in biomedical applications such as tissue engineering, implant and wound healing has been studied and fibres with specific properties suitable for such end-uses have been successfully commercialized [219, 220]. For example, CNFs were used to reinforce calcium phosphate composites for bone replacement [221].

2.2.3.2. Electrical applications

Because of their high electric conductivity, CNFs can be used as an anode in batteries [222]. They have also been used in transmitters as electrodes [223]. CNFs are also used as an anode material in fuel cells due to their capacity to store hydrogen (Figure 2.8) [187].



Figure 2.8. Schematic illustration of the structure of a fuel cell [172].

2.2.3.3. Thermal applications

CNFs are used as heat insulating and fire resistant materials [224, 226]. They are mixed with phenol resin and carbonised at high temperatures (>1000°C). These products are used in aircraft seats as fire blocking materials in post-crash fires, insulators in aircraft body between the exterior and interior, and personal fire retardants [226-228].

2.2.3.4. Aerospace applications

Use of CNFs in aerospace industry is increasingly becoming diverse. They are now used in army and civil aircrafts and helicopters as composite reinforcement agents in nose, engine blades, interior body, exterior body, wings, stabilizers, etc (Figure 2.9) [229-231]. CNFs not only bring strength to the structure, but also reduce the overall aircraft weight by up to 1.5 tons. CNFs are also used in aerospace industry as a sound insulator.



Figure 2.9. Cabin of Proteus aircraft fabricated by Viper 7-axis fibre placement system [187].

2.2.5. Production

CNFs are produced using 2 main methods: (i) chemical vapour deposition (CVD) and (ii) pyrolysis.

2.2. 5. 1. Chemical vapour deposition (CVD)

In this method, CNFs are produced from carbonaceous gases such as acetylene, methane, natural gas, benzene and ethylene in the presence of catalytic particles [232]. The catalysts are mainly iron, nickel, copper, cobalt or palladium [233]. At accelerating temperatures carbonaceous gases are flown onto the catalytic particles which have certain diameters [234]. The diameters of CNFs are normally determined by the diameter of catalyst nanoparticles.

The growth process comprises endothermic and exothermic reactions which occur on the particles surface. Carbon atoms are adsorbed on the hotter side of the catalyst's surface in which the exothermic reaction occurs. Then a CF starts to grow on the cooler side of the catalyst particles where the endothermic reaction occurs (Figure2.10). This process continues and more and more carbon atoms are built up on top of each other in the form of benzene rings. The growth develops along the fibre axis as well as in the direction perpendicular to the fibre axis (fibre thickening) (Figure 2.11). Hence CVD methods normally results in the production of nanotubes, although CNFs have also successfully produced using CVD. As the fibre grows, the thickness of the fibres can be increased by increasing the gas concentration. The fibre growth can be stopped by reducing the gas concentration [236-239].



Figure 2. 10. CNF growth using CVD methods.





Figure 2.11. CNF lengthening [240].



Figure 2.12. TEM image of a CVD grown CNF [241].

CVD grown CNFs (Figure 2.12) have high mechanical strength due to their highly graphitized structure. They show modulus of >200 GPa and tensile strength of >2 GPa [242].

2.2.5.2. Pyrolysis

Pyrolysis comprises a series of heat treatment stages where each stage has a particular heating rate, maximum temperature and holding time [243, 244]. Mainly organic materials are pyrolysed for the production of activated carbon, CFs and CNFs [245-247]. The type of atmosphere used in the pyrolysis of various materials can be inert such as N₂ and Argon or oxidative such as O₂, depending on the desired properties of the resultant carbon material. However, the atmosphere and the gas flow rate should be carefully controlled during the pyrolysis [248, 249]. The molecular structure of precursors breaks down during the pyrolysis which leads to the formation of gaseous species, tar and a carbon rich residue (Figure 2.13) [250-253].

Chapter 2



Figure 2.13. Schematic illustration of the pyrolysis of organic materials.

For the production of CNFs by pyrolysis, there are two classes of precursor nanofibres that can be used. One is polymeric nanofibres made using electrospinning techniques. Another is CellNFs extracted from naturally occurring materials.

2.2.5.2.1. Pyrolysis of Electrospun nanofibres

Electrospinning was first invented in the early 20th century. But it was until the early 21st century when the real novelty and functionality of this technique was realized [254]. Electrospinning setup comprises a high voltage power supply, an electrically conductive polymer solution, an electrode collector and an injection tip (usually syringe needle) [255]. Electrically charged polymer solution is exposed to an electric field which leads to the formation of droplets at the tip of the needle. As a result, the electric field forces the droplet to draw towards the opposite electrode that also serves as a collector. As the droplet travels along the electric field, it is then exposed to the electric field resulting in the formation of filaments before hitting the electrode collector [256].

Electrospun nanofibres present excellent characteristics such as [257]:

- High surface area;
- High porosity;
- High mechanical and structural strength;
- Low weight;
- Ability to integrate with other materials;
- Ability to form different fibre diameter and length and chemical and mechanical properties.

Electrospun nanofibres are used in many applications such as filtration [258], catalyst support [259], biomedical [260-262] and sensors in electronics [263, 264].

CNFs can be produced from electrospun nanofibres by undergoing carbonisation in an inert atmosphere [54, 265, 266]. Zhou et al. used PAN-based electrospun nanofibre precursors to produce CNFs of 200-300 nm in diameter using pyrolysis in an inert gas [216]. They applied tension during the stabilization process which increased the structural order within the fibres. Thus, the pyrolysed CNFs were more graphitic than the ones with no applied tension. The CNFs produced by this method showed tensile strength of 300-600 MPa, modulus of 40-60 GPa and improved electrical conductivity. Figure 2.14 shows the typical TEM images of PAN-based electrospun carbon nanofibres.



Figure 2.14. TEM images of the PAN-based electrospun CNFs [216].

Nataraj et al. studied the use of various acids for the preparation of PAN nanofibres and their effects on the morphological and electrical properties of resulting CNFs [267]. They reported that the CNFs showed improved morphological and electrical properties by using heteropolyacids with 5% concentration.

CNFs inherit their properties from their precursors. For example, precursors with high tensile strength and modulus will produce CNFs with similar properties [216]. Thus, electrospun-based CNFs display very high surface area, low weight and high mechanical and structural strengths, because of the properties of electrospun PAN fibres. Electrospun

CNFs are used in many applications such as batteries, hydrogen storage, catalyst supports, nanocomposites and fuel cells [268, 269].

CNFs with smaller diameters, when used as a structural material in composites, enhance the integration between the matrix and the fibres and hence distribute the applied force more evenly throughout the network. As a result, the composite has a longer life span and higher mechanical strength. Typical electrospun-based CNFs have diameters of >100 nm [270, 271]. Smaller diameters are difficult to obtain mainly due to the limitations in electrospinning techniques where the attempt in producing finer diameter fibres leads to an increase in the formation of beads (defects) within the nanofibre network [266]. Electrospinning requires a polymer solution, such as chloroform and methylene chloride [272], in which the polymer is well dispersed. The evaporation of such solvents during fibre formation can cause health risks.

2.2.5.2.2. Pyrolysis of CellNFs

The pyrolysis of CellNFs is expected to result in the formation of CNFs. Due to their small diameter, cellulose-based CNFs may require lower temperature for graphitization [273]. However, little research has been reported on the production of nano-scale CFs by pyrolysis of CellNFs. It is expected that the molecular and morphological properties of precursors strongly affect those of the pyrolysed carbon material [203]. Ishida et al. investigated the carbonisation of freeze-dried bacterial and tunicate CellNFs and found that the carbon residue retained its fibrous morphology by using HCl as a pyrolytic atmosphere and a pretreatment chemical [149]. Another study showed that the effect of surface area and thermal stability of chitin and softwood CellNFs was considered to be the main factors in altering the fibrous morphology in the resultant carbon residues. However, these studies did not report successful retention of the fibrous structure of softwood-based CellNFs in the carbon residue.

Recently, Deakin University has successfully developed the technology to produce CellNF from natural plant products [274]. Mechanical ball-milling was used in order to obtain CellNFs. It is known that when cellulose fibres are placed in water, the hydroxyl groups (OH) of water competes with the ones in cellulose to form hydrogen bonding. This can help separate cellulose fibres from each other [275-277]. Ball-milling of cellulose in aqueous suspension breaks down the raw fibres into CellNFs suitable for CNF production. Successful fabrication of CellNFs using such green and scalable

technique was a major step toward CNF fabrication. Unlike CVD or electrospinning, pyrolysis does not require the precursor to undergo pre-treatment using hazardous chemicals. Once the optimum conditions are found, carbonisation of cellulose can be undertaken in inert atmosphere.

2.2.6. Thermal decomposition mechanism of cellulose

Cellulose undergoes thermal decomposition without going through the melting stage and thus it does not behave like other precursors such as PAN or pitch [78, 278, 279]. If the decomposition of cellulose was an idealistic dehydration reaction, then the following reaction would happen [10]:

$$(C_6H_{10}O_5)_n \to 6_nC + 5_nH_2O$$
 (2.1)

However, cellulose decomposition is a complex phenomenon comprising a series of reactions occurring in succession or concurrently, which eventually leads to mainly three products: (a) water, (b) gases and (c) char. There have been many studies investigating cellulose decomposition and theories developed explaining the phenomenon [149, 200, 1202, 279-284]. However, since cellulose itself has many physical forms in nature, the decomposition mechanism is influenced by its structural configuration. Hence it is difficult to obtain a general explanation of the decomposition mechanism. However, the current consensus can be summarised as follow [10]. The thermal decomposition of cellulose occurs in several stages:

• External dehydration (120–180°C)

When cellulose is heated to 120°C, physically absorbed water molecules are released (external dehydration). The hydrogen bonds between water molecules and OH groups in cellulose are replaced with the hydrogen bonding between OH groups in cellulose.

• Internal dehydration $(180 - 250^{\circ}C)$

As the temperature increases to around 180°C the energy is sufficient to excite some bonds within the cellulose molecules such as —OH and —H atoms ready to break the bonds [126, 127, 285, 286]. This causes dehydration from the cellulose glycosidic rings (internal dehydration) and elimination of water molecules from the rings [279]. Whether the OH and H from one glycosidic ring or from two neighbouring rings link together is still being debated.



Figure 2.15. Internal dehydration of cellulose.

• Initial chain scission (180–250°C)

At ~250°C, levoglucosan (1-6-anhydro- β -D-glucopyranose) is formed by an intra-chain dehydration reaction between an OH group in CH₂OH and the oxygen in 1-4 glycoside links (Figure 2.16) [275-277, 287]. Levoglucosan is volatile and, at higher temperatures, breaks down into tar and some other by-products such as char, CO₂ and furan [288, 289]. Tar is a brown sticky by-product of levoglucosan decomposition [290-296]. Levoglucosan is also responsible for the flammability of cellulose since its structural breakdown results in the formation of flammable gases. Hence, the prevention of levoglucosan formation will increase the yield of carbon residue [152-154, 160, 297-299]. The use of fire retardants such as ZnCl₂ [300-302] was studied in order to reduce the amount of levoglucosan that was produced during cellulose decomposition [52, 303-306].



Figure 2.16. Formation of levoglucosan.

• Depolymerization (250–400°C)

Above 250° C, the scission of glycosidic rings occurs that leads to the formation of CO, CO₂, H₂O and char (Figure 2.17) [157, 307, 308]. However, the scission is not a straight reaction but a series of other reactions leading to such a phenomenon. A glycosidic ring has four OH groups located at 2-, 3-, 4- and 6-C [309]. On one hand, the removal of these OH groups from the rings results in the formation of C=C and C=O bonds, while the chair/bed [112, 310] conformation of cellulose is still maintained [311-314]. On the other hand, levoglucosan is also formed at above 250° C. These two reactions are always concurrent and compete with each other [10].



Figure 2.17. Chain scission of cellulose and formation of gaseous species.

• Aromatization (>400°C)

Depolymerization is completed at 400°C and further increase in temperature initiates aromatization of CFs. At the completion of cellulose decomposition, carbon atoms connect with each other and form hexagonal rings similar to graphite. However, the paths in which carbon atoms connect and form rings are not fully understood. It was suggested that longitudinal polymerization occurs where carbon atoms join to form carbon chains and chains connect to form rings (Figure 2.18). However, this model has little experimental support [317]. Davidson [318] suggested that increasing temperature causes the carbon chains to get closer to each other and join to form rings. Losty [159, 319] suggested that elimination of pyranose rings and glycoside links create aromatic rings with hydrogen attached to them. Attached hydrogen atoms act as isolators and keep the rings separate from each other [158]. The rings then connect and form graphitic layers at higher temperatures.



Figure 2.18. Aromatization of chains and formation of graphitic carbon.

2.3. Summary

CNFs are a new class of materials with many new applications from which various industries from automobile and aerospace to electronics and medicine can greatly benefit. While micron sized CFs are widely used, CFs with diameters in nanometer range exhibit improved properties such as mechanical strength and electrical conductivities. In addition, CNFs have high aspect ratios and high specific surface areas, which enable them to greatly interact with other materials and distribute the desired function evenly across the host material. In the past, some studies have been conducted on the development of new techniques to produce CNFs and utilisation of various precursors including natural polymers such as cellulose. The common methods to produce CNFs are CVD, electrospinning and pyrolysis techniques. CVD uses carbonaceous gases as a precursor and metallic catalysts as a platform to grow CNFs. Electrospinning produces precursor nanofibres using charged polymeric solutions and an electric field initiating the fibre-spinning process, and the resulting polymer fibres are subsequently carbonised. Both CVD and electrospinning have major drawbacks. CVD grown CNFs often contain a high percentage of impurities. Electrospinning methods result in precursor nanofibres with diameters larger than 100 nm. In addition, the cost of CVD and electrospun CNFs is high due to slow and rather difficult production processes. The environmental concerns of CVD and electrospun CNFs, due to the involvement of hazardous chemical treatments, limit the horizons for the commercial development of such techniques.

Carbonisation of cellulose using pyrolysis has some advantages:

- 1. Cellulose is a renewable material which is widely available on earth and hence the cost of the precursor is low. The environmental benefit is high, as it does not rely on petrol-based raw materials;
- 2. Precursor preparation does not require hazardous chemical treatments, which add another environmental benefit;
- 3. Pyrolysis does not require facilities as complex as for CVD;
- 4. Pyrolysis of plant-based cellulose is theoretically carbon-neutral and hence has reduced environmental impacts compared with CVD and electrospinning techniques;
- 5. The pyrolysis process is relatively easy to scale up for commercial purposes.

Although the pyrolysis of bacterial cellulose or chitin-based celluloses has been investigated in the past, the use of plant-based CellNFs as a precursor has rarely appeared in the literature. This research was conducted to fill this knowledge gap by investigating the potential to produce CNFs by pyrolysis of plant-based CellNFs.

Chapter 3 Chapter 3: Experimental

In this chapter, the details of the materials, instruments, experimental procedures and characterisation methods used in this research work are presented.

3.1. Cellulose nanofibre preparation

3.1.1. Overview

The fabrication procedure of CellNFs has been detailed elsewhere [274]. Briefly, softwood pulp was suspended in deionised water, blended in a mixer and stirred for 48 h. The pulp suspension was then ball-milled for 90 min using zirconia balls. The suspension was then dried (using oven-drying or freeze-drying techniques) for carbonisation.

3.1.2 Cellulose pulp suspension

Cellulose pulp (NIST standard material RM 8495 Northern Softwood Bleached Kraft Pulp) was kindly donated by the Australian Pulp and Paper Institute in Monash University. Cerium-doped zirconium oxide balls (Zirconox 0.4-0.6 mm in diameter) were purchased from Jyoti Ceramic Industries Pvt. Ltd.

Cellulose pulp sheeting was cut into 5×5 cm pieces and soaked in deionised water in the fridge overnight. The weight ratio between the pulp and water was 99:1. Wetted cellulose was then defibrillated using a conventional kitchen blender and then stirred at 70° C overnight. The cellulose suspension, thus obtained, was kept in the fridge and used as required.

3.1.3. Ball milling

In order to mill the pulp suspension, 45 g of cerium-doped zirconia balls and 20 mL of deionised water were added to 20 g of 1wt% pulp suspension in a 70 ml polypropylene container. Ball-milling was performed in a Spex 8000M shaker mill for 90 min (Figure 3.1). The CellNF suspension was then filtered to remove balls and clotted fibres using a polyethylene terephthalate mesh with a mesh opening of 125 μ m.



Figure 3.1. Spex 8000M shaker mill.

3.1.4. Drying

For oven drying, the CellNF suspension was poured into a purpose-made aluminium foil container and placed in a laboratory oven at 60°C overnight. A translucent CellNF film was collected and used for carbonisation and characterisation.

For freeze-drying, a small amount of CellNF suspension was poured in a flask and dipped and stirred in liquid nitrogen. The CellNF suspension was then vacuum dried using a Labconco freeze-drier for 72 h. CellNFs with a spongy appearance were collected and used for carbonisation and characterisation.

3.2. Carbonisation

3.2.1. Below 1000°C

A pyrolysis method was selected to carbonise CellNFs. A tubular furnace (TF55035C-1, Lindberg/Blue M) were used with a quartz tube with the length of 68 cm and diameter of 2.7 cm (Figure 3.2). A small piece of sample (~ 15 mg) was placed on an alumina boat of $85 \times 10 \times 13$ mm inside the quartz tube. A nitrogen gas flow at the rate of 70 ml/min was introduced into the tube to create an inert atmosphere and also to constantly remove the evolved gases from the system. Higher flow rates would blow the samples off the tube while lower flow rates would not remove the by-product gases and water from the system. The furnace temperature was accurately calibrated so that the temperature variation was negligible. The heating elements were aligned along and around the tube hole so that a uniform heating area was facilitated for accurate sample heating. The nitrogen flow rate, heating rate, maximum temperature and holding time were controlled throughout the experiment.



Figure 3.2. Tube furnace, TF55035C-1, Lindberg/Blue M.

3.2.2. Above 1600°C

Graphitization experiments at temperatures above 1600°C were conducted using a Tanmann furnace (Figure 3.3) in the laboratory of Professor Toyoda (Applied chemistry, Faculty of Engineering, Oita University). The heat treatment was carried out under vacuum in a graphite container. The heating rate was fixed to 10°C/min.



Figure 3.3. Tanmann furnace, school of applied chemistry, faculty of Engineering, Oita University.

3.3. Characterisation of nanofibres

3.3.1. Scanning Electron Microscopy (SEM)

The morphological properties of fibres were studied with scanning electron microscopy (SEM, SUPRA 55P). Unlike optical microscopes that use light for imaging, SEM microscope uses electrons to interact with the sample and produce images of the sample surface. An electron beam sends out electrons to the sample. These electrons then interact with the atoms which make up the sample and send back electrons with specific energies (secondary electrons). These electrons are then collected and used for producing images containing information about the surface morphology of the samples. The low energy of secondary electrons allows them to escape only from near the sample surface (a few nm) and hence SEM is an excellent method to analyse surface morphology of nano-scale objects.

A small amount of sample was placed on a sample holder using carbon tape and was sputter coated with gold or carbon, using a Baltec SCD50 sputter coater. Due to the sensitivity of samples to high voltage, all images were taken at an acceleration voltage of $\sim 1 \text{ kV}$ with magnifications of 5,000-10,000. The SEM images were used to calculate the average fibre diameters using an image analysis software (Image Pro-Plus 4. 5, Media Cybernetics Co., Santa Clara, CA).

For the analysis of fibre diameters, SEM images were taken from randomly selected spots on the sample. All the fibers in one image were measured using Image Pro Plus software. Since all the fibres in one image were measured, the bias in the selection of fibres was minimal. On average 7-15 images were taken from one sample and fiber diameter of all the fibers in the images were measured.

In this research, different magnification was used for different samples to estimate the fibre diameter. The magnifications of SEM images were selected based on the limitations arising from the sample damage caused by the electron beam. Higher magnification means higher concentration of electrons in one spot. It was noticed that, under high magnifications, cellulose fibers could be damaged by the electron beam before being measured. The degree of damage varied among the samples even using the same magnification, which reason should be further investigated. Therefore, different

magnifications were chosen depending on the sample's characteristics to tolerate electron beams.

3.3.2. Transmission Electron Microscopy (TEM)

TEM observations of the samples were undertaken on a JEM-2100 TEM. Similar to SEM technique, in TEM microscopy an electron beam sends out electrons to a thin sample (< 100 nm in thickness). The electrons interact with the sample atoms as they pass through the thickness of the sample. Electrons that were not scattered by the sample were transmitted through and were collected for producing images containing information about crystalline, chemical and electronic properties of the sample. Because of the very small wavelength of electrons (~2.5 pm for 200 keV) compared to visible light (400 – 750 nm), TEM enables the observation of nanoscale objects.

All the samples were ultrasonically dispersed in ethanol and then a drop of the dispersion was dried on a copper grid covered with a carbon film. TEM images were observed through a Gatan image filter at an accelerating voltage of 200 keV.

3.3.3. Fourier Transform Infrared Spectroscopy (FT-IR)

Fourier transform infrared (FT-IR) spectra were recorded by an FT-IR spectrophotometer (Bruker Vertex 70) to study the chemical configuration of the samples. Fourier transform infrared spectroscopy is a technique to analyse the chemical and crystalline properties of the sample. It uses a beam of infrared light that hits the sample in a wide range of wavelengths and measures that how much of the light is absorbed/transmitted. The atomic bonds that make up the sample absorb a particular wavelength of light depending on the characteristics of the chemical bonds. Hence the information about the types of chemical bonding that are present in the sample is recorded in the spectrum.

The spectra were obtained under the condition of 64 scans per spectra at 4 cm⁻¹ resolution. All test runs were undertaken in a controlled environment $(20\pm2^{\circ}C)$ and $65\pm2\%$ relative humidity). The data were analysed using OPUS 5.5 software. Samples pallets were produced using the KBr method. The samples were diluted and evenly distributed in the KBr disks so that appropriate effective thickness was achieved not to saturate the FTIR signal.

3.3.4. X-ray Diffraction (XRD)

The crystallinity of the samples was studied by X-ray diffraction analysis. X-ray diffraction technique is used to analyse the crystal structures of materials. When a monochromatic X-ray beam is projected onto a crystalline sample, X-ray diffraction occurs only in certain diffraction angles, depending on the crystal structure of the sample. Most atoms scatter X-ray through destructive interference in most directions whereas atoms that are in one lattice plane scatter the X-ray in a certain angle through constructive interference (Bragg's law). The constructive interference leaves high intensity spots (or peaks) on the detector which is then used to analyse the crystal properties such as crystal symmetry, crystallite size and crystallinity. It can also be used to analyse chemical compositions by identifying the crystalline phases of known materials in the observed diffraction patterns.

An X-ray diffraction instrument (Panalytical X'Pert PRO MRD XL) was used to obtain the curves at 2 Θ between 5° to 80°, step size of 0.01, time per step of 3 sec, and Cu ka (λ =1.54Å) radiation generated at 40 kV and 30 mA. All tests were undertaken at room temperature and atmospheric pressure. Samples were put on glass holders and flattened before measurement.

Crystallite sizes were calculated using the Bragg-Scherrer equation:

$$D = \frac{K\lambda}{\beta \times \cos\theta} \tag{3.1},$$

where θ is the Bragg angle in radian and λ is the wavelength of the incident radiation, 0.154 nm. K is the Scherrer constant, 0.94, and β is the diffraction peak's full width at half maximum in radian.

The degree of crystallinity was estimated using the Segal method that is frequently used for cellulose [320]:

$$I_{\rm cr} = \frac{I_{\rm max} - I_{\rm min}}{I_{\rm max}} \times 100$$
 (3.2),

where I_{max} is the maximum intensity of the peak at 22.5° and I_{min} is the peak intensity of the amorphous fraction represented by the diffraction intensity at 19°.

Interplanar spacing, d, is the distance between adjacent planes in the set (hkl) crystal plane. The cellulose I β structure has a monoclinic $P2_1$ unit cell containing two non-equivalent chains. The periodic chains run parallel to each other forming single planar sheets that stack into 3D crystals [321].

3.3.5. Thermal Gravimetry/ Differential Scanning Calorimetry (TG/DSC)

Thermal properties of the samples were investigated with TG/DSC using a Netzsch 407 PC Luxx instrument with the NETZSCH Proteus analysis software. In thermal gravimetry analysis (TGA), the changes in weight are measured to analyse the thermal behaviour/reaction of the sample. This requires precise measurements in weight, temperature and temperature change. By heating the sample to above decomposition temperature, at least one component decomposes into gas and is released from the sample by which the weight change can be measured. The data give information about the thermal characteristics of the material. In Differential Scanning Calorimetry (DSC) analysis, the difference in the required amount of heat energy to increase the temperature of a sample and reference is measured. DSC allows the detection of phase changes as the appearance of peaks associated with endothermal or exothermal reactions, or the change in specific heat as the change in the slope of temperature-heat curve.

Samples of 5-10 mg were placed in an alumina crucible and measured at the heating rate of 10°C/min in a nitrogen atmosphere. The measurements were carried out in the temperature range between 30°C-1000°C under atmospheric pressure.

3.3.6. Raman spectroscopy

Raman spectroscopy reveals the vibrational mode of atoms and molecules using a monochromatic light. The shift in the wavelength of detected light from the wavelength of incident light, Raman shift, is associated with the energy of atomic or molecular vibrations/ rotations that are excited by the incident light. The vibrational/rotational energy is specific to chemical bondings and molecular structures. For example, the peaks at 1350and 1580 cm⁻¹ in Raman spectra of carbonadoes materials represent the vibration in defective graphitic structure and the tangential vibration of carbon atoms in graphitic layers, respectively. By comparing these two peaks, the extent of carbonisation can be analysed.

Raman spectra were obtained using an InVia Confocal microscope system (Renishaw, Gloucestershire, UK) with 514 nm excitation from a Modu-Laser Stellar-Pro ML/150 argon ion laser through a X20 (0.4 na) objective. Nanofibre samples were packed into a 2 mm cavity cell held on a 250 mm \times 750 mm plate. Incident laser power was 1.1 mW and coaxial backscatter geometry was employed. Use of higher laser power was found to cause decomposition of the nanofibres. Laser power was measured at the sample using an Ophir Nova power meter fitted with a PD300-3W head. Spectra were collected over the range 3200 cm⁻¹ to 200 cm⁻¹ and averaged over at least 5 scans, each with an accumulation time of 40 s. Signal to noise was further reduced by averaging a minimum of two spectra obtained from different areas of the sample. These spectra were found to be highly reproducible. The Raman shifts were calibrated using the 520 cm⁻¹ line of a silicon wafer. The spectral resolution was ~ 1 cm⁻¹.

<u>Chapter 4: Carbonisation of oven-</u> dried cellulose nanofibre

4.1. Introduction

In this chapter, the pyrolysis of oven-dried CellNFs under atmospheric pressure and inert environment was investigated using controlled heating conditions. An oven-drying method was selected as the first drying technique to investigate its suitability for the production of carbon nanofibres, as the methods is simple and easily scalable. The oven drying method leaves a compact network of nanofibres due to strong hydrogen bonding between nanofibres. The effect of heat treatment conditions on the properties of carbonised residue was investigated using TGA, SEM, XRD and FT-IR.

4.2. Experimental procedure

4.2.1. Carbonisation equipment

The pyrolysis of oven-dried CellNFs was carried out using the tube furnace as described in Chapter 2. In order to study the pyrolysis effect on the original CellNFs and for the experimental accuracy, one batch of oven-dried CellNFs was split into two identical batches. Small pieces of one batch of oven-dried CellNFs ($1cm \times 1 cm$) were placed on the boat and placed in the middle of a quartz tube. The other batch was used for comparison with the pyrolysis results.

4.2.2. Carbonisation conditions

As discussed in chapter 2, it has been reported that the pyrolysis of cellulose comprises four stages: dehydration of physically absorbed water (25°C - 180°C), inter and intra molecular dehydration (180-250°C), depolymerisation (240 -350°C) and aromatization (over 350°C). Therefore, separate investigations in these temperature ranges are crucial to understand the pyrolysis characteristics of oven-dried CellNF.

Heating the CellNFs above 180° C is expected to cause the internal dehydration from the cellulose molecules. H₂O is formed by connection of and OH group and hydrogen in the

cellulose molecules. Cellulose dehydration continues up to 250°C where the major stage in decomposition is initiated. It was suggested that if hydrogen and oxygen were completely released from the cellulose molecules prior to the decomposition stage, the interference of these elements with carbon atoms would be minimum during cellulose decomposition. This has two advantages: 1) less carbon atoms are released from the molecules in the form of gas, so that less mass loss occurs, 2) upon completion of the decomposition, presence of oxygen and hydrogen complexes in the carbon residue would be minimal. Thus, holding CellNFs below 180°C would affect the CellNF properties and the pathways in which cellulose converts into carbon. The carbon residue would retain its original cellulosic morphology. In order to study such effects, three holding times of 1, 2 and 3 h at 170°C, a slightly lower temperature than 180°C where the internal dehydration starts, were investigated. Likewise, heating conditions up to 240°C, a slightly lower temperature than 250°C where the decomposition starts, were investigated.

Table 4.1 shows the heat treatment conditions selected for carbonisation of oven-dried CellNFs. Past studies showed that higher heating rates leads to higher mass loss and formation of tarry by-products whereas lower heating rates prohibits such results, for conventional micron sized cellulose [146]. Hence, different heating rates of 1°C/min, 2°C/min and 10°C/min, were selected in this temperature range.

Sample ID		T temp	arget erature 1		Target temperature 2			Target temperature 3
	Heating rate (°C/min)	Temp. (°C)	Residence time (h)	Heating rate (°C/min)	Temp. (°C)	Residence time (h)	Heating rate (°C/min)	Temp. (°C)
NF1	1	170						
NF2	2	170						
NF3	10	170						
NF4	10	170	1					
NF5	10	170	2					
NF6	10	170	3					
NF7	10	170	1	1	240			
NF8	10	170	1	2	240			
NF9	10	170	1	10	240			
NF10	10	170	1	1	240	1		
NF11	10	170	1	1	240	2		
NF12	10	170	1	1	240	3		
NF13	10	170	1	1	240	3	1	400
NF14	10	170	1	1	240	3	2	400
NF15	10	170	1	1	240	3	10	400

Table 4.1. Heat treatment conditions.

4.3. Results and discussion

4.3.1. Characteristics of oven-dried CellNFs before heat treatment

4.3.1.1. Morphology

Figure 4.1 (a) shows the SEM image of the oven-dried CellNFs. Oven-drying left a very compact film of fibres and little spatial gaps exist between the fibres. This suggested that cellulosic chains from neighbouring nanofibres could form hydrogen bonding and joined together. Figure 4.1(b) shows the fibre diameter distribution analysed from SEM images.

The fibre diameter distribution is in good agreement with the previous report by Zhang et al. [274].



Figure 4.1. (a) SEM image of the oven-dried CellNFs and (b) fibre diameter distribution analysed from SEM images.

4.3.1.2. Crystallinity

Figure 4.2 shows the XRD graphs of the oven-dried CellNFs and softwood pulp. It is shown that the characteristic peak of cellulose occurs at 22.5° , indicating that the cellulose-I crystal structure was maintained after ball-milling and drying. CellNFs show main peaks at 15° , 16° , 22.5° and 34.1° associating with the (110), (110), (200) and (004) lattice planes.



Figure 4.2. XRD graphs of softwood pulp and oven-dried CellNFs.

Table 4.2 shows the crystalline properties of softwood pulp and oven-dried CellNFs. Preparation of CellNFs including ball-milling did not drastically change the crystal properties.

	2 theta	hkl	d	D _{cr}	$I_{cr}(\%)$
	(deg)		(Å)	(nm)	
Softwood	16. 1 °	110, 1-10	5.4		
pulp	22. 5 °	200	4.1	3.6	73
	34. 1 °	004	2.6	_	_
Oven-	16. 1 °	110, 1-10	5.5	—	—
dried	22. 5 °	200	3.9	3.5	71
CellNFs	34. 1 °	004	2.6	_	_

Table 4.2. Crystal size D, index of crystallinity I_{cr}, and interplanar distances d, of raw pulp and oven-dried CellNFs, estimated from XRD patterns.

4.3.1.3. Chemical bonding

Figure 4.3 shows the FT-IR spectra of softwood pulp and oven-dried CellNFs. The spectra are identical with each other which indicate that CellNF preparation did not change the molecular structure of CellNFs.



Figure 4.3. FT-IR spectra of softwood pulp and oven-dried CellNFs.

4.3.1.4. Thermal properties

Figure 4.4 shows the thermal behaviour of the oven-dried CellNFs when heated with a rate of 10° C/min up to 600° C. The chemical breakdown process started from around 250°C and completed at ~360°C. From the TG graph, it was evident that the sample lost 80% of its original weight. From the DSC graph, it was noted that a peak occurred at 353°C and indicates that the decomposition stage is an endothermic process.

Chapter 4



Figure 4.4. TGA/DSC graphs of CellNFs.

4.3.2. Effects of heating conditions below 170°C

4.3.2.1. Morphology

Figure 4.5 shows the SEM images of the samples heated to 170°C with heating rates of 1, 2, and 10°C/min. The fibrous morphology was retained during the heat treatment under the three heating rates. The macroscopic morphology of the resultant materials was similar to the original CellNFs (Figure 4.1). It is also evident from the SEM images that nanofibres were greatly compacted after heating and little space was remained between the fibres.



Figure 4.5. SEM images of heat treated samples to 170°C with heating rate of 1°C/min (NF1), 2°C/min (NF2) and 10°C/min (NF3).

4. 3. 2. 2. Crystallinity

Figure 4.6 Shows the XRD patterns of the samples heat treated with the three heating rates. The position of the main peak at ~ 22.5° did not change with the heating rate. However, the peak at ~ 15° shifted to a slightly higher angle in the samples NF1 and NF2 with low heating rate (1 and 2° C/min respectively). This may suggest the formation of a disordered state of cellulose-I β structure [322], but further detailed investigation is required to confirm the structural change during heat treatment at such a low temperature range.



Figure 4.6. XRD graphs of heat treated samples with heating rates of 1°C/min (NF1), 2°C/min (NF2) and 10°C/min (NF3).

Table 4.3 lists the estimated crystal size D, index of crystallinity I_{cr} , and interplanar distances d, after heat treatment. It is evident that heating the CellNFs up to 170° C did not affect the crystallinity and crystallite size (see Tables 4.2 and 4.3). Hence, the crystal structure of CellNFs showed little change up to this temperature irrespective of heating rate.

Heating	2 theta	hkl	d	D _{cr}	Icr
rate	(deg)		(Å)	(nm)	(%)
1°C/min	16. 1 °	110, 1-10	5.4		—
	22. 5 °	200	3.9	3.7	73
	34. 1 °	004	2.6		—
2°C/min	16. 1 °	110, 1-10	5.7		—
	22. 5 °	200	4.0	3.6	70
	34. 1 °	004	3.3		—
10°C/min	16. 1 °	110, 1-10	5.8		—
	22. 5 °	200	3.9	3.7	71
	34. 1 °	004	2.3		

Table 4.3. Crystal size D, index of crystallinity I_{cr}, and interplanar distances d, of heat treated samples with various heating rate, estimated from XRD patterns.

4.3.2.3. Chemical structure

Figure 4.7 shows the FT-IR spectra of CellNFs heated to 170°C with the three heating rates. Characteristic peaks of cellulose occur between 1000 cm⁻¹ to 1500 cm⁻¹ as shown in Figure 4.7(b). These peaks were assigned to the chemical bonds in the cellulose molecules (Table 4.4). Figure 4.7 shows that these three spectra are identical and no peak shift was observed. This indicates that heating to 170°C does not alter the chemical configuration of CellNFs and that the heating rate does not influence the chemical bondings in CellNFs.



Figure 4.7. FT-IR spectra of heat treated CellNFs with heating rates of 1°C/min (NF1), 2°C/min (NF2) and 10°C/min (NF3); (a) 4000-800cm⁻¹, (b) 2000-800cm⁻¹.

Wavenumber (cm ⁻¹)	Chemical bond
1112	C-OH
1163	C-O
1227	O-H
1277	C-H
1315	CH ₂
1336	O-H
1365	C-H
1428	CH ₂

Table 4.4. Assignment of FT-IR peaks to chemical bonds.

4.3.3. Effect of holding time at 170°C

4.3.3.1. Morphology

Figure 4.8 shows the SEM images of the CellNFs held at 170° C for 1, 2 and 3 h, respectively. These images indicate that the morphological properties remained intact although dehydration and formation of H₂O from the cellulose molecule have already

started (180°C -250°C) (see the TG/DSC data Figure 4.4). The average fibre diameter before and after each experiment was calculated from the SEM images and are shown in Figures 4.1(b) and 4.9. The Average fibre diameter did not show remarkable changes before and after heat treatment at 170°C for various holding times. This suggests that major decomposition process of cellulose was not initiated at this temperature.



Figure 4.8. SEM images of heat treated samples at 170°C, with holding times of 1 h (NF4), 2 h (NF5) and 3 h (NF6).



Figure 4.9. Fibre diameter distribution (average of three samples) of NF4, NF5 and NF6 estimated from SEM images.

4.3.3.2. Crystallinity

Figure 4.10 Shows the XRD graphs of CellNF samples heated to 170° C and held for various times. These three XRD patterns of heat treated samples show similar patterns to that of CellNFs before heat treatment. The position of the main peak at ~ 22.5° did not change with heating rate. However, the peak at ~ 15° shifted to a slightly higher degree in the samples NF4 and NF5 with long holding times (2 and 3 h respectively). This may suggest the formation of a disordered state of cellulose-I β structure [322], but further detailed investigation is required to confirm the structural change during heat treatment at such a low temperature range.



Figure 4.10. XRD graphs of heat treated samples at 170°C with holding times of 1 h (NF4), 2 h (NF5) and 3 h (NF6).

Holding	2 theta	hkl	d	D _{cr}	$I_{cr}(\%)$
time	(deg)		(Å)	(nm)	
3 h	16.1 °	110, 1-10	4.3	_	
	22.5 °	200	3.8	3.8	72
	34.1 °	004	2.9	—	—
2 h	16.1 °	110, 1-10	5.1	—	—
	22.5 °	200	4.3	3.5	70
	34.1 °	004	3.3	—	—
1 h	16.1 °	110, 1-10	5.5	—	—
	22.5 °	200	3.9	3.9	72
	34.1 °	004	2.9	—	—

Table 4.5. Crystal size D, index of crystallinity I_{cr}, and interplanar distances d, of heat treated samples with various holding times, estimated from XRD patterns.

Table 4.5 lists the estimated crystal size D, index of crystallinity I_{cr} , and interplanar distances d, after heat treatment. It is noted that the crystalline properties of heat treated

CellNFs did not change dramatically. This indicates that heat treatment at 170°C has little effect on the structure of the material irrespective of holding time.

4.3.3.3. Chemical structure

Figure 4.11 shows the FT-IR spectra of CellNFs heat treated at 170°C for 1, 2 and 3 h. It is noted that the FT-IR spectra of heat treated cellulose resemble the one before heat treatment (Figure 4.3) indicating that external dehydration, not internal one, was mainly taking place at this temperature.



Figure 4.11. FT-IR spectra of heat treated CellNFs with holding times of 1 h (NF4), 2 h (NF5) and 3 h (NF6).

4.3.4. Effect of heating rate between 170°C to 240°C

4.3.4.1. Morphology

Figure 4.12 shows the SEM images of the samples after heating up to 240°C with various heating rates. These images show that fibrous morphologies from different heating rates resemble the original ones. Between 170°C to 240°C internal dehydration mainly takes place which comprises connection of OH groups with H atoms in the cellulose molecules and formation of water molecules.



Figure 4.12. SEM images of samples heat treated to 240°C, with heating rates of 1°C/min (NF7), 2°C/min (NF8) and 10°C/min (NF9).

4.3.4.2. Crystallinity

Figure 4.13 shows the XRD patterns of the samples before and after heating up to 240°C. The sharp peak at 22.5° indicates that the cellulosic structure still remained at 240°C and that the difference in heating rate in the temperature range between 170°C and 240°C does not influence the crystal structure.



Figure 4.13. XRD patterns of heat treated CellNFs, with heating rates of 1°C/min (NF7), 2°C/min (NF8) and 10°C/min (NF9).

Table 4.6. Crystal size D, index of crystallinity Icr, and interplanar distances d, of heat treated samples with various heating rates between 170°C to 240°C, estimated from XRD

Heating	2 theta	hkl	d	D _{cr}	$I_{cr}(\%)$
rate	(deg)		(Å)	(nm)	
1°C/min	16.1 °	110, 1-10	3.8		—
	22.5 °	200	3.6	3.1	70
	34.1 °	004	2.7	—	—
2°C/min	16.1 °	110, 1-10	4.3	—	—
	22.5 °	200	5.1	2.9	70
	34.1 °	004	3.7	—	—
10°C/min	16.1 °	110, 1-10	5.1	—	—
	22.5 °	200	3.8	3.7	71
	34.1 °	004	2.6		_

patterns.

4.3.4.3. Chemical structure

Figure 4.14 shows FT-IR spectra of the samples heat treated to 240°C. The FT-IR spectra of the samples were identical to that of CellNFs before heat treatment, except the 1700 cm⁻¹ band which was not identified and is a subject of further study. 1112 cm⁻¹ and 1165 $\mbox{cm}^{\text{-1}}$ bands assigned to \mbox{CH}_2 and \mbox{OH} groups were still present in the FT-IR spectra. The results indicate that cellulosic chemical configuration seemed to have not changed by the heat treatment regardless of heating rate.



Figure 4.14. FT-IR spectra of heat treated samples, with heating rates of 1°C/min (NF7), 2°C/min (NF8) and 10°C/min (NF9).

4.3.5. Effect of holding time at 240°C

4.3.5.1. Morphology

Figure 4.15 shows the SEM images of oven-dried CellNFs held at 240°C for 1, 2 and 3 h. It is noted that CellNFs still maintained their fibrous morphologies. The average fibre diameter was measured for three heat treated samples separately and averaged in order to analyse the fibre diameter after heat treatment. The average fibre diameter was 33.58 nm which is identical the oven-dried CellNFs (Figure 4.16).



Figure 4.15. SEM images of samples held at 240°C, with holding times of 1 h (NF10), 2 h (NF11) and 3 h (NF12).


Figure 4.16. Fibre diameter distribution (average of NF-10, NF11 and NF12) estimated from SEM images.

4.3.5.2. Crystallinity

Figure 4.17 shows the XRD results for the samples held at 1, 2 and 3 h. The peak at 22.5° was evident in all samples indicating that the cellulose-I crystal structure was retained [145].



Figure 4.17. XRD graphs of samples heat treated at 240°C, with holding times of 1 h (NF10), 2 h (NF11) and 3 h (NF12).

Holding	2 theta	hkl	d	D _{cr}	$I_{cr}(\%)$
time	(deg)		(Å)	(nm)	
3 h	16.1 °	110, 1-10	4.3	—	—
	22.5 °	200	4.2	3.7	70
	34.1 °	004	3.1	—	—
2 h	16.1 °	110, 1-10	5.3	—	—
	22.5 °	200	3.5	2.9	68
	34.1 °	004	3.8	—	—
1 h	16.1 °	110, 1-10	5.4	—	—
	22.5 °	200	3.3	3.9	69
	34.1 °	004	2.4		

Table 4.7. Crystal size D, index of crystallinity I_{cr}, and interplanar distances d, of heat treated samples with various holding time at 240°C, estimated from XRD patterns.

4.3.5.3. Chemical structure

Figure 4.18 shows FT-IR spectra of heat treated samples heat treated at 240°C. The FT-IR spectra show rather similar spectra to that of freeze-dried CellNFs before heat treatment. Cellulosic peaks in the spectra were identical for the three different holding times. No major change in the band position or disappearance of bands from the spectra was recorded. The result indicates that the difference in holding time at 240°C did not influence the chemical structure of cellulose. This is in agreement with the XRD results (Figure 4.17) indicating that the crystalline structure of the cellulose was also maintained.



Figure 4.18. FT-IR spectra of heat treated samples, with holding times of 1 h (NF10), 2 h (NF11) and 3 h (NF12).

4.3.6. Effect of heating rate between 240°C to 400°C

4.3.6.1. Morphology

Figure 4.19 shows the SEM images of the samples heat treated at 400°C. It is shown that the fibrous morphology was greatly damaged and a film-like structure was formed. Some traces of fibres were evident in the sample heated with the heating rate of 1°C/min. It is also shown that heating rate of 10°C/min resulted in agglomerated particles on the surface.



Figure 4.19. SEM images of heat treated samples to 400°C, with heating rates of 1°C/min (NF13), 2°C/min (NF14) and 10°C/min (NF15).

4.3.6.2. Crystallinity

Figure 4.20 shows the XRD patterns of the samples when heated to 400° C. The diffraction peak at 22.5° that is associated with cellulose-I structure disappeared. This indicates that the resultant material was amorphous.



Figure 4.20. XRD graphs of samples heat treated at 400°C, with heating rates of 1°C/min (NF13), 2°C/min (NF14) and 10°C/min (NF15).

4.3.6.3. Chemical structure

Figure 4.21 shows the FI-TR spectra of the samples heated to 400° C. Cellulosic characteristic bands were no longer observed in these samples. The peaks at 1603 cm⁻¹ and 1703 cm⁻¹ were assigned to C=O and C=C groups. The formation of C=C is the first step in progressing into graphitization at higher temperatures. However, the XRD results indicate that no long range atomic order associated with the graphene structure was formed at this temperature.



Figure 4.21. FT-IR spectra of samples, with heating rates of 1°C/min (NF13), 2°C/min (NF14) and 10°C/min (NF15); a) 4000-800cm⁻¹, (b) 2000-800cm⁻¹.

4.4. Conclusion

In this chapter, the effect of low-temperature heat treatment of oven-dried CellNFs on the morphology, crystal structure and chemical bonding was studied. Cellulose decomposition comprises 3 main stages: (i) physical dehydration from the surface, (ii) internal dehydration from the cellulose molecule and (iii) the breakdown of the glycosidic rings into volatile gases and water. The effects of pyrolysis conditions such as heating rate, holding time and maximum temperature in each decomposition stage were investigated.

Oven-drying left a compact network of cellulose fibres which connected to each other via hydrogen bonding. Between 25°C to 170°C, neither the heating rate nor the holding time altered the cellulosic structure. The glycosidic rings still maintained the chain-like structure of cellulose alongside with the strong hydrogen bonding.

However, XRD study showed that holding the oven-dried CellNFs at 170°C altered the crystal structure to cellulose-Iβ. Further study is required to confirm the phase transition, preferably by using synchrotron radiation. The change in crystal phase, however, did not break the glycosidic rings and the fibrous structure.

Furthermore, the oven-dried CellNFs samples showed little response to heat treatment to 240°C. Various heating rates did not change the cellulosic chemical or crystal structures; hence the samples maintained their fibrous morphology. Holding time at 240°C did not cause changes in cellulosic chemical and crystalline properties. The nanofibres retained their original fibrous morphology after various holding times.

Heating the samples up to 400°C caused cellulosic rings to break down. As a result, a carbon rich residue was formed. The morphological properties were independent of heating rate from 240°C to 400°C. Furthermore, the crystal and chemical properties of the samples were not influenced by heating rates.

Upon carbonisation, the original fibrous morphologies were lost. It appeared in the SEM images that the oven-dried CellNF sheets were transformed into sheet-like carbon residues. This may be caused by the fact that the distance between CellNFs was close enough to form bridging between fibres during carbonisation. As will be investigated in the next chapter, different drying technique resulting in wider spatial distance between fibres may results in different morphological characteristics of residual carbon.

<u>Chapter 5: Carbonisation of freeze-</u> dried cellulose nanofibre

5.1. Introduction

In the previous chapter, it was shown that oven-drying of CellNFs aqueous suspensions results in a compact network of nanofibres which were closely packed in the form of a translucent sheet-like structure. It was suggested that this close packing of CellNFs led to the destruction of fibrous structures in carbonised samples. On the other hand, freeze drying of CellNFs would retain the spatial gap between the fibres, and hence the carbonisation of freeze dried CellNFs would show characteristics different to that of carbonised oven-dried CellNFs. In this chapter, pyrolysis of freeze-dried CellNFs under atmospheric pressure and inert environment was investigated using controlled heating conditions. The freeze drying method leaves a network of nanofibres with spatial distance between the fibres and turns them into spongy looking masses. The effect of heat treatment conditions on the freeze-dried CellNFs and the carbonised residues was investigated using TGA, SEM, XRD and FT-IR. The results were compared with those from oven-dried CellNFs and the decomposition mechanism of CellNFs was discussed.

5.2. Experimental procedure

5.2.1. Freeze-drying

The procedure to produce CellNFs was described in chapter 3. A small amount of CellNF suspension was poured in a flask and dipped and stirred in liquid nitrogen. The CellNFs were then vacuum dried using a Labconco freeze-drier for 72 h during which water is removed. A spongy looking CellNF mass was collected which was used for carbonisation and characterisation.

5.2.2. Pyrolysis

Freeze-dried CellNFs were pyrolysed using a Lindberg tube furnace by gradually increasing the temperature to 400°C. The instrument calibration was carried out prior to experiments so that temperature variation was negligible.

An alumina boat was used to hold the samples in the tube. The macroscopic appearance of freeze-dried CellNFs was a spongy looking patch containing nanofibres. Small pieces of samples (1cm×1 cm) were cut and placed on the boat and was then put in the tube. Nitrogen gas was blown into the tube to create an inert atmosphere in the system. The flow rate was chosen to be 70 ml/min. Higher flow rates would blow the sample off the system whereas lower flow rates would not remove the by-product gases and water (which were produced from cellulose decomposition) from the system during pyrolysis.

5.2.2. Carbonisation of freeze-dried CellNFs

Table 5.1 shows the heat treatment conditions selected for carbonisation of freeze-dried CellNFs. The heating parameters were selected because of the same reasons as for ovendried samples described in the previous chapter.

Sample ID		Target temperature 1			Target temperature 2	
	Heating rate	Temp. (°C)	Residence time (h)	Heating rate	Temp. (°C)	Residence time (h)
	(C/min)			(C/min)		
NF16	10	170	0			
NF17	10	170	2			
NF18	10	170	3			
NF19	10	170	0	1	240	
NF20	10	170	0	2	240	
NF21	10	170	0	10	240	
NF22	10	240	1			
NF23	10	240	3			
NF24	10	240	17			
NF25	10	240	17	1	400	
NF26	10	240	17	10	400	
NF27	10	240	1	1	400	

Table 5.1. Heat treatment conditions.

Figure 5.1(a) shows the TG/DSC curves of freeze-dried CellNFs. Below 170°C, the mass loss was insignificant (<1%). Upon further heating to 240°C, the mass loss started to increase. The DSC curve in Figure 5.1(a) shows a peak at 360°C due to an endothermic reaction associated with the breakage of glycosidic rings of cellulose. CellNFs lost 81% of their weight between 240°C and 370°C. Further heating above 370°C caused negligible mass loss, indicating the near completion of the decomposition of cellulose and formation of a carbonaceous residue above 370°C. Figure 5.1(b) shows the mass loss of the sample during isothermal heat treatment at 240°C. After heat treatment for 1100 min, the mass loss nearly reached a plateau during which ~40% of its weight was lost. Based on this result, the effect of holding times of 0 h, 3 h (where weight loss was ~ 20%) and 17 h (where weight loss was ~ 40%) was investigated.



Figure 5.1. (a) TG/DSC graphs of CellNFs and (b) mass loss at 240°C as a function of holding time.

5.3. Results and Discussion

5.3.1. Effect of holding time at 170°C

5.3.1.1. Morphology

Figure 5.2 shows the SEM images of the samples at 170° C held for 0, 2 and 3 h. The fibre diameter distributions are also shown in Figure 5.2. It is noted that the fibrous morphology was retained after heat treatment. The average fibre diameter for the three heat treatment conditions is shown in Figure 5.2. The fibres show average diameters of < 50 nm.



Figure 5.2. SEM images and fibre diameter distributions of heat treated samples at 170°C, with holding time of (a) 0 h (NF16), (b) 2 h (NF17) and (c) 3h (NF18).

5.3.1.2. Crystallinity

Figure 5.3 Shows the XRD patterns of the samples heat treated with three holding times at 170°C. The XRD pattern of freeze dried CellNFs before heat treatment is also shown for comparison. Freeze-dried CellNFs show a sharp peak at 22.5°. This peak was assigned to the (200) lattice plane in cellulose-I crystal structure. This peak was observed in all the samples heat treated with various holding times indicating that the glycosidic rings have not been deformed by such pyrolysis conditions. As for the oven-dried

samples (Figure 4.10), the peak at ~ 15° shifted to a slightly higher degree in the samples NF17 and NF18 with long holding times (2 h and 3 h respectively), suggesting the formation of a disordered state of cellulose-I β structure.



Figure 5.3. XRD patterns of freeze-dried sample before heat treatment and heat treated samples at 170°C, with holding time of 0 h (NF16), 2 h (NF17) and 3h (NF18).

Table 5.2. Crystal size D, index of crystallinity I_{cr}, and interplanar distances d, of heat treated samples at 170°C with various holding times, estimated from XRD patterns.

Holding time	2 theta (deg)	hkl	d (Å)	D _{cr} (nm)	I _{cr}
3 h	16. 1 °	110, 1-10	4.3		
	22. 5 °	200	3.8	3.5	69
	34. 1 °	004	2.9	—	—
2 h	16. 1 °	110, 1-10	5.1	—	—
	22. 5 °	200	4.3	3.5	70
	34. 1 °	004	3.3	—	—
1 h	16. 1 °	110, 1-10	5.5	—	—
	22. 5 °	200	3.9	3.9	72
	34. 1 °	004	2.9	—	—

The crystal parameters of the heat treated CellNFs was calculated from the XRD patterns as shown in Table 5.2. It is evident that the crystal properties of heat treated CellNFs did not change dramatically with holding time.

5.3.1.3. Chemical bonding

Figure 5.4 shows the FT-IR spectra of freeze-dried CellNFs and heat treated samples at 170°C. The peaks were assigned to the chemical bonds listed in Table 4.4. For the sample with holding time of 0 and 2 hrs, the FT-IR spectra were nearly identical to that of unheat treated sample. However, in the spectrum of the sample held for 3 h (NF18), the relative intensities of the peaks at 1025 cm⁻¹ and 1050 cm⁻¹ (associated with C-OH bonds at the 1st and 2nd alcohols in the glucose rings), compared to the peak at 1112 cm⁻¹ (associated with C-O-C glycoside links in the cellulose chain), were reduced. This suggests that internal dehydration started to occur after holding the temperature at 170°C for 3 h.



Figure 5.4. FT-IR spectra of heat treated samples at 170°C with holding times of (a) 0 h (NF16), (b) 2 h (NF17) and (c) 3h (NF18); (a) 4000-800 cm⁻¹, (b) 2000-800 cm⁻¹.

5.3.1.4. Discussion on the effect of holding time at 170° C

The SEM images show that the temperature at 170°C was not enough to initiate decomposition of cellulosic rings (glycosidic rings) and hence the fibrous structure was maintained during heat treatment. According to the FT-IR spectra, internal dehydration may have started at 170°C though XRD study showed that the cellulose-I crystal structure was still retained.

5.3.2. Effect of heating rate between 170°C to 240°C

For the heat treatment up to 240°C, a residence time of 0 h was selected at 170°C.

5.3.2.1. Morphology

Figure 5.5 shows SEM images of the samples heat treated from 170°C to 240°C with heating rates of 1°C/min, 2°C/min and 10°C/min. The nanofibres retained their fibrous morphology with heating rates of 1°C/min and 2°C/min. The average fibre diameter measured for three samples showed similar value of 27 nm. It is noted that a heating rate of 10°C/min caused some damage to the fibrous structure in some areas, as indicated with arrows in Figure 5.5.



Figure 5.5. SEM images of freeze-dried samples heat treated samples to 240°C with heating rates of 1°C/min (NF19), 2°C/min (NF20) and 10°C/min (NF21).



Figure 5.6. Fibre diameter distribution (average of NF19, NF20 and NF21) estimated

from SEM images.

5.3.2.2. Crystallinity

Figure 5.7 shows the XRD patterns of the samples with different heating rates. The peak associated with the cellulose-I structure was observed at 22.5° for these samples. A broad peak at 16.1° also indicates that crystallinity has not been altered severely. Difference in heating rates between 170°C to 240°C did not greatly influence the crystal structure. Table 5.3 lists the crystal properties of the samples NF19, NF20 and NF21. The crystal size was reduced as the heating rate decreased.



Figure 5.7. XRD graphs of freeze-dried samples heat treated to 240°C with heating rates of 1°C/min (NF19), 2°C/min (NF20) and 10°C/min (NF21).

Table 5.3. Crystal size D, index of crystallinity I _{cr} , and interplanar distances d, of heat
treated samples at 240°C with various heating rates estimated from XRD patterns.

Heating rate (°C/min)	2 theta (deg)	hkl	d (Å)	D _{cr} (nm)	I _{cr}
1	16.1°	110.1-10	4 5		
1	10. 1	110,1-10	т. Ј		
	22. 5 °	200	3.4	2.2	69
	34. 1 °	004	3.1		—
2	16. 1 °	110,1-10	4.2		_
	22. 5 °	200	4.2	3.9	70
	34. 1 °	004	3.3	_	_
10	16°	110,1-10	4.6	—	—
	22. 5 °	200	3.9	3.6	72
	34. 1 °	004	2. 89	—	—

5.3.2.3. Chemical structure

Figure 5.8 shows the FT-IR spectra of samples heated to 240°C with three different heating rates. Peaks at 1336 cm⁻¹ and 1365 cm⁻¹ corresponding to O-H and C-H were still present in the spectra. Hence the difference in heating rates between 170°C to 240°C did not cause noticeable structural change within the molecule.

As described earlier, the bands at 1025 cm⁻¹ and 1050 cm⁻¹ were assigned to the vibration of C-OH bonds at the 1st and 2nd alcohols in the glucose rings, and 1112 cm⁻¹ was assigned to C-O-C glycoside links in the cellulose chain [323-326]. It is evident in Figure 5.8(b) that the relative intensities of the peaks at 1025 cm⁻¹ and 1050 cm⁻¹, compared to the peak at 1112 cm⁻¹, were reduced as the heating rate decreased. This indicates that, longer heat treatment between 170°C and 240°C induced more removal of the OH bonds via dehydration processes, while glycoside link remained intact.



Figure 5.8. FT-IR spectra of freeze-dried samples heat treated to 240°C with heating rates of 1°C/min (NF19), 2°C/min (NF20) and 10°C/min (NF21); (a) 4000-800 cm⁻¹, (b) 2000-800 cm⁻¹.

5.3.2.4. Discussion on the effect of heating rate between 170°C and 240°C

The SEM images show that fast heating rate of 10°C/min between 170°C and 240°C may influence the morphology by causing fusion of fibres and formation of film-like structures. However, according to the data shown in Figure 5.7 and Figure 5.8, the difference in heating rate caused no significant changes in crystal structures. However, further internal dehydration was initiated when the sample was heated longer. These

results show that slow heating rate between 170°C and 240°C is necessary to maintain the fibrous morphology of heat treated samples.

5.3.3. Effect of holding time at 240°C

5.3.3.1. Morphology

Figure 5.9 shows the SEM images of the samples held at 240° C for 0h, 3h and 17 h. The images did not show any distinct difference in morphology between the samples heat treated for different holding times at 240° C. The average fibre diameter was measured for three samples and was ~27 nm.



Figure 5.9. SEM images of freeze-dried samples heat treated to 240°C with holding times of 0 h (NF22), 3 h (NF23) and 17 h (NF24).



Figure 5.10. Fibre diameter distribution (average of NF22, NF23 and NF24) estimated from SEM images.

5.3.3.2. Crystallinity

Figure 5.11 shows the XRD patterns of the samples when held at 240°C for 0h, 3h and 17h. Table 5.4 shows the crystal properties of the samples estimated from the diffraction peaks at 16.1°, 22.5° and 34.1°. The values of d and D are similar to these of the samples held at 170°C. The results indicate that cellulose has retained its crystal structure irrespective of holding times.



Figure 5.11. XRD graphs of freeze-dried samples heat treated to 240°C with holding times of 0 h (NF22), 3 h (NF23) and 17 h (NF24).

Holding time (h)	2 theta (deg)	hkl	d (Å)	D _{cr} (nm)	I _{cr}
0	16. 1 °	110, 1-10	4. 7		
	22. 5 °	200	4. 1	2.5	62
	34. 1 °	004			
3	16. 1 °	110, 1-10	4.6		
	22. 5 °	200	3.8	3.9	60
	34. 1 °	004			—
17	16.1 °	110, 1-10	4. 2		—
	22. 5 °	200	4. 6	3.9	60
	34. 1 °	004	_	—	—

Table 5.4. Crystal size D, index of crystallinity I_{cr}, and interplanar distances d, of heat treated samples with various holding times at 240°C estimated from XRD patterns.

5.3.3.3. Chemical structure

Figure 5.12 shows the FT-IR spectra of the freeze-dried samples when held at 240°C for 0, 3 and 17 h. The spectra of samples NF16-NF23 resembled the spectrum of CellNFs. The bands at 1025 cm⁻¹ and 1050 cm⁻¹ were assigned to the vibration of C-OH bonds at the 1st and 2nd alcohols in the glucose rings, and 1112 cm⁻¹ was assigned to C-O-C glycoside links in the cellulose chain (Figure 5.12b) [323-326]. It is evident that the relative intensities of the peaks at 1025 cm⁻¹ and 1050 cm⁻¹, compared to the peak at 1112 cm⁻¹, were further reduced in NF24 from that in CellNFs. This indicates again that longer heat treatment below 240°C induced the removal of the OH bonds by internal dehydration processes, while glycoside link remained intact.



Figure 5.12. FT-IR spectra of freeze-dried samples heat treated to 240°C with holding times of 0 h (NF22), 3 h (NF23) and 17 h (NF24); (a) 4000-800 cm⁻¹, (b) 2000-800 cm⁻¹.

5.3.3.4. Discussion on the effect of holding time at 240° C

The results show that holding time at 240°C did not influence the morphology and crystal structure of heat treated samples. It is interesting to note that, even after 17 h of holding time, cellulose-I crystal structure was retained even after the mass loss of ~ 40 %. Similar results were also reported by Davidson and Losty [159]. The removal of hydrogen and oxygen from cellulose molecules continued as the heat treatment prolonged. Based on the results, two possible explanations can be drawn: a) dehydration was mostly taking place which was temperature-sensitive and b) the C=O formation seems to be reversible due to release of hydroxyl groups. The date indicates that glycosidic rings or C-O-C bonds (1, 4- β linkage) have been partially deformed although the bed/chair configuration has still been present in the molecule resulting the XRD peaks at 16.1° and 22.5°.

5.3.4. Effect of heating rate to 400°C

In order to investigate the effect of heat treatment condition in 240-400°C, the samples were heated to 240°C where they were held for 1h and 17 h. They were then heated to 400°C by 1°C/min and 10°C/min. Such experimental procedure was designed in order to distinguish the effects of heating rate and holding time.

5.3.4.1. Morphology

Figure 5.13 shows SEM images of the samples treated under various heat treatment conditions (NF25-NF27). These samples have undergone carbonisation and their appearance changed from a white spongy mass to a black char (Figure 5.15). The SEM

images of NF26 and NF27 show that, during carbonisation, the fibres were damaged by fusing together and forming bundles of carbon char rather than fibres. Although some fibres retained their fibrous structures, they did not retain their original high aspect ratios. On the other hand, the SEM image of NF25 shows the morphology similar to its cellulosic precursor with high aspect ratios. On the other hand, the SEM image of NF25 shows the morphology similar to its cellulosic precursor with high aspect ratios. If gure 5.14 shows the fibre diameter distribution of the sample NF25. The average diameter was around 25 nm, close to that of commercial multiwall carbon nanotubes.



Figure 5.13. SEM images of samples heat treated to 400°C under different conditions; holding time at 240°C and heating rate to 400°C are 17 h and 1°C/min (NF25), 17 h and 10°C/min (NF26), and 1 h and 1°C/min (NF27).

Chapter 5



Figure 5.14. fibre diameter distribution of sample NF25 estimated from SEM images.



Figure 5.15. Appearance of (a) freeze-dried CellNFs and (b) carbon residue after heat treatment at 400°C.

5.3.4.2. Crystallinity

The crystal structure of the precursor cellulose is expected to influence the decomposition and subsequently the resultant carbon materials [146]. Figure 5.16 shows XRD patterns of NF25-NF27. The peaks associated with cellulose-I crystal structure was no longer evident in the XRD patterns. Instead, the patterns showed an amorphous-like feature.

Chapter 5



Figure 5.16. XRD graphs of samples heated to 400°C under different conditions; holding time at 240°C and heating rate to 400°C are 17 h and 1°C/min (NF25), 17 h and 10°C/min (NF26), and 1 h and 1°C/min (NF27).

5.3.4.3. Chemical structure

Figure 5.17 shows the FT-IR spectra of samples under different heat treatment conditions. These samples show rather different curves to that of the CellNF precursors. The carbon residue shows peaks at 1603 cm⁻¹ and 1703 cm⁻¹ which were assigned to C=O and C=C bands respectively. The latter group did not appear in the samples heat treated below 400°C. This indicates the breakdown of glycosidic rings resulting in the formation of carbon rich char. However, oxygen was still present in the carbon chemical structure, as the peak corresponding to C=O was present in the spectra. It is suggested that complete removal of oxygen and hydrogen from the molecule occurs at above 400°C [279].



Figure 5.17. FT-IR spectra of samples heat treated to 400°C under different conditions; holding time at 240°C and heating rate to 400°C are 17 h and 1°C/min (NF25), 17 h and 10°C/min (NF26), and 1 h and 1°C/min (NF27); (a) 4000-800 cm⁻¹, (b) 2000-800 cm⁻¹.

5.3.4.4. Discussion on the effect of heating rates and holding time

The XRD study confirmed that the decomposition of cellulose-I crystal structure takes place in the temperature range of 240-350°C. The FT-IR data indicated that the decomposition of cellulose-I crystal structure is accompanied by the breakdown of glycosidic rings. The SEM study showed that the decomposition reactions are sensitive to heating rate; higher heating rates severely damage the fibrous morphology. The heating rate dependency of the morphology of carbon residue indicates the possibility of two or more competing reactions occurring during the decomposition of the cellulose-I crystal structure; one is to destroy cellulose chains that is favourable at high heating rates, another is to carbonise cellulose without chain scission that is favourable at low heating rates. These two reactions will be discussed further in the later sections.

In addition, holding time at 240°C has influence on the morphological properties of the resultant residue. Longer holding time of 17 h resulted in better preservation of the fibrous morphology. TG data showed that extensive mass loss occurred at 240°C (Figure5.1) while retaining the original cellulose-I crystal structure. The mass loss reached ~40 % after 17 h of isothermal heating. Since intra and inter molecular dehydration occur at 240°C, it is expected that longer holding time at 240°C removed much of the oxygen and hydrogen from the cellulose molecules. This will help prevent the formation of levoglucosan and the scission of cellulosic chain (Figure 2.17).

5.4. Discussion on the effect of heat treatment conditions and drying methods

5.4.1. Sensitivity of heat treatment conditions to the formation of CNFs

The results showed that heat treatment conditions around the thermal decomposition temperature greatly influence the morphological properties of CNFs. Longer holding time at 240°C and slower heating rate from 240°C to 400°C were necessary to preserve the original fibrous structures of CellNFs in CNFs. This may be explained from the viewpoint of intermolecular dehydration.

According to the results of XRD and SEM studies, the crystal structure, crystallinity and morphology were not significantly affected by the heat treatment conditions below 240° C. However, TG data showed that extensive mass loss occurred at 240° C (Figure 5.1). The mass loss reached ~40 % after 17 h of isothermal heating. The FT-IR

results indicate that the mass loss is associated with the loss of OH groups and hence is caused by intra- or inter-chain dehydration.

An intra-molecular dehydration model between two OH groups has frequently been used for explaining the mass loss of cellulose below 300°C [10]. This model predicts the formation of C=C and C=O bonds (Figure 5.18). However, as can be seen in the FT-IR spectrum of NF24, C=C or C=O bands in the aromatic rings did not appear after heat treatment for 17 h at 240°C. The result implies that the dehydration process occurred mainly between the interchain OH groups, but not between the OH groups in the same pyranose rings.



Figure 5.18. Intramolecular dehydration in cellulose chains.

In cellulose crystals, straight cellulose chains are interconnected with other chains through strong hydrogen bonds between the OH groups. Dehydration at 240° C may result in covalent bonding between cellulose single chains and help preserve fibrous structures upon carbonisation. In addition, the removal of OH from –CH₂OH groups may prevent (i) the chain scission at glycoside links and (ii) the formation of volatile and viscous tar (levoglucosan) (Figure 5.19), which may play a critical role in preserving the original fibrous structures (Figure 5.20).



Figure 5.19. Formation of levoglocosan associated with depolymerisation.



Figure 5.20. Model of cellulose pyrolysis [327].

The D-glucose unit in cellulose has 3 OH groups and their loss is equivalent to the mass loss of ~33 %. This value is smaller than the observed mass loss of 40 % during the isothermal heating at 240°C. Hence, apart from dehydration, the evolution of some gaseous species such as CO, CO₂ and CH₄ may have occurred at 240°C, without affecting the crystal structure of cellulose. Further study is required to identify these gases evolved during the isothermal heat treatment at 240°C by, for example, FT-IR or mass spectroscopy [328].

The SEM images of NF25 and NF26 show that, even after dehydration at 240°C for 17 h, fast heating rate can still destroy the original fibrous structure. This result suggests that the removal of remaining oxygen in the pyranose rings and glycoside links could lead to depolymerisation at fast heating rates (e.g. 10°C/min). The slow heating rate (1°C/min) may provide the carbon atoms with sufficient time to diffuse inside the nanofibres and

form aromatic rings without causing morphological damages. In addition, it has been reported that more oxygen and hydrogen can escape the cellulosic chains during carbonisation if slower heating rates are applied between 240°C to 350°C [1, 203]. This helps reduce oxygen-to-carbon ratio within the carbon structure, contributing to the preservation of fibrous structures in carbon residues.

5.4.2. Influence of drying methods on the morphology of carbon residue.

The difference in fibrous morphology of the carbon residues between the two drying methods can be explained from the viewpoint of the spatial distance between the fibres.

It is known that, cellulose goes through thermal decomposition between $180-400^{\circ}C$ and further aromatization above $400^{\circ}C$ starting with a dehydration ($180-240^{\circ}C$) followed by depolymerisation ($240-400^{\circ}C$) [283]. Tang et al. suggested that between $240-400^{\circ}C$ cellulosic rings start to break down which leads to the formation of highly active free radicals. They cause formation of gaseous products such as CO_2 and CO as well as H_2O and tar [279]. Due to the hydrophilic property of cellulose and upon oven-drying the CellNFs, the hydroxyl groups tend to bridge between the cellulosic chains of neighbouring nanofibres via hydrogen bonds. Thus the fibres tend to be in close contact with each other in the oven-dried sample. Viscous tar may connect fibres with each other when their distance is very short, leading to the disappearance of fibrous structure in oven-dried samples. On the other hand, freeze drying creates sufficient space between the sample becomes faster by the flow of an inert gas, which allows individual fibres to remain relatively isolated from adjacent fibres.

The difference in spatial distance between the fibres may also contribute to the difference in the formation of tar/levoglucosan. As shown in Figure 4.10, after isothermal heat treatment at 170°C, the FT-IR spectra of oven-dried sample did not show much change in the 1000–1100 cm⁻¹ region. On the other hand, the FT-IR spectra of freeze-dried sample did show changes in the 1000–1100 cm⁻¹ region as a function of holding time at 170°C; the relative intensities of the peaks at 1025 cm⁻¹ and 1050 cm⁻¹ (associated with C-OH bonds at the 1st and 2nd alcohols in the glucose rings), compared to the peak at 1112 cm⁻¹ (associated with C-O-C glycoside links in the cellulose chain), were reduced as the isothermal heat treatment time increased. The same trend was observed during heating from 170°C to 240°C. As shown in Figures 4.13 and 5.8, the FT-IR spectra of oven-dried

sample did not show much change in the 1000–1100 cm⁻¹ region, whilst the FT-IR spectra of freeze-dried sample did show changes in the 1000–1100 cm⁻¹ region, as a function of heating rate. In freeze-dried sample, lower heating rate (i.e., longer heat treatment time) caused reduced relative intensity of the peaks at 1025 cm⁻¹ and 1050 cm⁻¹, compared to the peak at 1112 cm⁻¹. The results indicate that internal dehydration process occurs below 240°C in freeze-dried samples but not in oven-dried samples. This may be because water molecules can escape cellulose easier from the open network structure in freeze-dried samples than from tightly fused structures in oven-dried samples.

Thus, large spatial distance between fibres in the precursor CellNFs can serve two proposes to preserve original fibrous structures in the carbon residue; (1) by promoting internal dehydration to prevent the formation of tar and chain scission, and (2) by preventing tar from bridging neighboring fibres so as not to form fusion between carbonised fibres.

5.5. Conclusion

In this chapter, the effects of heat treatment conditions on amorphous CNFs and their precursor CellNFs were investigated. Amorphous CNFs were produced by the pyrolysis of CellNFs below 400°C. It is shown that heat treatment conditions around the thermal decomposition temperature greatly influence the morphological properties of CNFs. Longer holding time at 240°C and slower heating rate from 240°C to 400°C were critical to preserve the original fibrous structures of the precursor CellNFs after pyrolysis. It was suggested that intermolecular dehydration was induced during the prolonged isothermal heat treatment at 240°C and that the removal of OH groups reduced the chance of chain scission and the formation of tar. This in turn contributed to the retention of the original fibrous structure during carbonisation between 240°C and 400°C. Slow heating rate helps the carbon chains to join neighbouring chains to form aromatic rings within the nanofibre boundaries without causing morphological damage to the nanofibre. These heat treatment conditions will have stronger influence on the morphology of resulting carbon residues in nanofibres than in conventional micron-diameter fibres, as the fibrous structures can be altered by atomic diffusion over much shorter distances in nanofibres than in microndiameter fibres. This study demonstrated that, upon optimization of heat treatment

conditions, CNFs with diameters of <100 nm can be produced from renewable raw materials in a simple and scalable manner.

Chapter 6 <u>Chapter 6</u>: Graphitization of <u>carbon nanofibres from freeze-</u> <u>dried cellulose nanofibre</u>

6.1. Introduction

In the previous chapter, CNFs were successfully fabricated from freeze-dried CellNFs. However, the fibres were made in an amorphous structure. In order to increase the mechanical properties, the fibres need to be graphitized. In this chapter, CNFs were heat treated at high temperatures. The effect of the graphitization temperatures on the properties of CNFs were investigated using TEM, SEM, XRD, FT-IR and Raman spectroscopy.

6.2. Experimental

CNFs were prepared by the pyrolysis of freeze-dried CellNFs under the optimized conditions described in Chapter 5. Amorphous CNFs were heated to 1600°C, 1800°C, 2000°C, 2200°C, 2500°C and 2800°C to study the effect of temperature on the properties of CNF. Each sample was held at these temperatures for 30 min before cooling down to room temperature. The heating rate of 10°C/min was used for all the heat treatment.

CVD grown CNFs with average fibre diameters of 200-500 nm and lengths of 10-40 μ m were purchased from Nanostructured and Amorphous Materials Inc., USA and used as received. Graphite powder was purchased from Asbury Graphite Mills, USA and used as received. The CVD grown CNFs and graphite powders were used as reference samples in Raman spectroscopy.

6.3. Results and discussion

6.3.1. Morphology

Figure 6.1 shows TEM images of the heat treated samples between 1600 and 2800°C. After heating up to 1600°C, aligned crystalline regions have been formed in some parts of

the fibres. The crystallized fibres were observed only in a localized area and the bulk for the samples did not show this feature. Nonetheless, it is interesting to note that the growth of crystalline structures started from such low temperatures. Ishida et al. suggested that the crystallization at low temperature will lead to the crystal structure different from graphite [149]. However, detailed structural analysis is required as a part of future work, by electron diffraction study. Upon increasing the heat treatment temperature from 1600°C, most of the fibrous structures were lost. Furthermore, in all samples, there were particles in different sizes. These particles developed hexagonal shapes as the temperature reached 2800°C.



Figure 6.1. TEM images of heat treated samples; a) 1600°C, b) 1800°C, c) 2000°C, d) 2200°C, e) 2500°C and f) 2800°C.

Figure 6.2 shows the SEM images of the heat treated samples between 1600° C and 2800° C. It appears that the fibres of ~20 nm in diameters are entangled and fused in a network structure or a film-like structure. Free nanofibres with high aspect ratios were rarely observed.



Figure 6.2. SEM images of heat treated samples; a) 1600°C, b) 1800°C, c) 2000°C, d) 2200°C, e) 2500°C and f) 2800°C.

Figure 6.3 shows the SEM images of the sample graphitized at 2000° C. Some fine fibrillar structures of ~5 nm in diameter were formed between two parts of the carbon residues. They were aligned in one direction and stretched parallel to each other. Further

investigation needs to be undertaken to understand how these fibrillar structures are formed.



Figure 6.3. SEM images of fine fibrillar structure in 2000°C sample; a) lower magnification and b) higher magnification.

6.3.2. XRD analysis

Figure 6.4 shows the XRD patterns of heat treated samples at various temperatures. All samples showed broad peaks at 15° similar to CNFs at 400°C (Figure 5.16). However, above 2000°C, the samples show crystalline peaks at 33°, 39°, 55° 65° and 69°. The peaks were identified to be associated with zirconium carbide. No peak associated with graphite was observed.



Figure 6.4. XRD patterns of heat treated samples.

6.3.3. FT-IR analysis

Figure 6.5 shows the FT-IR spectra of samples heat treated at various temperatures. The peaks at 1112cm⁻¹ and 1163cm⁻¹ that are associated with oxygen bonds were still present. A sharp peak was present at 1380cm⁻¹ associated with the strong Zr—O covalent bonds [329-331].



Figure 6.5. FT-IR spectra of heat treated samples; a) 4000-800 cm⁻¹, (b) 2000-800 cm⁻¹.

6.3.4. Raman analysis

Figure 6.6 shows Raman spectra of heat treated samples and Figure 6.7 shows Raman spectra of graphite and CVD CNFs. Compared to graphite, heat treated CNFs exhibit broader peaks at 1350 cm⁻¹ and 1580 cm⁻¹. These peaks are associated with D and G bands, respectively. D band shows the defective graphitic structure and G band is associated with the tangential vibration of carbon atoms in graphitic layers. The D/G ratio

gives an understanding about how much of the material is graphitic. A larger D/G value indicates more disordered structure [332-335].



Figure 6.6. Raman spectra of heat treated samples; a) 1600°C, B) 1800°C, C) 2000°C, D) 2200°C, E) 2500°C and F) 2800°C



Figure 6.7. Raman spectra of CVD grown CNF and graphite.

Table 6.1 shows the D and G band parameters associated with the graphitized samples. Figure 6.8 shows the D/G peak intensity ratio as a function of heat treatment temperature. As shown in Figure 6.8, D/G ratio decreased as the heat treatment temperature increased, indicating the progress of graphitization with increasing the heat treatment temperature. It is noted that the D/G ratio of the samples heat treated above 2200°C is between the D/G ratios of CVD CNFs and graphite. Hence the degree of graphitization in the samples heat treated in this temperature range is as high as that of commercial CVD-based CNFs. Figure 6.9 shows the Raman shift of the peak G as a function of heat treatment temperature temperature. The peak position shifted toward lower wavenumber as the heat treatment temperature increased. This result indicates that the number of graphite stacking layers increased as a function of heat treatment temperature [336].These results indicate that

graphitization started at around 1600°C. The investigation about the effect of holding time on the degree of graphitization is suggested for future study.

Sample	D/G	G (cm ⁻¹)	D (cm ⁻¹)
1600°C	1.447	1596	1340
1800°C	1. 192	1582	1347
2000°C	1.270	1581	1345
2200°C	0. 762	1577	1348
2500°C	0. 611	1572	1345
2800°C	0. 487	1579	1351
CVD CNF	0. 809	1587	1350
Graphite	0. 238	1579	1353

Table 6.1. D and G band parameters.



Figure 6.8. D/G peak intensity ratio as a function of heat treatment temperature.



Figure 6.9. Raman shift of the peak G as a function of heat treatment temperature.

6.3.4. Source of contaminant particles

As shown in Figures 6.1 and 6.4, a significant quantity of contaminant particles was observed in the heat treated samples. The contaminants were identified as ZrC. It is likely that the contaminants stemmed from the milling balls made of ceria doped zirconia. The reaction between zirconia (ZrO_2) and carbon to form ZrC via the following reaction is thermodynamically favourable above 1700°C;

$$ZrO_2 + 3C = ZrC + 2CO(g)$$
(6.1)

Thus ZrC particles were formed during heat treatment in the Tanmann furnace. It was assumed that the fragments of milling balls were not detected by XRD before graphitization, possibly because the quantity and crystallite sizes of the fragments were small compared to CellNFs.

6.4. Conclusion

In this chapter, CNFs obtained from freeze-dried CellNFs were heat treated at various temperatures above 1600°C. The effects of heat treatment temperatures on the graphitization and morphology of CNFs were characterised by TEM, SEM, XRD, FT-IR and Raman spectroscopy.
TEM and SEM studies showed that the original fibrous morphology was not retained during graphitization. XRD results showed that the samples were mainly amorphous, apart from ZrC crystal contaminants. FT-IR spectra of samples showed that oxygen related bonds were still present in samples after heat treatment. However, a small part of the sample heat treated at 1600°C showed crystal-like structures under TEM. Raman spectroscopy also indicated that graphitization started gradually from above 1600°C. The number of graphite stacking layers increased steadily as a function of heat treatment temperature. It was also found that in the sample heat treated at 2000°C, some fine fibres of around 5 nm in diameter were bridged between bundles of fibres. They were aligned in one direction and were parallel to each other.

A significant quantity of contaminant particles was observed in the heat treated samples. The contaminants were not detected before high temperature heat treatment. As the contaminant particles may act as catalysts, it is critical to minimize the amount of contaminant arising from milling balls in order to control the graphitization reaction in CNFs. Further study is required to optimize the heat treatment conditions to retain fibrous structures during high temperature heat treatment.

7.1. Main conclusions

In this thesis, the potential of pyrolysis methods to form CNFs from CellNFs was investigated. Although carbonisation of micron-sized cellulose fibres has been widely investigated in the past, in this thesis, the novel CellNFs obtained from ball-milled softwood pulp was used to investigate the carbonisation process. Various pyrolysis conditions including the effect of heating stages, heating rate in each stage, maximum temperature in each stage and holding time at the maximum temperature were investigated for their effects on the morphological and structural properties of resulting carbon residues. By the use of freeze-dried CellNFs and optimized heat treatment conditions, CNFs with diameters of <100 nm were produced from renewable raw materials in a simple and scalable manner. Several conclusions were drawn from the investigation as listed below:

1) Heat treatment conditions below 400°C affect the morphology of the resulting carbon nanofibres.

CNFs were produced by the pyrolysis of CellNFs below 400°C. It is shown that heat treatment conditions around the thermal decomposition temperature greatly influence the morphological properties of CNFs. Longer holding time at 240°C and slower heating rate from 240°C to 400°C were critical to preserve the original fibrous structures of precursor CellNFs in CNFs. It was suggested that intermolecular dehydration was induced during the prolonged heat treatment below 240°C and that the removal of OH groups reduced the chance of chain scission and the formation of tar/ levoglucosan. This in turn contributed to the retention of the original fibrous structure during carbonisation between 240°C and 400°C. Slow heating rate helps the carbon chains to join neighbouring chains to form aromatic rings within the nanofibre boundaries without causing morphological damage to the nanofibre.

2) The method to dry CellNF precursors affects the morphology of resulting carbon nanofibres.

The freeze drying method left a network of nanofibres with wider spatial distance between the fibres than oven-dried samples, and turned them into spongy looking masses. Hydrogen bonding still occurred between nanofibres and connected the fibres together in an open-network structure.

The open structure allowed effective removal of oxygen and hydrogen from the cellulose molecules during heat treatment below 240°C. This prevents the formation of levoglucosan and the scission of cellulosic chain during decomposition of cellulose between 240°C and 400°C and, in turn, the retention of the original fibrous structure.

On the other hand, oven-drying left a compact network of fibres which connected to each other via hydrogen bonding. During heat treatment below 240°C, the oxygen and hydrogen were not efficiently removed from the cellulose molecules. This may be due to the closely packed structure without much open space between fibres. As a result, the formation of levoglucosan and the scission of cellulosic chain were promoted during decomposition of cellulose between 240°C and 400°C, resulting in the loss of original fibrous structure. The viscous tar that was formed from levoglucosan may bridge the closely packed fibres, which further contributed to the loss of original fibrous structure.

It was also found that the method to dry CellNF precursors affects the yield of resulting CNFs. Oven-dried and freeze-dried CellNFs show mass loss of 80-85% and 90-95% respectively after carbonisation to 400°C. Greater mass loss in freeze-dried CellNFs may be the result of adjacent nanofibres having greater spatial distance which facilitates higher rates of oxygen and hydrogen removal from the cellulose molecules.

3) The possibility of graphitizing carbon nanofibres at temperatures above 1600°C was demonstrated.

Since graphitization starts from the surface and develops to the core of the fibre, having finer diameter would help the graphitization process to initiate at lower temperatures. It was shown that crystal-like structures were developed at lower temperatures (1600°C). However, the development of such crystals was localized in a small area of the sample and the bulk of the samples did not show such structures. Nonetheless, the results of Raman spectroscopy showed that formation of crystalline regions in the carbon residues

started at about 1600°C. Furthermore, sever contamination was reported occurring. It is speculated that the contaminant derived from zirconia milling balls and that zirconium carbide contaminants were formed during heat treatment at high temperatures. FT-IR analysis showed that oxygen related bonds were still present in samples after heat treatment at high temperatures

In summary, this thesis has provided a new understanding of the carbonisation of CellNFs. It was demonstrated that pyrolysis of CellNFs produces CNFs of < 100 nm in diameter which have potential in various applications such as biomedical, electronic and structural devices.

7.2. Suggested future work

While interesting and encouraging results were drawn, a lot of new horizons have been opened which need to be studied in the future. Based on the conclusions made in this work, future works are proposed to further deepen the understanding towards the carbonisation and graphitization processes of CellNFs:

1- Although the effect of heat treatment on the oven-dried CellNFs was investigated, the heat treatment conditions used were not exactly the same as those for freeze-dried CellNFs. This occurred because the experiments on freeze-dried samples were conducted after oven-dried samples when additional knowledge was accumulated. Thus, using the heat treatment conditions optimized for freeze-dried samples, the experiments on oven-dried samples should be conducted and the results should be compared between the two drying methods.

2- The formation of gaseous species during carbonisation between 240-400°C should be investigated using mass spectroscopy or FT-IR. It is thought that identification of gaseous species would provide better insight on the breakdown of cellulosic rings and complex reactions that occur between 240-400°C.

3- Graphitization should be thoroughly investigated using pure CellNFs free from contamination. Starting temperature should be around 1600°C with longer holding time. It is suggested that longer holding times may provide enough time and energy for the graphitized regions to develop across the fibres.

4- Investigations should be re-visited by addressing the yield of CNFs. Most of the investigations in this thesis were focused on the effects of heat treatment and drying conditions on the morphological feature of carbon residues. The optimization of pyrolysis conditions from the minimisation of mass loss should be investigated. Levoglucosan is volatile and breaks down into tar and some other by-products such as char, CO₂ and furan at higher temperatures. Once pyrolysis conditions were optimized, hydrogen and oxygen atoms could be eliminated from the cellulosic chains. Thus, less carbon atoms reacted with hydrogen and oxygen and released from the chains, and the formation of levoglucosan will be suppressed. This in turn had two effects:

- a. A smaller mass loss occurred;
- b. Fibrous structure was maintained.

Both effects are desirable and improve the cost-related issues and CNF properties.

5- The effect of atmospheric conditions during pyrolysis should be investigated. In this thesis, only nitrogen gas was used. Reducing atmosphere using hydrogen or ammonium gas may influence the decomposition processes of cellulose and in turn change the morphological and structural properties of carbon residues.

6- A thorough investigation should be carried out on how chemical pre-treatments on the CellNF precursors will affect the yield and morphology of carbonised fibres. Use of fire retardants for mass loss reduction should be investigated. The presence of chemicals absorbed onto CellNF precursors may also influence the decomposition reaction of CellNFs and in turn affects the chemical and physical structures of carbon residues.

7- Another method to increase the spatial distance between CellNFs upon drying should be explored. It is suggested that chemical treatment of the CellNFs within the suspension may avoid the formation of hydrogen bonding between CellNFs during drying. Hence, upon drying, the fibres will remain independent from each other. This will help retain the fibrous morphology during carbonisation.

8- The method to quantitatively analyse the fibre morphology should be developed. In this thesis, the retention of fibrous morphology was assessed qualitatively based on the SEM image. Quantitative analysis is required to further investigate the effect of drying and heat treatment conditions on the morphological properties of carbon residue

in detail. For example the use of specific surface area measurements should be considered.

9- In depth statistical analysis should be undertaken to source and eliminate fiber diameter measurements using Image Pro Plus software. Sources of error include:

- a) Operator eye
- b) Number of images
- c) Scale bar
- d) Fiber uniformity
- e) Angle of bar to fiber axis

Since the diameter is in nanometer range, these factors could potentially affect the fiber diameter.

REFERENCES

- Fitzer E, Carbon Fibers Filaments and Composites, Figueiredo J.L, Bernardo CA, Baker RTK, Huttinger KJ, editors. Kluwer Academic; Dordrecht; 1990 p 3-41.
- [2] Chung DDL, Carbon fiber composites. Newton, MA: Butterworth-Heinemann. 1994: 3-11.
- [3] Baggott JE, In: Perfect Symmetry The accidental discovery of buchminsterfullerene, Vol 89, Oxford University Press: Oxford; 1994.
- [4] Bokros JC, Dekker M, Deposition,Structure and Properties of Pyrolytic Carbon. In: Chemistry and Physics of Carbon—A Series of Advances,Chapter 5; 1969.
- [5] Morgan P. Carbon fibers and their composites, structure of the carbon atom.Boca Raton:CRC Press. 2005: 1-13
- [6] Askeland DR. The Science and Engineering of Materials. PWS-Kent, 1989: p. 591.
- [7] Smith WS, composites, In: Engineered Materials Handbook. ASM International, Ohio; 1987.
- [8] Fridman LI, Grebennikov SF. Theoretical aspects in the preparation and use of fibrous carbon adsorbents. Khimicheskie Volokna 1990; 6: 10-3.
- [9] Chung DDL, Carbon fiber composites. Newton, MA: Butterworth-Heinemann. 1994: 13-53.
- [10] Morgan P. Carbon fibers and their composites, Carbon Fiber Production using a Cellulosic based Precursor. Boca Raton:CRC Press. 2005: p. 269-294.
- [11] Morgan P. Carbon fibers and their composites, Carbon Fiber Production using a PAN based Precursor. Boca Raton:CRC Press. 2005: p. 185-267.
- [12] Gupta AK, Paliwal DK, Bajaj P, J Macromol Sci, 1991; 1(C31): 1-89.
- [13] Jazaeri E, Zhang L, Wang X, Tsuzuki T. Fabrication of carbon nanofiber by pyrolysis of freeze-dried cellulose nanofiber. Cellulose 2011; 18(6): 1481-5.
- [14] Mochida I, Korai Y, Ku CH, Watanabe F, Sakai Y. Chemistry of synthesis, structure, preparation and application of aromatic-derived mesophase pitch. Carbon 2000; 38(2): 305-28.
- [15] Peters ST. Handbook of Composites. Springer-Verlag. 2nd Ed. 1998: 169-86.

- [16]. Peebles L. Carbon Fibers: Formation, Structure, and Properties. CRC Press Inc, Boca Raton; 1995: 29.
- [17] Cho T, Lee YS, Rao R, Rao AM, Edie DD, Ogale AA. Structure of carbon fiber obtained from nanotube-reinforced mesophase pitch. Carbon, 2003; 41(7): 1419-24.
- [18] Singer LS. The mesophase and high modulus carbon fibers from pitch. Carbon, 1978; 16(6): 409-15.
- [19] Park YD, Ōya A, Ōtani S. Factors governing the solubility of mesophase into pitch distillate. Fuel, 1983; 62(12): 1499-502.
- [20] Korai Y, Wang YG, Yoon SH, Ishida S, Mochida I, Nakagawa Y, et al. Preparation of meso-carbon microbeads with uniform diameter from ARisotropic pitch in the presence of carbon black. Carbon, 1996; 34(9): 1156-59.
- [21] Watanabe F, Ishida S, Korai Y, Mochida I, Kato I, Sakai Y, et al. Pitch-based carbon fiber of high compressive strength prepared from synthetic isotropic pitch containing mesophase spheres. Carbon, 1999; 37(6): 961-67.
- [22] Bermejo J, Granda M, Menéndez R, García R, Tascón JMD, Thermal behaviour of extrographic fractions of coal tar and petroleum pitches. Fuel, 1997; 76(2): 179-87.
- [23] Braga CP, Dutra CHMC, Depine de Castro L, Andrade CT. Influence of heat and pressure treatment on the rheological behavior of petroleum pitches. Fuel, 2009; 88(5): 853-60.
- [24] Blanco C, Prada V, Santamaría R, Bermejo J, Menéndez R. Pyrolysis behaviour of mesophase and isotropic phases isolated from the same pitch. Journal of Analytical and Applied Pyrolysis. 2002; 63(2): 251-65.
- [25] Carbide U. High modulus, high strength carbon fibers produced from mesophase pitch. US Patent 4,005,183. 1977.
- [26] Otani S, Oya A. Recent Advances in Japan and the United States. In: Kawata K, Umekawa S, Kobayashi A, editors. Composites; 1986 1-10.
- [27] Rhee B, Chung DH, In SJ, Edie DD, A comparison of pressure and reflux in the two-stage production of mesophase. Carbon, 1991; 29(3): 343-50.
- [28] Edie DD, Fox NK, Barnett BC, Fain CC. Melt-spun non-circular carbon fibers. Carbon, 1986; 24(4): 477-82.

- [29] Hayashi JI, Nakashima M, Kusakabe K, Morooka S, Mitsuda S. Rapid stabilization of pitch fiber precursor by multi-step thermal oxidation. Carbon, 1995; 33(11): 1567-71.
- [30] Miyajima N, Dohi S, Akatsu T, Yamamoto T, Yasuda E, Tanabe Y, Stabilization and carbonization behavior of pitch-model compounds treated with iodine or bromine. Carbon, 2002; 40(9): 1533-39.
- [31] Tzeng SS, Pan JH. Oxidative stabilization of petroleum pitch at high pressure and its effects on the microstructure and carbon yield after carbonization/graphitization. Materials Chemistry and Physics, 2002; 74(2): 214-21.
- [32] Vilaplana-Ortego E, Alcañiz-Monge J, Cazorla-Amorós D, Linares-Solano A.
 Stabilisation of low softening point petroleum pitch fibres by HNO3. Carbon, 2003; 41(5): 1001-7.
- [33] Matsumoto T, Mochida I. Oxygen distribution in oxidatively stabilized mesophase pitch fiber. Carbon, 1993; 31(1): 143-47.
- [34] Singer L. Carbon fibres from mesophase pitch. Fuel, 1981; 60(9): 839–47.
- [35] Edie DD. The effect of processing on the structure and properties of carbon fibers. Carbon, 1998; 36(4): 345-62.
- [36] Baker RTK, Carbon Fibers Filaments and Composites, Figueiredo J.L, Bernardo CA, Baker RTK, Huttinger KJ, editors. Kluwer Academic; Dordrecht; 1990 p. 405-39.
- [37] Olive GH, Olive S. Molecular interactions and macroscopic properties of polyacrylonitrile and model substances. advanced polymer science, 1980; 32(123).
- [38] Jin S, Zheng Y, Gao G, Jin Z. Effect of polyacrylonitrile (PAN) short fiber on the mechanical properties of PAN/EPDM thermal insulating composites. Materials Science and Engineering: A, 2008; 483–484(0): 322-24.
- [39] Mathur RB, Bahl OP, Mittal J. A new approach to thermal stabilisation of PAN fibres. Carbon, 1992; 30(4): 657-63.
- [40] Limited, JEC. Process for producing Carbon Fibers, US Patent 4,069,297, 1976.
- [41] Daumit GP, Ko YS, Slater CR, Venner JG, Young CC. Melt-spun acrylic fibers possessing a highly uniform internal structure which are particularly suited for thermal conversion to high strength carbon fibers, U.S. Patent 5,168,004, 1992.

- [42] Mascia L, Paxton EG, Physical changes during thermal oxidation of PAN fibres and effects of γ-radiation pre-treatment. Thermochimica Acta, 1991; 184(2): 251-67.
- [43] Rašković V, Marinković S. Temperature dependence of processes during oxidation of PAN fibres. Carbon, 1975 13(6): 535-38.
- [44] Hou X, Yang X, Zhang L, Waclawik E, Wu S. Stretching-induced crystallinity and orientation to improve the mechanical properties of electrospun PAN nanocomposites. Materials & amp; Design, 2010; 31(4): 1726-30.
- [45] Liao CC, Wang CC, Chen CY, Lai WJ. Stretching-induced orientation of polyacrylonitrile nanofibers by an electrically rotating viscoelastic jet for improving the mechanical properties. Polymer, 2011; 52(10): 2263-75.
- [46] Mathur RB, Bahl OP, Matta VK, Nagpal KC. Modification of PAN precursor— Its influence on the reaction kinetics. Carbon, 1988; 26(3): 295-301.
- [47] Rašković V, Marinković S. Processes in sulfur dioxide treatment of PAN fibers. Carbon, 1978; 16(5): 351-57.
- [48] Marra SP, Ramesh KT, Douglas AS. Mechanical characterization of active poly(vinyl alcohol)–poly(acrylic acid) gel. Materials Science and Engineering: C, 2001; 14(1–2): 25-34.
- [49] Sahoo A, Ramasubramani KRT, Jassal M, Agrawal AK. Effect of copolymer architecture on the response of pH sensitive fibers based on acrylonitrile and acrylic acid. European Polymer Journal, 2007; 43(3): 1065-76.
- [50] Thorne DJ, Gough VJ, Hipkiss G. Experimental acrylic fibres with control of surface and internal irregularities. Fibre Science and Technology, 1970; 3(2): 119-27.
- [51] Tibbetts GG. Carbon Fibers Filaments and Composites, Figueiredo J.L, Bernardo CA, Baker RTK, Huttinger KJ, editors. Kluwer Academic; Dordrecht; 1990 p. 525-40.
- [52] Crook V, Ebdon J, Hunt B, Joseph P, Wyman P. The influence of comonomers on the degradation and flammability of polyacrylonitrile: Design input for a new generation of flame retardants. Polymer Degradation and Stability, 2010; 95(12): 2260-68.
- [53] Dalton S, Heatley F, Budd PM. Thermal stabilization of polyacrylonitrile fibres. Polymer, 1999; 40(20): 5531-43.

- [54] Rahaman MSA, Ismail AF, Mustafa A. A review of heat treatment on polyacrylonitrile fiber. Degradation and Stability, 2007; 92(8): 1421-32.
- [55] Xue TJ, McKinney MA, Wilkie CA. The thermal degradation of polyacrylonitrile. Degradation and Stability, 1997; 58(1–2): 193-202.
- [56] Ko TH, Lin JS. The effect of a chemical vapor deposited carbon film from acetylene on the properties of graphitized PAN-based carbon fibers. New Carbon Materials, 2006; 21(4): 297-301.
- [57] Kwizera P, Dresselhaus MS, Uhlmann DR, Perkins JS, Desper CR. The microstructure of intercalated graphite fibers. Carbon, 1982; 20(5): 387-94.
- [58] Nataraj SK, Yang KS, Aminabhavi TM. Polyacrylonitrile-based nanofibers—A state-of-the-art review. Progress in Polymer Science, 2012; 37(3): 487-513.
- [59] Wangxi Z, Jie L, Gang W. Evolution of structure and properties of PAN precursors during their conversion to carbon fibers. Carbon, 2003; 41(14): 2805-12.
- [60] Montes-Morán MA, Martí nez-Alonso A, Tascón JMD, Young RJ. Effects of plasma oxidation on the surface and interfacial properties of ultra-high modulus carbon fibres. Composites Part A: Applied Science and Manufacturing, 2001; 32(3–4): 361-71.
- [61] Rong H, Ryu Z, Zheng J, Zhang Y. Effect of air oxidation of Rayon-based activated carbon fibers on the adsorption behavior for formaldehyde. Carbon, 2002; 40(13): 2291-300.
- [62] Tong Y, Wang X, Su H, Xu L. Oxidation kinetics of polyacrylonitrile-based carbon fibers in air and the effect on their tensile properties. Corrosion Science, 2011; 53(8): 2484-88.
- [63] Mathur RB, Bahl OP, Mittal J, Advances in the development of highperformance carbon fibres from pan precursor. Composites Science and Technology, 1994; 51(2): 223-30.
- [64] Ogawa H, Saito K. Oxidation behavior of polyacrylonitrile fibers evaluated by new stabilization index. Carbon, 1995; 33(6): 783-8.
- [65] Ferguson J, Mahapatro B. Pyrolysis studies on polyacrylonitrile fibres: Physical property/chemical structure changes during the initial stages of pyrolysis. Fibre Science and Technology, 1976; 9(3): 161-75.

- [66] Grassie N, McGuchan R. Pyrolysis of polyacrylonitrile and related polymers—
 I. Thermal analysis of polyacrylonitrile. European Polymer Journal, 1970; 6(9):
 1277-91.
- [67] Grassie N, McGuchan R. Pyrolysis of polyacrylonitrile and related polymers—
 III. Thermal analysis of preheated polymers. European Polymer Journal, 1971;
 7(10): 1357-71.
- [68] Grassie N, McGuchan R, Pyrolysis of polyacrylonitrile and related polymers—
 VIII. Copolymers of acrylonitrile with vinyl acetate, vinyl formate, acrolein and methyl vinyl ketone. European Polymer Journal, 1973; 9(2): 113-24.
- [69] Mengoli G, Pagura C, Salmaso R, Tomat R, Zecchin S. Pyrolysis of polyacrylonitrile prepared by electrochemical initiation and bromine doping of pyrolysis derivatives. Synthetic Metals, 1986; 16(2): 173-88.
- [70] Nielsen M, Jurasek P, Hayashi J, Furimsky E. Formation of toxic gases during pyrolysis of polyacrylonitrile and nylons. Journal of Analytical and Applied Pyrolysis, 1995; 35(1): 43-51.
- [71] Varshavskii VY. Kinetics and mechanism of high-temperature pyrolysis of polyacrylonitrile. Polymer Science U.S.S.R., 1983; 25(4): 958-66.
- [72] Sacco AJ. Carbon Fibers Filaments and Composites, Figueiredo J.L, Bernardo CA, Baker RTK, Huttinger KJ, editors. Kluwer Academic; Dordrecht; 1990 p. 459-505.
- [73] Watt W. Strong Fibers, Watt W, Perov BV, editors. North-Holland; Amsterdam; 1985 p. 327-87.
- [74] Thorne DJ, Watt W, Perov BV, editors. North-Holland; Amsterdam; 1985 p. 475-93.
- [75] Donnet JB, Bansal RC. International Fiber Science and Technology, In: Carbon Fibers, Marcel Uekker, New York; 1990.
- [76] Kawaguchi M, Nozaki K, Motojima S, Iwanaga H. A growth mechanism of regularly coiled carbon fibers through acetylene pyrolysis. Journal of Crystal Growth, 1992; 118(3–4): 309-13.
- [77] Dunham MG, Edie DD. Model of stabilization for pan-based carbon fiber precursor bundles. Carbon, 1992; 30(3): 435-50.
- [78] Oberlin A. Carbonization and graphitization. Carbon, 1984; 22(6): 521-41.

- [79] Ayache J, Oberlin A, Inagaki M. Mechanism of carbonization under pressure, part I: Influence of aromaticity (polyethylene and anthracene). Carbon, 1990 28(2–3): 337-51.
- [80] Despres JF, Oberlin A. Microscopy studies of deposition of pyrolytic carbon on polished substracts. Carbon, 1996; 34(11): 1455-8.
- [81] Deurbergue A, Oberlin A. Stabilization and carbonization of pan-based carbon fibers as related to mechanical properties. Carbon, 1991 29(4–5): 621-8.
- [82] Goma J, Oberlin A. Microtexture and structure of high temperature massive pyrocarbons prepared on graphite substrates. Carbon, 1985; 23(1): 85-90.
- [83] Huttepain M, Oberlin A. Microtexture of nongraphitizing carbons and tem studies of some activated samples. Carbon, 1990; 28(1): 103-11.
- [84] Lafdi K, Bonnamy S, Oberlin A. Mechanism of anisotropy occurrence in a pitch precursor of carbon fibres: Part III—Hot stage microscopy of pitch B and C. Carbon, 1991; 29(7): 857-64.
- [85] Lafdi K, Bonnamy S, Oberlin A. Mechanism of anisotropy occurrence in a pitch precursor of carbon fibres: Part II—Pitch C. Carbon, 1991; 29(7): 849-55.
- [86] Lafdi K, Bonnamy S, Oberlin A. Mechanism of formation of the texture and microtexture in as-spun then oxidized anisotropic pitch-based carbon fibers, part I: Radial with wedge type i1. Carbon, 1992L; 30(4): 551-67.
- [87] Lafdi K, Bonnamy S, Oberlin A. Anisotropic pitch-based carbon fibers carbonization and graphitization (radial-with-wedge i1 and radial i2 types). Carbon, 1992; 30(4): 533-49.
- [88] Lafdi K, Bonnamy S, Oberlin A. Textures and structures in heterogeneous pitch-based carbon fibres (as-spun, oxidized, carbonized, and graphitized). Carbon, 1993; 31(1): 29-34.
- [89] Oberlin A, Terriere G. Graphitization studies of anthracites by high resolution electron microscopy. Carbon, 1975; 13(5): 367-76.
- [90] Rouzaud JN, Oberlin A. Structure, microtexture, and optical properties of anthracene and saccharose-based carbons. Carbon, 1989; 27(4): 517-29.
- [91] Takaku A, Shioya M, Manganzi T. The application of temperature-time superposition to the study of structural development in the heat-treatment of carbon fibers. Carbon, 1996; 34(11): 1449-52.
- [92] Gerald JDF, Pennock GM, Taylor GH. Domain structure in MP (mesophase pitch)-based fibres. Carbon, 1991; 29(2): 139-64.

- [93] Gerald JDF, Taylor GH, Brunckhorst LF, Pang LSK, Wilson MA. Filaments with conical microstructures formed during fullerene production. Carbon, 1993; 31(1): 240-4.
- [94] Gerald JDF, Taylor GH, Pennock GM. Microstructure of hydrogenated coal-tar pitches. Carbon, 1994; 32(7): 1389-95.
- [95] Pennock GM, Taylor GH, Gerald JDF. Microstructure in a series of mesophase pitch-based fibers from du pont: zones, folds, and disclinations. Carbon, 1993; 31(4): 591-609.
- [96] Taylor GH, Pennock GM, Gerald JDF, Brunckhorst LF. Influence of QI on mesophase structure. Carbon, 1993; 31(2): 341-54.
- [97] Bennett SC, Johnson DJ. Electron-microscope studies of structural heterogeneity in pan-based carbon fibres. Carbon, 1979; 17(1): 25-39.
- [98] Bennett SC, Johnson DJ, Murray R. Structural characterisation of a highmodulus carbon fibre by high-resolution electron microscopy and electron diffraction. Carbon, 1976; 14(2): 117-22.
- [99] Johnson DJ. Direct lattice resolution of layer planes in polyacrylonitrile based carbon fibres. Nature, 1970; 226()5247: 750-6
- [100] Johnson DJ. The measurement of stacking size and lattice distortion in carbon fibres. Carbon, 1982; 20(2): 150-5.
- [101] Johnson DJ, Crawford D, Oates C. The fine structure of a range of pan-based carbon fibres. Carbon, 1972; 10(3): 330.
- [102] Johnson JW, Thorne DJ. Effect of internal polymer flaws on strength of carbon fibres prepared from an acrylic precursor. Carbon, 1969; 7(6): 659-661.
- [103] Johnson JW, Thorne DJ. Effect of internal polymer flaws on strength of carbon fibres prepared from an acrylic precurser. Composites, 1970; 1(4): 258.
- [104] Dobb MG, Guo H, Johnson DJ. Image analysis of lattice imperfections in carbon fibres. Carbon, 1995; 33(8): 1115-20.
- [105] Dobb MG, Guo H, Johnson DJ, Par CR. Structure-compressional property relations in carbon fibres. Carbon, 1995; 33(11): 1553-59.
- [106] Dobb MG, Johnson DJ, Saville BP. Compressional behaviour of Kevlar fibres. Polymer, 1981; 22(7): 960-5.
- [107] Johnson DJ, Tomizuka I, Watanabe O. The fine structure of lignin-based carbon fibres. Carbon, 1975; 13(4): 321-5.

- [108] Johnson DJ, Tomizuka I, Watanabe O. The fine structure of pitch-based carbon fibres. Carbon, 1975; 13(6): 529-34.
- [109] Watt W. Strong Fibers, Watt W, Perov BV, editors. North-Holland; Amsterdam; 1985 p. 389-413.
- [110] Tse-Hao K, Tzy-Chin D, Jeng-An P, Ming-Fong L. The characterization of PAN-based carbon fibers developed by two-stage continuous carbonization. Carbon, 1993; 31(5): 765-71.
- [111] Chen W, Yu H, Liu Y, Chen P, Zhang M, Hai Y. Individualization of cellulose nanofibers from wood using high-intensity ultrasonication combined with chemical pretreatments. Carbohydrate Polymers, 2011; 83(4): 1804-11.
- [112] Pérez S, Samain D. Structure and Engineering of Celluloses, In Advances in Carbohydrate Chemistry and Biochemistry, Derek H, editor, Academic Press;2010: 25-116.
- [113] Peter Z. Conformation and packing of various crystalline cellulose fibers. Progress in Polymer Science, 2001; 26(9): 1341-417.
- [114] Williamson RE, Burn JE, Hocart CH. Towards the mechanism of cellulose synthesis. Trends in Plant Science, 2002; 7(10): 461-7.
- [115] Kroon-Batenburg LMJ, Kroon J, Northolt MG. Chain modulus and intramolecular hydrogen bonding in native and regenerated cellulose fibres. polymer communications, 1986; 27: 290–2.
- [116] Howsmon JA, Sisson WA. Cellulose and Cellulose Derivatives. Ott E, Spurlin HM, Graffin MW, editors, Interscience: New York:1954.
- [117] Hearle JWS. The structural mechanics of fibers. journal of polymer science, 1967; 20: 215–51.
- [118] Boerstoel H, Maatman H, Westerink JB, Koenders BM. Liquid crystalline solutions of cellulose in phosphoric acid. Polymer, 2001; 42(17): 7371-9.
- [119] Wooding NS. Rayon and acetate fibres, In Fibre Structure, Hearle JWS, Peters RH, editors, The Textile Institute: Manchester;1963.
- [120] Harland WG. The degree of polymerization and its distribution in cellulose rayons: Part V. journal of textile institute, 1955; 46: 483–99.
- [121] Iller E, Stupińska H, Starostka P. Properties of cellulose derivatives produced from radiation—Modified cellulose pulps. Radiation Physics and Chemistry, 2007; 76(7): 1189-94.

- [122] Toda K, Asakura T, Fukaya M, Entani E, Kawamura Y. Cellulose production by acetic acid-resistant Acetobacter xylinum. Journal of Fermentation and Bioengineering, 1997; 84(3): 228-31.
- [123] Jandura P, Riedl B, Kokta BV. Thermal degradation behavior of cellulose fibers partially esterified with some long chain organic acids. Polymer Degradation and Stability, 2000; 70(3): 387-94.
- [124] Kim UJ, Eom SH, Wada M. Thermal decomposition of native cellulose: Influence on crystallite size. Polymer Degradation and Stability, 2010; 95(5): 778-81.
- [125] Kim UJ, Kuga S. Thermal decomposition of dialdehyde cellulose and its nitrogen-containing derivatives. Thermochimica Acta, 2001; 369(1–2): 79-85.
- [126] Matsuoka S, Kawamoto H, Saka S. Thermal glycosylation and degradation reactions occurring at the reducing ends of cellulose during low-temperature pyrolysis. Carbohydrate Research, 2011; 346(2): 272-9.
- [127] Poletto M, Zattera AJ, Forte MMC, Santana RMC. Thermal decomposition of wood: Influence of wood components and cellulose crystallite size. Bioresource Technology, 2012; 109(0): 148-53.
- [128] Wikipedia.com. 2012.
- [129] Divne C, Ståhlberg J, Teeri TT, Jones TA. High-resolution crystal structures reveal how a cellulose chain is bound in the 50 Å long tunnel of cellobiohydrolase I from Trichoderma reesei. Journal of Molecular Biology, 1998; 275(2): 309-25.
- [130] French AD, Miller DP, Aabloo A. Miniature crystal models of cellulose polymorphs and other carbohydrates. International Journal of Biological Macromolecules, 1993; 15(1): 30-6.
- [131] Kobayashi K, Kimura S, Togawa E, Wada M. Crystal transition from cellulose II hydrate to cellulose II. Carbohydrate Polymers, 2011; 86(2): 975-81.
- [132] Maurer RJ, Sax AF. Molecular dynamics of cellulose crystal surfaces with ChemShell. Procedia Computer Science, 2010; 1(1): 1149-54.
- [133] Mitchell GR, Guo W, Davis FJ. Liquid crystal elastomers based upon cellulose derivatives. Polymer, 1992; 33(1): 68-74.
- [134] Oudiani AEl, Chaabouni Y, Msahli S, Sakli F, Crystal transition from cellulose
 I to cellulose II in NaOH treated Agave americana L. fibre. Carbohydrate
 Polymers, 2011; 86(3): 1221-9.

- [135] Pertsin AJ, Nugmanov OK, Marchenko GN. Crystal structure of cellulose polymorphs by potential energy calculations: 2. Regenerated and native celluloses. Polymer, 1986; 27(4): 597-601.
- [136] Reis D, Vian B, Chanzy H, Roland JC. Liquid crystal-type assembly of native cellulose-glucuronoxylans extracted from plant cell wall. Biology of the Cell, 1991; 73(2–3): 173-8.
- [137] Yui T, Nishimura S, Akiba S, Hayashi S. Swelling behavior of the cellulose Iβ crystal models by molecular dynamics. Carbohydrate Research, 2006; 341(15): 2521-30.
- [138] Morton WE, Hearle JWS. Physical Properties of Textile Fibres, The Textile Institute: Manchester,1993.
- [139] French AD. Physical and theoretical methods for determining the supramolecularstructure of cellulose, In Cellulose Chemistry and its Applications. Zeronian SH, editor, Ellis Horwood: Chichester;1985 p. 84–137.
- [140] Hearle JWS. Textile Fibers: A Comparative Overview, In: Encyclopedia of Materials: Science and Technology. Buschow KHJ, Cahn RW, Flemings MC, Ilschner B, Kramer EJ, Mahajan S, editors. Elsevier: Oxford; 2001 p. 9100-16.
- [141] Hearle JWS. Fracture of common textile fibres, In: Fiber Fracture, Elices M, Llorca J, editors, Elsevier Science Ltd: Oxford; 2002 p. 329-53.
- [142] O'Sullivan AC. Cellulose: the structure slowly unravels. Cellulose, 1997;4:173-207.
- [143] Kataoka Y, Kondo T. Quantitative analysis for the cellulose I alpha crystalline phase in developing wood cell walls. International Journal of Bio Macromolecule, 1992; 24: 37–41.
- [144] Zugenmaier P. Crystalline Cellulose and Devivatives: Characterization and Structures. Springer Series in Wood Science. Berlin: Springer-Verlag. 2008: p. 101-51.
- [145] Matthews JF, Skopec CE, Mason PE, Zuccato P, Torget RW, Sugiyama J, Himmel ME, Brady JW. Computer simulation studies of microcrystalline cellulose Iβ. Carbohydrate Research, 2006; 341(1): 138-52.
- [146] Morgan P. Carbon fibers and their composites, Carbon Fiber Production using a cellulosic based Precursor. Boca Raton:CRC Press. 2005: p.269-294.
- [147] Morgan P. Carbon fibers and their composites, Carbon Fiber Production using pitch based Precursor. Boca Raton:CRC Press. 2005: p.295-324.

- [148] Salmon S, Hudson SM. Crystal morphology, biosynthesis, and physical assembly of cellulose, chitin, and chitosan. Journal of Macromolecular Science-Reviews in Macromolecular Chemistry and Physics, 1997; C37(2): 199-276.
- [149] Ishida O, Kim DY, Kuga S, Nishiyama Y, Brown RM. Microfibrillar carbon from native cellulose.(Author abstract). Cellulose, 2004; 11(3): 475-6.
- [150] Agoda-Tandjawa G, Durand S, Berot S, Blassel C, Gaillard C, Garnier C, et al. Rheological characterization of microfibrillated cellulose suspensions after freezing. Carbohydrate Polymers, 2010; 80(3): 677-86.
- Bilbao R, Arauzo J, Millera A. Kinetics of thermal decomposition of cellulose:
 Part II. Temperature differences between gas and solid at high heating rates. Thermochimica Acta, 1987; 120(0): 133-41.
- [152] Broido A, Javierso AC, Ouano AC, Barrall EM. Molecular-weight decrease in early pyrolysis of crystalline and amorphous cellulose. Journal of Applied Polymer Science, 1973; 17(12): 3627-35.
- [153] Broido A, Nelson MA. Char yield on pyrolysis of cellulose. Combustion and Flame, 1975; 24(0): 263-8.
- [154] Broido A, Yow H. Resolution of molecular-weight distributions in slightly pyrolyzed cellulose using wiebull function. Journal of Applied Polymer Science, 1977; 21(6): 1677-85.
- [155] Brunner PH, Roberts PV. The significance of heating rate on char yield and char properties in the pyrolysis of cellulose. Carbon, 1980; 18(3): 217-24.
- [156] Cagnon B, Py X, Guillot A, Stoeckli F, Chambat G. Contributions of hemicellulose, cellulose and lignin to the mass and the porous properties of chars and steam activated carbons from various lignocellulosic precursors. Bioresource Technology, 2009; 100(1): 292-8.
- [157] Gaur S, Reed TB. Prediction of cellulose decomposition rates from thermogravimetric data. Biomass and Bioenergy, 1994; 7(1–6): 61-7.
- [158] Losty HHW, Blakelock HD. The structure and properties of partially pyrolysed cellulose carbon. 2nd conf. on carbon and graphite. London (UK), 1965: 29–35.
- [159] Davidson HW, Losty HHW. The initial pyrolyses of celluloses. GEC journal, 1963; 30: 22-8.
- [160] Kilzer FJ, Broido A. Speculations on nature of cellulose pyrolysis. Pyrodynamic, 1965; 2(2-3): 151-159.

- [161] Northolt MG, Veldhuizen LH, Jansen H. Tensile deformation of carbon fibers and the relationship with the modulus for shear between the basal planes. Carbon, 1991; 29(8): 1267-79.
- [162] Donnet JB, Bansal RC. Carbon Fibers, In: International Fiber Science and Technology. Marcel Dekker: New York; 1990 p. 267-366.
- [163] Johnson DJ, Carbon Fibers Filaments and Composites. Figueiredo JL, Bernardo CA, Baker RTK, Huttinger KJ, editors,Kluwer Academic: Dordrecht; 1990 p. 119-46.
- [164] Oshima H, Woollam JA, Yavrouian A, Dowell MB. Electrical and mechanical properties of copper chloride-intercalated pitch-based carbon fibers. Synthetic Metals, 1983; 5(2): 113-23.
- [165] Gaier JR, Slabe ME, Shaffer N. Stability of the electrical resistivity of bromine, iodine monochloride, copper(II) chloride, and nickel(II) chloride intercalated pitch-based graphite fibers. Carbon, 1988; 26(3): 381-7.
- [166] Gall NR, Rut'kov EV, Tontegode AY. Intercalation of nickel atoms under twodimensional graphene film on (111)Ir. Carbon, 2000; 38(5): 663-7.
- [167] Ho CT, Chung DDL. Carbon fibers brominated by electrochemical intercalation. Carbon, 1990; 28(4): 521-8.
- [168] Chiou JM, Ho CT, Chung DDL. Effect of bromination on the oxidation resistance of pitch-based carbon fibers. Carbon, 1989; 27(2): 227-31.
- [169] Wang X, Chung DDL. Electromechanical behavior of carbon fiber. Carbon, 1997; 35(5): 706-9.
- [170] Ho CT, Chung DDL. Bromination of graphitic pitch-based carbon fibers. Carbon, 1990; 28(6): 831-7.
- [171] Ho CT, Chung DDL. Kinetics of intercalate desorption from carbon fibers intercalated with bromine. Carbon, 1990; 28(6): 825-30.
- [172] Adams PM, Katzman HA, Rellick GS, Stupian GW. Characterization of high thermal conductivity carbon fibers and a self-reinforced graphite panel. Carbon, 1998; 36(3): 233-45.
- [173] Katzman HA, Adams PM, Le TD, Hemminger CS. Characterization of low thermal conductivity pan-based carbon fibers. Carbon, 1994; 32(3): 379-91.
- [174] Lavin JG, Boyington DR, Lahijani J, Nystem B, Issi JP. The correlation of thermal conductivity with electrical resistivity in mcsophase pitch-based carbon fiber. Carbon, 1993; 31(6): 1001-2.

- [175] Wang Jl, Gu M, Ma WG, Zhang X, Song Y. Temperature dependence of the thermal conductivity of individual pitch-derived carbon fibers. New Carbon Materials, 2008; 23(3): 259-63.
- [176] Wang Jl, Gu M, Ma WG, Zhang X, Song Y.Temperature dependence of the thermal conductivity of individual pitch-derived carbon fibers. Carbon, 2009; 47(1): 350-354.
- [177] Park SJ, Kim BJ. Roles of acidic functional groups of carbon fiber surfaces in enhancing interfacial adhesion behavior. Materials Science and Engineering: A, 2005; 408(1–2): 269-73.
- [178] Pittman Jr CU, Jiang W, Yue ZR, Gardner S, Wang L, Toghiani H, et al. Surface properties of electrochemically oxidized carbon fibers. Carbon, 1999; 37(11): 1797-807.
- [179] Bismarck A, Wuertz C, Springer J. Basic surface oxides on carbon fibers. Carbon, 1999; 37(7): 1019-27.
- [180] Boskovic B O, Golovko VB, Cantoro M, Kleinsorge B, Chuang ATH, Ducati C. Low temperature synthesis of carbon nanofibres on carbon fibre matrices. Carbon, 2005; 43(13): 2643-8.
- [181] Bengisu M, Yilmaz E. Oxidation and pyrolysis of chitosan as a route for carbon fiber derivation. Carbohydrate Polymers, 2002; 50(2): 165-75.
- [182] Dhami TL, Manocha LM, Bahl OP. Oxidation behaviour of pitch based carbon fibers. Carbon, 1991; 29(1): 51-60.
- [183] Fukunaga A, Ueda S, Nagumo M. Air-oxidation and anodization of pitch-based carbon fibers. Carbon, 1999; 37(7): 1081-5.
- [184] Gulyás J, Földes E, Lázár A, Pukánszky B. Electrochemical oxidation of carbon fibres: surface chemistry and adhesion. Composites Part A: Applied Science and Manufacturing, 2001; 32(3–4): 353-60.
- [185] Ismail IMK, Hurley WC. Modelling carbon fibers oxidation in air at constant heating rates. Carbon, 1992; 30(3): 419-27.
- [186] Merino C, Brandl W. Oxidation behaviour and microstructure of vapour grown carbon fibres. Solid State Sciences, 2003; 5(4): 663-8.
- [187] Morgan P. The Uses of Carbon Fibers, In: Carbon Fibers and Their Composites, CRC Press; 2009.
- [188] Toutanji HA, El-Korchi T, Katz RN. Strength and reliability of carbon-fiberreinforced cement composites. Cement Concrete Comp, 1994; 16(1): 15-21.

- [189] Scholz MS, Blanchfield JP, Bloom LD, Coburn BH, Elkington M, Fuller JD, et al. The use of composite materials in modern orthopaedic medicine and prosthetic devices: A review. Composites Science and Technology, 2011; 71(16): 1791-803.
- [190] Ruyter IE, Ekstrand K, Björk N. Development of carbon/graphite fiber reinforced poly(methyl methacrylate) suitable for implant-fixed dental bridges. Dental Materials, 1986; 2(1): 6-9.
- [191] Viguie G, Malquarti G, Vincent B, Bourgeois D. Epoxy/carbon composite resins in dentistry: Mechanical properties related to fiber reinforcements. The Journal of Prosthetic Dentistry, 1994; 72(3): 245-9.
- [192] Hastings GW. Carbon fibre composites for orthopaedic implants. Composites, 1978; 9(3): 193-7.
- [193] Prakash R, Marwah S, Goel SC, Tuli SM. Carbon fibre reinforced epoxy implants for bridging large osteoperiosteal gaps. Biomaterials, 1988; 9(2): 198-202.
- [194] Tancredi A, Agrillo A, Delfini R, Fiume D, Frati A, Rinaldi A. Use of carbon fiber cages for treatment of cervical myeloradiculopathies. Surgical Neurology, 2004; 61(3): 221-6.
- [195] Nolan L. Carbon fibre prostheses and running in amputees: A review. Foot and Ankle Surgery, 2008 14(3): 125-9.
- [196] Paton W. Carbon fibre in sports equipment. Composites, 1970; 1(4): 221-6.
- [197] liquid control corp. RTM forms carbon fibre bicycle frames. Reinforced Plastics, 1997; 41(4): 28-29.
- [198] Chen YS, James-Lin MJ, Chang CH, Liu FM. Technological innovations and industry clustering in the bicycle industry in Taiwan. Technology in Society, 2009; 31(3): 207-17.
- [199] Liu TJC, Wu HC. Fiber direction and stacking sequence design for bicycle frame made of carbon/epoxy composite laminate. Materials & amp; Design, 2010; 31(4): 1971-80.
- [200] Sevilla, M. and A.B. Fuertes, Graphitic carbon nanostructures from cellulose. Chemical Physics Letters, 2010. 490(1-3): p. 63-68.
- [201] Tran PA, Zhang L, Webster TJ. Carbon nanofibers and carbon nanotubes in regenerative medicine. Advanced Drug Delivery Reviews, 2009; 61(12): 1097-114.

- [202] Uchida T, Anderson DP, Minus ML, Kumar S. Morphology and modulus of vapor grown carbon nano fibers. Journal of Materials Science, 2006; 41(18): 5851-6.
- [203] Yoshino K, Matsuoka R, Nogami K, Yamanaka S, Watanabe K, Takahashi M, et al. Graphite film prepared by pyrolysis of bacterial cellulose. Journal of Applied Physics, 1990; 68(4): 1720-5.
- [204] Jang J, Bae J. Carbon nanofiber/polypyrrole nanocable as toxic gas sensor. Sensors and Actuators B: Chemical, 2007; 122(1): 7-13.
- [205] Kang SC, Im JS, Lee SH, Bae TS, Lee YS. High-sensitivity gas sensor using electrically conductive and porosity-developed carbon nanofiber. Colloids and Surfaces A: Physicochemical and Engineering Aspects, 2011; 384(1–3): 297-303.
- [206] Veerappan G, Kwon W, Rhee SW. Carbon-nanofiber counter electrodes for quasi-solid state dye-sensitized solar cells. Journal of Power Sources, 2011; 196(24): 10798-805.
- [207] Zhu H, Wei J, Wang K, Wu D. Applications of carbon materials in photovoltaic solar cells. Solar Energy Materials and Solar Cells, 2009; 93(9): 1461-70.
- [208] Martone A, Faiella G, Antonucci V, Giordano M, Zarrelli M. The effect of the aspect ratio of carbon nanotubes on their effective reinforcement modulus in an epoxy matrix. Composites Science and Technology, 2011; 71(8): 1117-23.
- [209] Naji H, Zebarjad SM, Sajjadi SA. The effects of volume percent and aspect ratio of carbon fiber on fracture toughness of reinforced aluminum matrix composites. Materials Science and Engineering: A, 2008; 486(1–2): 413-20.
- [210] Lin CT, Kao PW, Jen MHR. Thermal residual strains in carbon fibre-reinforced aluminium laminates. Composites, 1994; 25(4): 303-7.
- [211] Koráb J, Štefánik P, Kavecký Š, Šebo P, Korb G. Thermal conductivity of unidirectional copper matrix carbon fibre composites. Composites Part A: Applied Science and Manufacturing, 2002; 33(4): 577-81.
- [212] Wang M, Kang Q, Pan N. Thermal conductivity enhancement of carbon fiber composites. Applied Thermal Engineering, 2009; 29(2–3): 418-21.
- [213] Chung DDL. Thermal analysis of carbon fiber polymer-matrix composites by electrical resistance measurement. Thermochimica Acta, 2000; 364(1–2): 121-32.

- [214] Lodge KJ. The electrical properties of joints in carbon fibre composites. Composites, 1982; 13(3): 305-10.
- [215] Parikh N, Allen J, Yassar RS. Effect of deformation on electrical properties of carbon fibers used in gas diffusion layer of proton exchange membrane fuel cells. Journal of Power Sources, 2009 193(2): 766-8.
- [216] Zhou Z, Lai C, Zhang L, Qian Y, Hou H, Reneker DH, et al. Development of carbon nanofibers from aligned electrospun polyacrylonitrile nanofiber bundles and characterization of their microstructural, electrical, and mechanical properties. Polymer, 2009; 50(13): 2999-3006.
- [217] Etika KC, Liu L, Hess LA, Grunlan JC. The influence of synergistic stabilization of carbon black and clay on the electrical and mechanical properties of epoxy composites. Carbon, 2009; 47(13): 3128-36.
- [218] Morgan P. Carbon fibers and their composites, The Uses of Carbon Fibers. Boca Raton:CRC Press. 2005 951-1042.
- [219] Kurtz SM. Composite UHMWPE biomaterials and fibers, In: UHMWPE Biomaterials Handbook. 3rd ed. Steven MK, editors. Academic Press: Boston; 2009 p. 249-258.
- [220] Ramakrishna S, Mayer J, Wintermantel E, Leong KW. Biomedical applications of polymer-composite materials: a review. Composites Science and Technology, 2001; 61(9): 1189-224.
- [221] Reed KP, van den Berg SS, Rudolph A, Albright JA, Casey HW, Marina AA. Treatment of tendon injuries in thoroughbred racehorses using carbon-fiber implants. Journal of Equine Veterinary Science, 1994; 14(7): 371-7.
- [222] Ohsaki T, Kanda M, Aoki Y, Shiroki H, Suzuki S. High-capacity lithium-ion cells using graphitized mesophase-pitch-based carbon fiber anodes. Journal of Power Sources, 1997; 68(1): 102-5.
- [223] Millar J, Williams GV. Ultra-low noise silver-plated carbon fibre microelectrodes. Journal of Neuroscience Methods, 1988; 25(1): 59-62.
- [224] Qiu J, Zhang SH, Wang GJ, Gong Yl. Surface modification and application of multi-walled carbon nanotubes in fire-retardant coatings. Carbon, 2010; 48(5): 1694-5.
- [225] Wu Q, Zhu W, Zhang C, Liang Z, Wang B. Study of fire retardant behavior of carbon nanotube membranes and carbon nanofiber paper in carbon fiber reinforced epoxy composites. Carbon, 2010; 48(6): 1799-806.

- [226] Carter JT, Emmerson GT, Faro CL, McGrail PT, Moore DR. The development of a low temperature cure modified epoxy resin system for aerospace composites. Composites Part A: Applied Science and Manufacturing, 2003; 34(1): 83-91.
- [227] Hailian Y, Xiongqing Y. Integration of Manufacturing Cost into Structural Optimization of Composite Wings. Chinese Journal of Aeronautics, 2010; 23(6): 670-6.
- [228] Noor AK, Venneri SL, Paul DB, Hopkins MA. Structures technology for future aerospace systems. Computers & amp; Structures, 2000; 74(5): 507-19.
- [229] Mills A. Automation of carbon fibre preform manufacture for affordable aerospace applications. Composites Part A: Applied Science and Manufacturing, 2001; 32(7): 955-62.
- [230] Parks BR. Carbon fibres in helicopter blades. Composites, 1974 5(4):144.
- [231] Goodwin, J. R. Fibre reinforced blade. UK Patent 1,186,486, 1970.
- [232] van der Lee MK, van Dillen AJ, Geus JW, de Jong KP, Bitter JH. Catalytic growth of macroscopic carbon nanofiber bodies with high bulk density and high mechanical strength. Carbon, 2006; 44(4): 629-37.
- [233] Morgan P. Carbon fibers and their composites, Production of Vapor Grown Carbon Fibers (VGCF). Boca Raton:CRC Press. 2005; 325-345.
- [234] Kamada K, Ikuno T, Takahashi S, Oyama T, Yamamoto T, Kamizono M, et al. Surface morphology and field emission characteristics of carbon nanofiber films grown by chemical vapor deposition on alloy catalyst. Applied Surface Science, 2003; 212–213(0): 383-7.
- [235] Bi H, Kou KC, Ostrikov K, Yan LK, Zhang JQ, Zheng Ji T, et al. Unconventional Ni–P alloy-catalyzed CVD of carbon coil-like micro- and nanostructures. Materials Chemistry and Physics, 2009; 116(2–3): 442-8.
- [236] Bronikowski MJ. CVD growth of carbon nanotube bundle arrays. Carbon, 2006; 44(13): 2822-32.
- [237] Kim SS, Kim KW, Ahn HJ, Cho KK. Characterization of graphitic nanofibers synthesized by the CVD method using nickel–copper as a catalyst. Journal of Alloys and Compounds, 2008; 449(1–2): 274-8.
- [238] Lee S, Choi S, Park KH, Chae KW, Cho JB, Ahn Y. Hot-filament CVD synthesis and application of carbon nanostructures. Thin Solid Film, 2008; 516(5): 700-5.

- [239] Zheng GB, Kouda K, Sano H, Uchiyama Y, Shi YF, Quan HJ. A model for the structure and growth of carbon nanofibers synthesized by the CVD method using nickel as a catalyst. Carbon, 2004; 42(3): 635-40.
- [240] Figueiredo JL, Bernardo CA, Baker RTK, Huttinger KJ, Carbon Fibers Filaments and Composites. Kluwer Academic: Dordrecht; vol 74, 1990
- [241] Martínez-Latorre L, Armenise S, Garcia-Bordejé E. Temperature-mediated control of the growth of an entangled carbon nanofiber layer on stainless steel micro-structured reactors. Carbon, 2010; 48(7): 2047-56.
- [242] Al-Saleh MH, Sundararaj U. A review of vapor grown carbon nanofiber/polymer conductive composites. Carbon, 2009; 47(1): 2-22.
- [243] Blazsà M. Recent trends in analytical and applied pyrolysis of polymers. Journal of Analytical and Applied Pyrolysis, 1997; 39(1): 1-25.
- [244] Cheung KY, Lee KL, Lam KL, Chan TY, Lee CW, Hui CW. Operation strategy for multi-stage pyrolysis. Journal of Analytical and Applied Pyrolysis, 2011; 91(1): 165-82.
- [245] LÃ QF, He ZW, Zhang JY, Lin Q. Fabrication of nitrogen-containing hollow carbon nanospheres by pyrolysis of self-assembled poly(aniline-co-pyrrole). Journal of Analytical and Applied Pyrolysis, 2012 93(0): 147-52.
- [246] Vera-Agullo J, Varela-Rizo H, Font R, Conesa JA, Martin-Gullon I. Analytical pyrolysis as a characterization technique for monitoring the production of carbon nanofilaments. Journal of Analytical and Applied Pyrolysis, 2007 79(2): 484-9.
- [247] Yusof N, Ismail AF. Post spinning and pyrolysis processes of polyacrylonitrile (PAN)-based carbon fiber and activated carbon fiber: A review. Journal of Analytical and Applied Pyrolysis.2012; 93(0): 1-13.
- [248] PÃ E, Ate F, PÃ AE. Catalytic pyrolysis of biomass in inert and steam atmospheres. Fuel, 2008; 87(6): 815-24.
- [249] Wang P, Jin L, Liu J, Zhu S, Hu H. Analysis of coal tar derived from pyrolysis at different atmospheres. Fuel, 2010; In press.
- [250] Di BC. Comparison of semi-global mechanisms for primary pyrolysis of lignocellulosic fuels. Journal of Analytical and Applied Pyrolysis, 1998; 47(1): 43-64.

- [251] Lehrle RS. Polymer pyrolysis mechanisms: Experimental approaches for investigating them. Journal of Analytical and Applied Pyrolysis, 1987; 11(0): 55-64.
- [252] Liu Q, Wang S, Zheng Y, Luo Z, Cen K. Mechanism study of wood lignin pyrolysis by using TG–FTIR analysis. Journal of Analytical and Applied Pyrolysis, 2008; 82(1): 170-7.
- [253] Savage PE. Mechanisms and kinetics models for hydrocarbon pyrolysis. . Journal of Analytical and Applied Pyrolysis, 2000; 54(2): 109-26.
- [254] Formhals A. Apparatus for producing artificial filaments from material such as cellulose acetate. Schreiber-Gastell; 1934.
- [255] Chronakis IS. Novel nanocomposites and nanoceramics based on polymer nanofibers using electrospinning process—A review. journal of materials processing technology, 2005; 167(2–3): 283-93.
- [256] Li D, Xia Y. Electrospinning of nanofibers: reinventing the wheel? Advanced materials, 2004; 16: 1151.
- [257] Chronakis I. Novel nanocomposites and nanoceramics based on polymer nanofibers using electrospinning process- A review. journal of materials processing technology, 2005; (167): 283.
- [258] Barhate RS, Ramakrishna S. Nanofibrous filtering media: Filtration problems and solutions from tiny materials. Journal of Membrane Science, 2007; 296(1–2): 1-8.
- [259] Barhate RS, Koeppl S, Ramakrishna S. Porous nano- and microfibrous polymeric membrane material for catalytic support. Chemical Engineering Research and Design, 2011; 89(6): 621-30.
- [260] Leclerc E, Furukawa KS, Miyata F, Sakai Y, Ushida T, Fujii T. Fabrication of microstructures in photosensitive biodegradable polymers for tissue engineering applications. Biomaterials, 2004; 25(19): 4683-90.
- [261] Park JH, Allen MG, Prausnitz MR. Biodegradable polymer microneedles: Fabrication, mechanics and transdermal drug delivery. Journal of Controlled Release, 2005; 104(1): 51-66.
- [262] Agarwal S, Wendorff JH, Greiner A. Use of electrospinning technique for biomedical applications. Polymer, 2008; 49(26): 5603-21.

- [263] Ding B, Kim J, Miyazaki Y, Shiratori S. Electrospun nanofibrous membranes coated quartz crystal microbalance as gas sensor for NH3 detection. Sensors and Actuators B: Chemical, 2004; 101(3): 373-80.
- [264] Ding B, Wang M, Wang X, Yu J, Sun G. Electrospun nanomaterials for ultrasensitive sensors. Materials Today, 2010; 13(11): 16-27.
- [265] Lee KJ, Shiratori N, Lee GH, Miyawaki J, Mochida I, Yoon SH, et al. Activated carbon nanofiber produced from electrospun polyacrylonitrile nanofiber as a highly efficient formaldehyde adsorbent. Carbon, 2010; 48(15): 4248-55.
- [266] Zhang H, Nie H, Yu D, Wu C, Zhang Y, White CJB, et al., Surface modification of electrospun polyacrylonitrile nanofiber towards developing an affinity membrane for bromelain adsorption. Desalination, 2010; 256(1–3): 141-7.
- [267] Nataraj SK, Kim BH, Yun JH, Lee DH, Aminabhavi TM, Yang KS. Morphological characterization of electrospun carbon nanofiber mats of polyacrylonitrile containing heteropolyacids. Synthetic Metals, 2009; 159(14): 1496-504.
- [268] Liao Y, Zhang L, Gao Y, Zhu ZT, Fong H. Preparation, characterization, and encapsulation/release studies of a composite nanofiber mat electrospun from an emulsion containing poly(lactic-co-glycolic acid). Polymer, 2008; 49(24): 5294-9.
- [269] Wang L, Yu Y, Chen PC, Chen CH. Electrospun carbon–cobalt composite nanofiber as an anode material for lithium ion batteries. Scripta Materialia, 2008; 58(5): 405-8.
- [270] Li N, Hui Q, Xue H, Xiong J. Electrospun Polyacrylonitrile nanofiber yarn prepared by funnel-shape collector. Materials Letters, 2012; 79(0): 245-7.
- [271] Ma Z, Kotaki M, Ramakrishna S. Electrospun cellulose nanofiber as affinity membrane. Journal of Membrane Science, 2005; 265(1–2): 115-23.
- [272] Luo CJ, Nangrejo M, Edirisinghe M. A novel method of selecting solvents for polymer electrospinning. Polymer, 2010; 51(7): 1654-62.
- [273] Kim DY, Nishiyama Y, Wada M, Kuga S. Graphitization of highly crystalline cellulose. Carbon, 2001; 39(7): 1051-6.
- [274] Zhang L, Tsuzuki T, Wang X. Preparation and characterization on cellulose nanofiber film. Materials Science Forum, 2010; 654-656: 1760-3.

- [275] Collinson SR, Thielemans W. The catalytic oxidation of biomass to new materials focusing on starch, cellulose and lignin. Coordination Chemistry Reviews, 2010; 254(15–16): 1854-70.
- [276] Knill CJ, Kennedy JF. Degradation of cellulose under alkaline conditions. Carbohydrate Polymers, 2003; 51(3): 281-300.
- [277] Robert R, Barbati S, Ricq N, Ambrosio M. Intermediates in wet oxidation of cellulose: identification of hydroxyl radical and characterization of hydrogen peroxide. Water Research, 2002; 36(19): 4821-9.
- [278] Lie JA, Hägg MB. Carbon membranes from cellulose and metal loaded cellulose. Carbon, 2005; 43(12): 2600-7.
- [279] Tang MM, Bacon R. Carbonization of cellulose fibers--I. Low temperature pyrolysis. Carbon, 1964; 2(3): 211-4
- [280] Bacon R, Silvaggi AF. Electron microscope study of the microstructure of carbon and graphite fibers from a rayon precursor. Carbon, 1971; 9(3): 321-5.
- [281] Plaisantin H, Pailler R, Guette A, Daudé G, Pétraud M, Barbe B, et al. Conversion of cellulosic fibres into carbon fibres: a study of the mechanical properties and correlation with chemical structure. Composites Science and Technology, 2001; 61(14): 2063-8.
- [282] Pastor AC, Rodríguez-Reinoso F, Marsh H, Martínez MA. Preparation of activated carbon cloths from viscous rayon. Part I. Carbonization procedures. Carbon, 1999; 37(8): 1275-83.
- [283] Sekiguchi Y, Frye JS, Shafizadeh F. Structure and formation of cellulosic chars. Journal of Applied Polymer Science, 1983; 28(11): 3513-25.
- [284] Sevilla M, Fuertes AB. The production of carbon materials by hydrothermal carbonization of cellulose. Carbon, 2009 47(9): 2281-9.
- [285] Pastorova I, Botto RE, Arisz PW, Boon JJ. Cellulose char structure: a combined analytical Py-GC-MS, FTIR, and NMR study. Carbohydrate Research, 1994; 262(1): 27-47.
- [286] Piskorz J, Radlein D, Scott DS. On the mechanism of the rapid pyrolysis of cellulose. Journal of Analytical and Applied Pyrolysis, 1986; 9(2): 121-37.
- [287] Šelih VS, Strlič M, Kolar J, Pihlar B. The role of transition metals in oxidative degradation of cellulose. Polym Degrad Stabil, 2007; 92(8): 1476-81.

- [288] Lu Q, Xiong WM, Li WZ, Guo QX, Zhu XF. Catalytic pyrolysis of cellulose with sulfated metal oxides: A promising method for obtaining high yield of light furan compounds. Bioresource Technology, 2009; 100(20): 4871-6.
- [289] Tao F, Song H, Chou L. Efficient conversion of cellulose into furans catalyzed by metal ions in ionic liquids. Journal of Molecular Catalysis A: Chemical, 2012 357(0): 11-8.
- [290] Han J, Kim H. The reduction and control technology of tar during biomass gasification/pyrolysis: An overview. Renewable and Sustainable Energy Reviews, 2008; 12(2): 397-416.
- [291] López D, Acelas N, Mondragón F. Average structural analysis of tar obtained from pyrolysis of wood. Bioresource Technology, 2010; 101(7): 2458-65.
- [292] Suuberg EM, Milosavljevic I, Oja V. Two-regime global kinetics of cellulose pyrolysis: The role of tar evaporation. International Symposium on Combustion, 1996; 26(1): 1515-21.
- [293] Fushimi C, Katayama S, Tsutsumi A. Elucidation of interaction among cellulose, lignin and xylan during tar and gas evolution in steam gasification. Journal of Analytical and Applied Pyrolysis, 2009; 86(1): 82-9.
- [294] Hajaligol MR, Howard JB, Peters WA. An experimental and modeling study of pressure effects on tar release by rapid pyrolysis of cellulose sheets in a screen heater. Combustion and Flame, 1993; 95(1–2): 47-60.
- [295] Shen DK, Gu S, Bridgwater AV. The thermal performance of the polysaccharides extracted from hardwood: Cellulose and hemicellulose. Carbohydrate Polymers, 2010; 82(1): 39-45.
- [296] Di BC. Influences of model assumptions on the predictions of cellulose pyrolysis in the heat transfer controlled regime. Fuel, 1996; 75(1): 58-66.
- [297] Harris PJF. Non-graphitizing Carbons, In: Encyclopedia of Materials: Science and Technology, K.H.J.B. Editors-in-Chief: , et al., Editors. 2001, Elsevier: Oxford. p. 6197-6202.
- [298] White, R.H. and M.A. Dietenberger, Wood products: thermal degradation and fire, In: Encyclopedia of materials: science and technology, Jürgen Buschow KH, Cahn RW, Flemings MC, Ilschner B, Kramer DJ, Mahajan S, et al. editors,2nd ed, Elsevier: Oxford; 2001 p. 9712-9716.

- [299] Halpern Y, Riffer R, Broido A. Levoglucosenone (1,6-anhydro-3,4-dideoxydelta3-beta-d-pyranosen-2-one)-major product of acid-catalyzed pyrolysis of cellulose and related. Journal of Organic Chemistry, 1973; 38(2): 204-9.
- [300] Amarasekara AS, Ebede CC. Zinc chloride mediated degradation of cellulose at 200°C and identification of the products. Bioresource Technology, 2009; 100(21): 5301-4.
- [301] Azzam AM. Saccharification of bagasse cellulose pretreated with ZnCl2 and HCl. Biomass, 1987; 12(1): 71-7.
- [302] Dollimore D, Hoath MJ. The kinetics of the oxidative degradation of cellulose and cellulose doped with chlorides. Thermochimica Acta, 1987; 121(0): 273-82.
- [303] Horrocks AR. Developments in flame retardants for heat and fire resistant textiles—the role of char formation and intumescence. Polymer Degradation and Stability, 1996; 54(2–3): 143-54.
- [304] Gaan S, Sun G. Effect of phosphorus and nitrogen on flame retardant cellulose: A study of phosphorus compounds. Journal of Analytical and Applied Pyrolysis, 2007; 78(2): 371-7.
- [305] Price D, Horrocks AR, Akalin M, Faroq AA. Influence of flame retardants on the mechanism of pyrolysis of cotton (cellulose) fabrics in air. Journal of Analytical and Applied Pyrolysis, 1997; 40–41(0): 511-24.
- [306] Rupper P, Gaan S, Salimova V, Heuberger M. Characterization of chars obtained from cellulose treated with phosphoramidate flame retardants. Journal of Analytical and Applied Pyrolysis, 2010; 87(1): 93-8.
- [307] Mamleev V, Bourbigot S, Yvon J. Kinetic analysis of the thermal decomposition of cellulose: The main step of mass loss. Journal of Analytical and Applied Pyrolysis, 2007; 80(1): 151-65.
- [308] Várhegyi G, Szabó P, Antal Jr MJ. Kinetics of the thermal decomposition of cellulose under the experimental conditions of thermal analysis. Theoretical extrapolations to high heating rates. Biomass and Bioenergy, 1994; 7(1–6): 69-74.
- [309] Madorsky SL. Termicheskoje Razloshenie Organischeskikh Polimerov. MIR Moscow, 1967: 327.
- [310] Zugenmaier P. Conformation and packing of various crystalline cellulose fibers.Progress in Polymer Science, 2001; 26(9): 1341-417.

- [311] Lédé J. Cellulose pyrolysis kinetics: An historical review on the existence and role of intermediate active cellulose. Journal of Analytical and Applied Pyrolysis, 2012; 94(0): 17-32.
- [312] Paris O, Zollfrank C, Zickler GA. Decomposition and carbonisation of wood biopolymers—a microstructural study of softwood pyrolysis. Carbon, 2005; 43(1): 53-66.
- [313] Soares S, Camino G, Levchik S. Comparative study of the thermal decomposition of pure cellulose and pulp paper. Polymer Degradation and Stability, 1995; 49(2): 275-83.
- [314] Hajaligol M. Low temperature formation of aromatic hydrocarbon from pyrolysis of cellulosic materials. Fuel and Energy Abstracts, 2002; 43(4): 252
- [315] Radlein D, Piskorz J, Scott DS. Fast pyrolysis of natural polysaccharides as a potential industrial process. Journal of Analytical and Applied Pyrolysis, 1991; 19(0): 41-63.
- [316] Shafizadeh F. Pyrolysis and Combustion of Cellulosic Materials, In: Advances in Carbohydrate Chemistry, Melville LW, Tipson RS, editors, Academic Press;1968 p. 419-74.
- [317] Konkin AA. Production of Cellolose Based Carbon Fibrous Materials, In: Strong Fibres, Watt W, Perov VB, editors, Elsevier Science Publishers: Amsterdam, 1985 p. 275-325.
- [318] Davidson HW.Production of graphite.US patent 3,104,159.
- [319] Phillips M, Losty HHW. An electron microscopic examination of porosity in cellulose carbon. Carbon, 1964; 1(3): 376.
- [320] Segal L, Creely JJ, Martin AE, Conrad JR, Conard CM. An Empirical Method for Estimating the Degree of Crystallinity of Native Cellulose Using the X-Ray Diffractometer. Textile Research Journal, 1959; 29: 786-94.
- [321] Li Y, Lin M, Davenport JW. Ab Initio Studies of Cellulose I: Crystal Structure, Intermolecular Forces, and Interactions with Water. Journal of Physical Chemistry C, 2011; 115(23): 1533–9.
- [322] Langan P, Sukumar N, Nishiyama Y, Chanzy H. Synchrotron X-ray structures of cellulose I-beta and regenerated cellulose II at ambient temperature and 100 K. Cellulose, 2005; 12(6): 551-62.
- [323] Oh SY, Yoo DI, Shin Y, Kim HC, Kim HY, Chung YS, et al. Crystalline structure analysis of cellulose treated with sodium hydroxide and carbon

dioxide by means of X-ray diffraction and FTIR spectroscopy. Carbohydrate Research, 2005; 340(15): 2376-91.

- [324] Nelson ML, O'connor RT. Relation of certain infrared bands to cellulose crystallinity and crystal lattice type. Part I: Spectra of lattice types I, II, III and of amorphous cellulose. Journal of Applied Polymer Science, 1964; 8: 1311-24.
- [325] Cao Y, Tan HM. Effects of cellulase on the modification of cellulose. Carbohydrate Research, 2002; 337(14): 1291-6.
- [326] Nelson ML, O'connor RT. Relation of certain infrared bands to cellulose crystallinity and crystal lattice type. Part II: A new infrared ratio for estimation of crystallinity in celluloses I and II. Journal of Applied Polymer Science, 1964; 1325-41.
- [327] Antal MJ, Friedman HL, Rogers FE. Kinetics of cellulose pyrolysis in nitrogen and steam. Combustion Science and Technology, 1980; 21(3-4): 141-52.
- [328] Shen DK, Gu S. The mechanism for thermal decomposition of cellulose and its main products. Bioresource Technology, 2009; 100(24): 6496-504.
- [329] Escalona Platero E, Peñarroya Mentruit M. FTIR characterisation of Re₂(CO)₂O adsorbed on sulfated zirconium dioxide. Polyhedron, 1997; 16(20): 3489-92.
- [330] Ramasubramanian N, Ling VC. Fourier transform infrared reflection (FTIR) spectroscopy of corrosion films on irradiated zirconium alloys. Journal of Nuclear Materials, 1990; 175(3): 237-43.
- [331] Weishauptová Z, Machovič V, Novotná M, Medek J. FTIR spectroscopy and sorption of water vapour as methods of identification of hydrated zirconium dioxide on a corrosion layer of zirconium alloy. Materials Chemistry and Physics, 2007; 102(2–3): 219-24.
- [332] Dillon AC, Parilla PA, Alleman JL, Gennett T, Jones KM, Heben MJ. Systematic inclusion of defects in pure carbon single-wall nanotubes and their effect on the Raman D-band. Chemical Physics Letters, 2005; 401(4–6): 522-8.
- [333] Luo Z, Cong C, Zhang J, Xiong Q, Yu T. The origin of sub-bands in the Raman D-band of graphene. Carbon, 2012; 50(11): 4252-8.
- [334] Park JS, Reina A, Saito R, Kong J, Dresselhaus G, Dresselhaus MS. Band Raman spectra of single, double and triple layer graphene. Carbon, 2009; 47(5): 1303-10.

- [335] Yoshida A, Kaburagi Y, Hishiyama Y. Full width at half maximum intensity of the G band in the first order Raman spectrum of carbon material as a parameter for graphitization. Carbon, 2006; 44(11): 2333-5.
- [336] Ferrari, A.C., Raman spectroscopy of graphene and graphite: Disorder, electron-phonon coupling, doping and nonadiabatic effects. Solid State Communications, 2007. 143(1-2): p. 47-57.